# IDAHO TRANSPORTATION DEPARTMENT RESEARCH REPORT

Implementation of Balanced Asphalt Mix Design of Asphalt Mixtures Prepared with Reclaimed Asphalt Pavements and Rejuvenators for Enhanced Performance RP 292

Ву

Emad Kassem, Hussain Al Hatailah, Mohammed Abu Saq, and Fouad Bayomy University of Idaho

and

Yang Lu and Amanda Mullins Boise State University

Prepared for

Idaho Transportation Department ITD Research Program, Planning Services Highways Planning and TECM Division

April 2023



YOUR Safety 💿 💿 🕨 YOUR Mobility 💿 💿 🕨 YOUR Economic Opportunity

## Disclaimer

This document is disseminated under the sponsorship of the Idaho Transportation Department and the United States Department of Transportation in the interest of information exchange. The State of Idaho and the United States Government assume no liability of its contents or use thereof.

The contents of this report reflect the view of the authors, who are responsible for the facts and accuracy of the data presented herein. The contents do not necessarily reflect the official policies of the Idaho Transportation Department or the United States Department of Transportation.

The State of Idaho and the United States Government do not endorse products or manufacturers. Trademarks or manufacturers' names appear herein only because they are considered essential to the object of this document.

This report does not constitute a standard, specification or regulation.

# **Technical Report Documentation Page**

1. Report No. FHWA-ID- 23-292	2. Government Accession No	o. 3.	3. Recipient's Catalog No.				
4. Title and Subtitle	5.	5. Report Date					
Implementation of Balanced Asphalt M	repared	April 2023					
with Reclaimed Asphalt Pavements and	d Rejuvenators for Enhanced	6.	Performing Organization Co	ode			
Performance							
7. Author(s)		8.	8. Performing Organization Report No.				
Emad Kassem, <u>https://orcid.org/0000-0</u>	0002-4331-6692		[Billing code]				
Hussain Al Hatailah, Mohammed Abu Sa	aq, Fouad M.S. Bayomy, Yang L	u, and					
Amanda Mullins							
9. Performing Organization Name and Add	dress	10	). Work Unit No. (TRAIS)				
University of Idaho							
875 Perimeter Drive MS 1022, Moscow,	, ID 83844-1022	11	. Contract or Grant No.				
Boise State University			EN4988				
1910 University Drive Boise, Idaho 8372	25-2060						
12. Sponsoring Agency Name and Address		13	<ol> <li>Type of Report and Period</li> </ol>	Covered			
Idaho Transportation Department (SPR	l) . Contracting Comisso Research	h	Final Report				
Program	s, contracting services, Research	n	09/01/2020 - 05/31/2023				
PO Box 7129		14	14. Sponsoring Agency Code				
Boise. ID 83707-7129			RP 292				
15. Supplementary Notes							
Project performed in cooperation with	the Idaho Transportation Depai	rtment and Fe	deral Highway Administratio	n.			
The use of Reclaimed Asphalt Pavement transportation agencies limit the RAP conter into asphalt mixtures, it may result in stiffer binder grade adjusting is typically followed to increase the cost of Hot Mix Asphalt (HMA) the use of rejuvenators to improve the perfor to ensure sufficient resistance to cracking an Seven different commercially available rejuve that the use of rejuvenators in asphalt mixtures more cost effective to incorporate rejuvenate mixtures with certain rejuvenators and high cost savings. The use of tall oil and waste ve compared to other rejuvenators. The result approach to optimize the design of asphalt m and rutting resistance. The balanced mix de provide acceptable performance. Furtherm produced and used in Idaho. The results der addition, the current practice followed by IT	sustainable soli e environmenta AP content, tha of RAP which m lability of such ifferent RAP cor- botential to rest d offer environ e content (e.g., 9 e to that of the n improving the sted the import vonducting cost sing and rutting expected to exilo	utions into civil engineering p I benefits and cost savings of inc t could be prone to premature ay require the use of softer bind softer binders in practice. This s itents, through a balanced mix d ore the rheological properties of orogram of this study. The results mental benefits and cost savings 50 or 70 percent). The cracking p virgin mixture (0 percent RAP) - cracking resistance and providi ance of implementing a balance vide adequate performance in te analysis to compare different al performance of 23 different as hibit good cracking and rutting p	ractices. Many orporating RAP cracking. Virgin ders. This could tudy examined esign approach of aged binders. a demonstrated be However, it is performance of with additional ng cost savings ced mix design rms of cracking ter natives that phalt mixtures performance. In				
17. Key Words		18. Distribu	tion Statement				
Reclaimed Asphalt Pavement, RAP, Recy	cling Agent, Rejuvenators,	Copies av	ailable from the <u>ITD Researc</u>	h Program			
Fatigue Cracking, Thermal Cracking, Ruth	ting						
19. Security Classification (of this report)	20. Security Classification (of	this page)	21. No. of Pages	22. Price			
Unclassified	Unclassified		160	None			

## Acknowledgments

This project is funded by Idaho Transportation Department (ITD) from SPR funds. It is performed in cooperation with ITD. The authors would like to acknowledge all members of the research project Technical Advisory Committee (TAC) for their valuable feedback and cooperation all over the project tasks. The authors would like also to acknowledge the support from the National Institute for Advanced Transportation Technology (NIATT) and the Department of Civil and Environmental Engineering at the University of Idaho.

## **Technical Advisory Committee**

Each research project is overseen by a Technical Advisory Committee (TAC), which is led by an ITD project sponsor and project manager. The TAC is responsible for monitoring project progress, reviewing deliverables, ensuring that study objectives are met, and facilitating implementation of research recommendations, as appropriate. ITD's Research Program Manager appreciates the work of the following TAC members in guiding this research study.

- Project Sponsor: Chad Clawson
- Project Manager: Mike Copeland
- TAC Members: John Arambarri, Chad Clawson, Kyle Holman, Ben Ward, Randy Durland
- FHWA-Idaho Advisor: Kyle Holman, P.E.

# **Table of Contents**

Exe	ecutive Summary	. 12
	Key Findings	. 13
1.	Introduction	. 15
	Problem Statement	. 15
	Objectives	. 16
	Tasks	. 16
	Task 1: Literature review	. 16
	Task 2: Select and procure testing materials	. 16
	Task 3: Prepare asphalt mixture test specimens	. 17
	Task 4: Conduct laboratory testing	. 17
	Task 5: Analyze the laboratory testing results	. 18
	Task 6: Conduct cost analysis to assess economic savings	. 18
	Task 7: Develop recommendations and guidelines	. 19
	Task 8: Prepare the final report	. 19
	Report Organization	. 19
2.	Literature Review	. 20
	Introduction	20
		. 20
	Rejuvenators	. 20 . 20
	Rejuvenators Incorporating RAP in Asphalt Mixtures	. 20 . 20 . 33
	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach	. 20 . 20 . 33 . 37
	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test	. 20 . 20 . 33 . 37 . 40
	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature	. 20 . 20 . 33 . 37 . 40 . 42
	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature Hamburg Wheel-Tracking Test	. 20 . 20 . 33 . 37 . 40 . 42 . 42
3.	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature Hamburg Wheel-Tracking Test Material Description and Experimental Design	. 20 . 20 . 33 . 37 . 40 . 42 . 42 . 45
3.	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature Hamburg Wheel-Tracking Test Material Description and Experimental Design Material Description	. 20 . 20 . 33 . 37 . 40 . 42 . 42 . 45
3.	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature Hamburg Wheel-Tracking Test Material Description and Experimental Design Material Description Reclaimed Asphalt Pavement	. 20 . 20 . 33 . 37 . 40 . 42 . 42 . 45 . 45
3.	Rejuvenators Incorporating RAP in Asphalt Mixtures Balanced Mix Design Approach Indirect Tensile Asphalt Cracking Test Creep-Compliance and Strength Test as low Temperature Hamburg Wheel-Tracking Test Material Description and Experimental Design Material Description Reclaimed Asphalt Pavement Rejuvenators	. 20 . 20 . 33 . 37 . 40 . 42 . 42 . 45 . 45 . 45 . 46

	Plant-Mixed Laboratory-Compacted (PMLC) Test Specimens	49
	Laboratory Testing	51
	Evaluation of Cracking Performance	51
	Evaluation of Rutting Performance	52
4.	Evaluation of the Performance of Rejuvenators Incorporated in Mixtures with High RAP Content	54
	Evaluation of Asphalt Mixtures Prepared with RAP No. 1	54
	Effect of RAP Content	55
	Effect of Rejuvenator Type	58
	Effect of Rejuvenator Dose	60
	Evaluation of RAP No. 2 Mixtures	61
	Effect of RAP Content	61
	Effect of Rejuvenator Type	64
	Effect of Rejuvenator Dose	66
	Effect of Rejuvenator on Cracking Performance of Mixtures with Different Binder Grades	69
	Evaluation of Rutting Performance	71
	Evaluation of RAP No. 3 Mixtures	73
	Effect of RAP and Rejuvenator Type	74
	Evaluation of Low-Temperature Cracking	76
	Evaluation of Rutting Performance	80
	Evaluation of RAP No. 4 Mixtures	80
	Effect of RAP Content	81
	Effect of Rejuvenator Type	81
	Effect of Rejuvenator Dose	83
	Evaluation of Rutting Performance	84
5.	Evaluation of Field Mixes	86
	Cracking and Rutting Performance Thresholds	86
	Evaluation of Cracking Resistance of Field Mixes	87
	Evaluation of Correlation and Coefficient of Variation for Cracking Performance Indicators	90
	Evaluation of Rutting Resistance of Field Mixes	92
6.	Cost Analysis	94

	Materials Cost	94
	Cost Comparison Associated with RAP No. 2	95
	Cost Associated with RAP No. 3	98
	Cost Associated with RAP No. 4	99
7.	Conclusions and Recommendations	102
8.	Cited Work	106
Арр	endix A. RAP Properties	111
Арр	pendix B. Rejuvenator Dose using Blending Charts	115
Арр	pendix C. LMLC Mix Design Summary	117
Арр	endix D. PMLC Mix Design Summary	122
Арр	pendix E. Additional Figures of Chapter 4	144
Арр	endix F. Cost Analysis Detail	159

## List of Tables

Table 1. Physical Properties of Recycling Agents (ASTM D4552)	. 21
Table 2: Recycling Agent Types (NCAT 2014)	. 21
Table 3. Semi-Circular Bending Fracture Test Results (Nabizadeh et al. 2017)	. 24
Table 4. Rejuvenator Content Range and Dose Validation (S. Im et al. 2016)	. 31
Table 5. Percent of RAP Mixes Using Softer Binder and/or Recycling Agents by State (NAPA 2021)	. 35
Table 6. Grade Adjustment for RAP (ITD Specification 2018)	. 37
Table 7. Typical Adjusted Binder Grades (ITD Specification 2018)	. 37
Table 8. Intermediate temperature cracking most promising performance indicators and its associate	
testing standards after (Alkuime et al. 2020)	. 39
Table 9. HWTT rutting performance threshold (after: Kassem at al. 2019)	. 44
Table 10. Rejuvenator Test Doses	. 48
Table 11. Laboratory-Mixed, Laboratory-Compacted Testing Matrix at UI	. 49
Table 12. Laboratory-Mixed, Laboratory-Compacted Testing Matrix at BSU	. 49
Table 13. PMLC Properties	. 50
Table 14. Proposed Thresholds for Monotonic Cracking Performance Indices (after Kassem at al. 2019	)86
Table 15. Pearson Coefficient (r) for Cracking Performance Indicators	. 91
Table 16. Summary Materials Cost	. 95
Table 17. RAP No. 2 Cost Analysis Summary	97
Table 18. RAP No. 3 Cost Analysis Summary	. 99

# List of Figures

Figure 1. Effect of Rejuvenators on Creep Compliance and Tensile Strength at -10°C (Zaumanis et al.	
2013)	22
Figure 2. Creep Compliance and Tensile Strength at -10 °C (Zaumanis et al. 2015)	23
Figure 3. Effect of Different Recycling Agents on IDT Load-Displacement Curve (Nabizadeh et al. 2017)	23
Figure 4. Effect of Rejuvenators on Flexibility Index and Cracking Resistance Index (Kaseer et al. 2020).	24
Figure 5. Effect of Rejuvenator and Aging on Cracking Performance (NRRA 2022)	25
Figure 6. Effect of Rejuvenators on Rutting Performance (Zaumanis et al. 2014)	26
Figure 7. Effect of Rejuvenated RAP Mixture on Rutting Performance (Bajaj et al. 2020)	26
Figure 8. HWTT and APA Rutting Performance of Rejuvenated Mixtures (Kaseer et al. 2020)	27
Figure 9. Effect of Rejuvenators on Performance Grade (Zaumanis et al. 2014)	28
Figure 10. Effect of Rejuvenators on Workability of RAP Mixtures (Zaumanis et al. 2014)	28
Figure 11. Performance Grade blending Chart (Arámbula-Mercado et al. 2017)	29
Figure 12. High, Low, and intermediate PG for Rejuvenated Extracted Binder (Zaumanis et al. 2014)	30
Figure 13. Minimum Rejuvenator Dose (Zaumanis et al. 2014)	30
Figure 14. High and Low Temperature Performance Grade for Rejuvenated Mixtures (S. Im et al. 2016)	32
Figure 15. Cost Saving Potential Associated with Incorporating RAP and Rejuvenator Zaumanis et al.	
2014)	33
Figure 16. RAP Use by Sector (Million Tons) and Average Percent RAP Used by Sector (NAPA 2021)	34
Figure 17. Average Estimated RAP Content in Asphalt Mixes by State (NAPA 2021)	34
Figure 18. An example of balancing rutting and cracking resistance (Zhou et al. 2006)	38
Figure 19. Hamburg wheel tracking test setup (Kassem et al. 2011)	43
Figure 20. APA Jr. for Hamburg wheel tracking test	43
Figure 21. RAP Gradations	46
Figure 22. RAP Binder Content	46
Figure 23. Rejuvenators evaluated at UI	47
Figure 24. Rejuvenators evaluated at BSU	47
Figure 25. IDT Test Setup	51
Figure 26. Creep-Compliance Test specimen	52
Figure 27. APA Jr. and HWTT Test Specimen and Setup	53
Figure 28. Effect of RAP Content on Cracking Performance for RAP No. 1	56
Figure 29. Effect of RAP Content of RAP No. 1 on IDT <sub>Strength</sub>	57
Figure 30. Effect of RAP Content of RAP No. 1 on Weibull <sub>CRI</sub>	57
Figure 31. Effect of Rejuvenator Type on Cracking Performance at 25% RAP No. 1	59
Figure 32. Effect of Rejuvenator Type on Cracking Performance at 50% RAP No. 1	59
Figure 33. Effect of Rejuvenator Type on Cracking Performance at 75% RAP No. 1	60
Figure 34. Effect of Rejuvenator Dose on Cracking Performance for RAP No. 1	61
Figure 35. Effect of RAP Content on Cracking Performance for RAP No. 2	62

Figure 36.	Effect of RAP Content of RAP No. 2 on IDT <sub>Strength</sub>	63
Figure 37.	Effect of RAP Content of RAP No. 2 on $Weibull_{CRI}$	63
Figure 38.	Effect of Rejuvenator Type on Cracking Performance at 25% RAP No. 2	65
Figure 39.	Effect of Rejuvenator Type on Cracking Performance at 50% RAP No. 2	65
Figure 40.	Effect of Rejuvenator Type on Cracking Performance at 70% RAP No. 2	66
Figure 41.	Effect of Rejuvenator Dose on Cracking Performance at 25% on RAP No. 2	67
Figure 42.	Effect of Rejuvenator Dose on Cracking Performance at 50% on RAP No. 2	68
Figure 43.	Effect of Rejuvenator Dose on Cracking Performance at 70% on RAP No. 2	68
Figure 44.	Effect of Rejuvenator on Cracking Performance for Different Binder Grades	70
Figure 45.	Effect of Rejuvenator on IDT <sub>Strength</sub> for Different Binder Grades	70
Figure 46.	$\label{eq:effect} Effect \ of \ Rejuvenator \ on \ Weibull_{CRI} \ for \ Different \ Binder \ Grades \ Effect \ of \ Rejuvenators \ \dots \dots \dots$	71
Figure 47.	Evaluation of HWTT Rutting Performance of Asphalt Mixtures with RAP No. 2	72
Figure 48.	HWTT Test Specimens with 70% RAP and Rejuvenator 4 at 1 Percent Dose	73
Figure 49.	HWTT Test Specimens with 70% RAP and Rejuvenator 4 at 2 Percent Dose	73
Figure 50.	Effect of RAP Content and Rejuvenator Type on IDEAL-CT_{\text{index}} for RAP No. 3	75
Figure 51.	Effect of RAP Content and Rejuvenator Type on IDT <sub>Strength</sub> for RAP No. 3	75
Figure 52.	Effect of RAP Content and Rejuvenator Type on Weibull $_{\mbox{\tiny CRI}}$ for RAP No. 3	76
Figure 53.	Creep Compliance Results at -20 °C	77
Figure 54.	Creep Compliance Results at -10 °C	78
Figure 55.	Creep Compliance Results at 0 °C	78
Figure 56.	Creep Compliance Results at Different Temperatures	79
Figure 57.	Measured IDT <sub>Strength</sub> of Test Mixtures at -10 °C	79
Figure 58.	Evaluation of Rutting Performance of Mixtures with RAP No. 3	80
Figure 59.	Effect of RAP Content on Cracking Performance of Mixtures Prepared with RAP No. 4	81
Figure 60.	Effect of Rejuvenator Type on 25% RAP No. 4	82
Figure 61.	Effect of Rejuvenator Dose on 25% RAP No. 4	83
Figure 62.	Evaluation of Rutting Performance on RAP No. 4	84
Figure 63.	Effect of Rejuvenator Type and Dose on Rutting Performance of RAP No. 4	85
Figure 64.	IDEAL-CT <sub>Index</sub> for PMLC Mixes	88
Figure 65.	WeibullCRI for PMLC Mixes	89
Figure 66.	IDT <sub>Strength</sub> for PMLC Mixes	90
Figure 67.	Coefficient of Variation of Cracking Performance Indicators for PMLC Mixes	91
Figure 68.	Hamburg Rutting Depth for PMLC Mixes	93
Figure 69.	Cost Comparison of Mixtures with RAP No. 2 and Rejuvenators	97
Figure 70.	Cost Comparison of Mixtures with RAP No. 3 and Rejuvenators	99
Figure 71.	Cost Comparison of Mixtures with RAP No. 4 and Rejuvenators1	01

## List of Abbreviations and Acronyms

FHWA Federal Highway Administration
ITD Idaho Transportation Department
TAC Technical Advisory Committee
IDT Indirect Tensile Strength
AASHTO American Association of State Highway and Transportation Officials
AASHTO American Association of State Highway and Transportation Officials
HWTT Hamburg Wheel Tracking Test
APA Asphalt Pavement Analyzer
HMA Hot Mix Asphalt
NMASNominal Maximum Aggregate Size
UIUniversity of Idaho
BSU Boise State University
IDEAL-CT <sub>Index</sub> Indirect Tension Asphalt Cracking Test
RAP Reclaimed Asphalt Pavement
BMDBalanced Mix Design
COVCoefficient of Variation
ASTM American Standards for Testing and Materials
NCATNational Center for Asphalt Technology
OBC Optimum Binder Content
NCHRP National Cooperative Highway Research Program
MTS Material Testing System
HSD Honestly Significant Difference Test

# **Executive Summary**

Many State Departments of Transportation (DOTs) allow incorporation of Reclaimed Asphalt Pavement (RAP) into asphalt mixtures. The use of RAP promotes and integrates sustainable solutions into civil engineering practices. The amount of virgin aggregates and asphalt binders can be reduced by using RAP in asphalt mixtures which offers environmental benefits and cost savings. Many transportation agencies limit the RAP content to a small percentage. For several years, the Idaho Transportation Department (ITD) allowed a RAP content of over 50 percent as selected by the contractor. Currently, ITD only allows up to 30 percent RAP Binder Replacement (RAPBR) in asphalt mixtures. Despite the environmental benefits and cost savings of incorporating RAP into asphalt mixtures, it may result in stiffer mixtures, especially at a higher RAP content, that could be prone to premature cracking. Adjusting the virgin binder grade (i.e., softer binder) is the most common method used to account for the stiffening effect of the RAP in the asphalt mixture. This could increase the production cost of the Hot Mix Asphalt (HMA) due to the limited availability of such softer binders in practice.

Rejuvenators have been used mainly in surface treatment emulsions to improve the properties of existing asphalt surface layers. Rejuvenators have the potential to restore the rheological properties of aged binders. This study examined the use of rejuvenators to improve the performance of asphalt mixtures, with different RAP contents, through a balanced mix design approach to ensure sufficient resistance to cracking and rutting. Seven different commercially available rejuvenators were acquired and included in the testing program. These rejuvenators included tall oil, aromatic extract, bio-based forestry, engineered product, triglycerides and fatty acids product, bio-based oil, petroleum-based oil for rejuvenators R1 through R7, respectively. Asphalt mixtures prepared with different doses of each rejuvenator were also examined. Furthermore, the researchers prepared and tested asphalt mixtures at different binder contents (i.e., optimum binder content [OBC] and OBC+0.5 percent) and binder grades.

This study also evaluated the cracking and rutting performance of asphalt mixtures currently produced and used in the state. Loose asphalt mixtures from 23 projects were obtained and tested. The mixtures included different mix designs, binder grades, RAP content, and percent RAP binder replacement. The researchers evaluated the performance of the test mixtures based on the performance thresholds developed in RP 261 (Kassem et al. 2019) and the ones proposed in the respective standards and literature. The cracking resistance was evaluated for all test mixtures using the Indirect Tensile (IDT) strength test and calculation of IDEAL-CT<sub>Index</sub>. The researchers also calculated additional cracking parameters including IDT<sub>strength</sub> and WeibullCRI. The Hamburg Wheel Tracking Test (HWTT) was conducted using the Asphalt Pavement Analyzer JR. (APA Jr.) to evaluate the rutting performance and moisture susceptibility of test mixtures. In addition, the researchers measured the creep compliance and strength for selected test mixtures at low temperatures to evaluate the thermal cracking performance.

## **Key Findings**

The key findings of this study are summarized below:

- The cracking resistance decreased with the increase of RAP content for all RAP examined in this study. Both IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub> decreased with the increase of RAP content which demonstrates reduced cracking resistance. Meanwhile, the IDT<sub>Strength</sub> increased with the increase of RAP content which demonstrates that the mixtures become stiffer with the addition of RAP.
- The use of rejuvenators in mixtures with low RAP content (e.g., 25 percent), especially for mixtures with good cracking performance, did not improve the cracking resistance (i.e., did not increase IDEAL-CT<sub>Index</sub>). It was observed that the addition of rejuvenators could be detrimental to the cracking resistance at low RAP content for mixtures with good cracking resistance.
- Mixtures prepared with 70 percent RAP had lower IDEAL-CT<sub>Index</sub> values compared to the ones without RAP (0 percent RAP) irrespective of the binder grade.
- The favorable effect of rejuvenators in asphalt mixtures is observed in mixtures with higher RAP content (e.g., 70 percent) for different RAP sources evaluated in this study. In some cases (e.g., RAP No. 2 and RAP No. 3), it was possible to produce mixtures prepared with 70 percent RAP and rejuvenators that provided comparable cracking performance to the mixture without RAP.
- Increasing the rejuvenator dose does not necessarily improve the cracking resistance (i.e., R4 and R6). In fact, in some cases, it could adversely impact cracking performance.
- The use of rejuvenator R1 (tall oil) and rejuvenator R5 (waste vegetable oil) with mixtures with high RAP contents provided the best performance compared to other rejuvenators examined in this study, and these mixtures had comparable cracking performance to the virgin mixture (i.e., 0 percent RAP).
- Rejuvenator R4 (engineered product) at a higher dose improved the cracking performance of laboratory mixtures with RAP; however, these mixtures failed the rutting criteria prematurely (i.e., the mixtures were too soft). These results demonstrated the importance of the balanced mix design approach to satisfy both cracking and rutting criteria.
- Increasing the binder content was found to increase the cracking resistance for some mixtures with or without RAP. This further emphasizes the effectiveness of the balanced mix design approach.
- The use of rejuvenator R1 (tall oil) and rejuvenator R5 (waste vegetable oil) provided higher creep compliance compared to the virgin mixture (0 percent RAP) and control mixture (70 percent RAP) which demonstrated improved cracking resistance at a low temperature.

- Some rejuvenators (e.g., R1 and R5) were effective in improving the cracking performance at higher RAP content which offers significant environmental and economic benefits. Furthermore, these rejuvenators enhanced the thermal cracking performance.
- The use of rejuvenators in asphalt mixtures with RAP is beneficial and could offer environmental benefits and cost savings. However, it is more cost effective to incorporate rejuvenators in mixtures with high RAP content (i.e., 50 or 70 percent).
- Most of the field mixes (18 out of 23) had IDEAL-CT<sub>Index</sub> values of 73.7 or higher which indicates that these mixtures are expected to exhibit good cracking resistance in the field. Four projects were within the moderate cracking performance range, while only one project had an IDEAL-CT<sub>Index</sub> less than the minimum threshold of 26.4 which demonstrates poor cracking resistance.
- The HWTT rut depth for the field mixes ranged from 1.12 mm to 4.41 mm after 20,000 passes. Therefore, all the mixtures had a rut depth way below the maximum threshold of 12.5 mm after 20,000 passes.
- The Pearson correlation results demonstrated that Weibull<sub>CRI</sub> and IDEAL-CT<sub>Index</sub> had a strong correlation (r = 0.964); however, Weibull<sub>CRI</sub> had lower variability in the test results (average COV = 6.6 percent) compared to IDEAL-CT<sub>Index</sub> (average COV = 18.8 percent) which is consistent with the results of ITD RP 280 (Kassem et al. 2021).

# 1. Introduction

## **Problem Statement**

Many State Departments of Transportation (DOTs) allow incorporation of Reclaimed Asphalt Pavement (RAP) into asphalt mixtures. The use of RAP promotes and integrates sustainable solutions into civil engineering practices. The amount of virgin aggregates and asphalt binders can be reduced by using RAP in asphalt mixtures which offers environmental benefits and cost savings. Many transportation agencies limit the RAP content to about 25 percent. For several years, the Idaho Transportation Department (ITD) allowed a RAP content of over 50 percent as selected by the contractor. Currently, ITD only allows up to 30 percent RAP in asphalt mixtures. Despite the environmental benefits and cost savings of incorporating RAP into asphalt mixtures, it may result in stiffer mixtures, especially at a higher RAP content, that could be prone to premature cracking. Furthermore, the addition of RAP may adversely affect thermal cracking even if a small amount of RAP is used. Virgin binder grade adjusting is typically followed to account for the stiffening effect of the RAP. The current adjustment process is performed using modified AASHTO M 323 where a blending chart is used for virgin binder selection if the RAP content exceeds 30 percent. If the RAP content is between 17 percent and 30 percent, one grade softer virgin binder is typically selected. There is no need to change the virgin binder grade if the RAP content is less than 17 percent. The use of softer binders may increase the cost of Hot Mix Asphalt (HMA) production due to the limited availability of such softer binders in practice.

Rejuvenators have been used mainly in surface treatment emulsions to improve the properties of existing asphalt surface layers. Rejuvenators have the potential to restore the chemical and rheological properties of aged binders. The ratio of maltenes to asphaltenes is reduced due to oxidative aging, and rejuvenators contain maltenes to restore the original ratio. There is a need to investigate the use of rejuvenators to improve the performance of asphalt mixtures containing different percentages of RAP (i.e., 25, 50, and 70 percent) using a balanced mix design approach that aims to produce mixtures with adequate performance in terms of cracking and rutting resistance based on the performance thresholds developed in RP261 (Kassem et al. 2019).

RP 261 conducted a comprehensive testing program that included laboratory-mixed laboratorycompacted, plant-mixed laboratory-compacted test specimens, and cores extracted from the field. The laboratory testing included two rutting tests (i.e., Hamburg Wheel Tracking Test [HWTT] and Asphalt Pavement Analyzer [APA]) and three monotonic cracking tests (i.e., indirect tension test, semi-circle bending flexibility index test, semi-circle bending Jc test), in addition to a dynamic cracking test. The researchers examined more than 20 performance indicators to evaluate the cracking and rutting performance of asphalt mixtures in Idaho. Based on the results, RP 261 proposed performance thresholds to ensure adequate resistance to cracking, rutting, and moisture damage. Performance thresholds for HWTT as well as APA rutting tests were proposed. In addition, the researchers developed performance thresholds for selected cracking tests including IDEAL-CT<sub>Index</sub>, N<sub>flex</sub> factor, and Weibull<sub>CRI</sub>. ITD has recently selected and implemented HWTT and IDEAL-CT<sub>Index</sub> to evaluate the resistance of asphalt mixtures to rutting and cracking, respectively. The developed performance thresholds in RP 261 especially for IDEAL-CTIndex and HWTT need to be further evaluated using additional asphalt mixtures currently produced in Idaho and revised as needed.

# Objectives

The main objectives of this study are:

- Evaluate the effect of rejuvenators on improving the performance of asphalt mixtures containing different percentages of RAP and reducing the need for softer binders which are more costly to obtain as compared to rejuvenators.
- Apply a balanced (engineered) mix design approach and performance thresholds, developed in RP 261, to optimize the mix design of asphalt mixtures prepared with RAP and rejuvenators for improved performance.
- Evaluate the performance thresholds developed in RP 261 especially for IDEAL-CT<sub>index</sub> and HWTT using additional asphalt mixtures currently produced in the state.
- Study the economic savings of using rejuvenators and RAP in asphalt mixtures as compared to the control mixtures.
- Develop recommendations on incorporating rejuvenators and RAP into asphalt mixtures that provide comparable or superior performance with respect to the control mixtures.

## Tasks

The researchers conducted five main tasks to achieve the objectives of this research study. These tasks are discussed in detail in this section.

### Task 1: Literature review

In this task, the researchers conducted a literature review of previous published research studies and collected pertinent information on the following subjects:

- Different types of rejuvenators and recycling agents used in asphalt mixtures prepared with RAP,
- Cracking and rutting performance of asphalt mixtures with RAP and rejuvenators,
- Effect of rejuvenators on binder performance grade,
- Methods used to select rejuvenator dose,
- Economic benefits of using rejuvenators in asphalt mixtures,
- Current practices of incorporating RAP in asphalt mixtures,
- Review of proposed balanced mix design and performance measures of RAP mixtures, and
- Test methods and associated performance thresholds to assess the cracking and rutting performance of asphalt mixtures.

#### Task 2: Select and procure testing materials

Under this task, the research team selected and procured local materials including aggregates, virgin binders, RAP, and rejuvenators, as well as loose asphalt mixtures. The test materials included the following:

- Virgin aggregates: two sources of river gravel were used; one in Lewiston and another one in Boise, Idaho.
- Asphalt binders: five binder grades were included (i.e., PG 58-34, PG 58-28, PG 64-28, PG 64-34, and PG 70-28).
- RAP materials: four sources of RAP were examined. Three were obtained from asphalt plants in Idaho, in addition to an artificial RAP that was produced in the laboratory.
- Rejuvenators: seven rejuvenators were obtained from different sources and included in the laboratory evaluation.
- Plant mixtures: a total number of 23 loose mixtures were collected from ITD paving projects with different characteristics including different mix designs, binder grades, percent of RAP content, and percent of RAP binder replacement.

### Task 3: Prepare asphalt mixture test specimens

Under this task, the team prepared asphalt mixture test specimens needed to execute the testing program. They prepared laboratory-mixed, laboratory-compacted specimens with the following characteristics.

- Different RAP content (e.g., 0, 25, 50, 70, and 75 percent).
- Seven different rejuvenators. Their doses to obtain optimum cracking and rutting performance varied.
- Five binder grades (e.g., PG 58-28, PG 58-34, PG 64-28, PG 64-34, and PG 70-28).
- Different binder content (Optimum binder content (OBC) and OBC +0.5 percent).

In addition, the researchers prepared plant-mixed, laboratory-compacted specimens to evaluate the performance thresholds developed in RP 261 with the focus on IDEAL- $CT_{Index}$  and HWTT tests. The IDEAL- $CT_{Index}$  test specimens are 150 mm in diameter and 62 mm in height and don't need to be cut or notched which is an advantage over the semicircular test specimens. The HWTT test specimens are 150 mm in diameter and 60 mm in height.

#### Task 4: Conduct laboratory testing

Under this task, the researchers conducted laboratory testing to evaluate the properties of test specimens. The following characteristics were evaluated.

- Rutting resistance and moisture susceptibility using HWTT. The HWTT is conducted in accordance with AASHTO T324. Cylindrical test specimens are subjected to accelerated reciprocating wheel loading. HWTT applies 705 N load on the surface of test specimens at a constant moving rate of 52 pass/minute. A set of four test specimens are used in each run. The test specimens are submerged in water for at least one hour before testing at a specified temperature (e.g., 50 °C). This test is used to assess the resistance to rutting as well as moisture susceptibility. The test is completed when the test specimen achieves a certain rut depth (e.g., 12.5 mm) or the specified number of passes (e.g., 20,000 passes) is applied.
- Fatigue cracking resistance (e.g., IDEAL-CT<sub>Index</sub> and WeibullCRI). IDEAL-CT<sub>Index</sub> and WeibullCRI values can be calculated from the same load-displacement curve obtained using the indirect tensile test (IDT). The IDT uses a circular specimen which is subjected to a compressive loading at 50 mm/min and the test is conducted at 25 °C. The IDEAL-CT<sub>Index</sub> was proposed by the researchers

at Texas A&M (ASTM D8225 – 19), while WeibullCRI was developed in Idaho in RP 261. Higher IDEAL-CT<sub>Index</sub> and WeibullCRI values indicate better resistance to cracking. IDEAL-CT<sub>Index</sub> is calculated as a function of fracture energy, post-peak slope, and strain tolerance. The WeibullCRI describes the entire load-displacement curve.

Thermal cracking resistance at low temperature. The creep compliance and strength test are conducted in accordance with AASHTO T 322. The test is conducted at three temperatures (i.e., -20, -10, and 0 °C) where a vertical static load is applied on the test specimen for 100 sec to produce a horizontal deformation between 0.00125 to 0.0190 mm to ensure that the test specimen is in the linear viscoelastic range. Once the creep compliance test is completed at all temperatures, the tensile strength test is conducted at a testing temperature of -10 °C by applying a vertical load rate of 12 mm/min until failure.

## Task 5: Analyze the laboratory testing results

Under this task, the researchers analyzed the laboratory testing results to evaluate the performance of the asphalt mixtures. The researchers examined the cracking and rutting performance as well as the low-temperature cracking of mixtures with different characteristics including RAP source and content, rejuvenator type and content, and binder grade and content. The results for mixtures prepared with RAP collected from four different sources (e.g., RAP Source No. 1 through RAP Source No. 4) were carefully discussed. The cracking resistance was evaluated for all mixtures using the IDT test and calculations of IDEAL-CT<sub>Index</sub>. The researchers calculated additional cracking parameters including IDT<sub>Strength</sub> and Weibull<sub>CRI</sub>. The Hamburg Wheel Tracking Test (HWTT) was conducted using the Asphalt Pavement Analyzer JR. (APA Jr.) to evaluate the rutting performance of the test mixtures. The effect of RAP content, rejuvenator type, and rejuvenator doses on the performance was examined for each RAP source separately. Furthermore, the researchers conducted statistical analysis of the test results using Tukey's Honestly Significant Difference (Tukey's HSD). The Tukey's HSD is a one-way analysis of variance (ANOVA) and is performed at 95 percent confidence interval (i.e.,  $\alpha = 0.05$ ). Tukey's HSD can identify test means with significant difference.

Task 5 also evaluated the cracking and rutting performance of loose mixtures collected from new paving projects. These loose mixtures were collected from different districts in Idaho and have different properties (mix design, binder grade, binder content, RAP, etc.). The researchers calculated different cracking and rutting performance parameters and compared to performance thresholds proposed and used in previous ITD studies (Kassem et al. 2019 and Kassem et al. 2021). Furthermore, the researchers examined the coefficient of variation of various cracking performance indicators of the field projects along with their correlations.

## Task 6: Conduct cost analysis to assess economic savings

Under this task, the researchers conducted cost analysis to assess economic savings associated with using rejuvenators with high RAP content in asphalt mixtures without compromising the performance (i.e., cracking and rutting resistance). In addition, the researchers examined the benefits and cost savings associated with increasing the binder content at different RAP contents. The use of RAP can cut down the percent of virgin binder added leading to cost savings. In addition, the current practice is to use one grade

softer virgin binder if the RAP content between 17 percent to 30 percent and the blending chart to select the grade of softer virgin binder if RAP content exceeds 30 percent. The use of softer binder costs more than the conventional virgin binder due to its limited availability. Therefore, the use of rejuvenators may eliminate or reduce the need for softer binders.

## Task 7: Develop recommendations and guidelines

Under this task, the researchers developed recommendations on incorporating RAP and rejuvenators in asphalt mixtures for improved performance and cost savings. Furthermore, the researchers provided recommendations on implementing a balanced mix design approach using the performance thresholds for cracking and rutting proposed in RP 261 and further evaluated in this study. Also, the researchers provided recommendations on future studies for implementation.

## Task 8: Prepare the final report

The researchers prepared a final report that includes the research methodology, results, analysis, findings, and recommendations.

## **Report Organization**

This report consists of seven chapters and five appendices. Chapter 1 provides an overview, objectives, tasks, and report organization. Chapter 2 provides a literature review of previous studies and summarizes the main findings. Chapter 3 discusses the material properties, testing matrix, and laboratory experimental design used and conducted in this study. Chapter 4 discusses the performance of asphalt mixtures prepared with different amounts of RAP and rejuvenators. Chapter 5 discusses the cracking and rutting performance of asphalt mixtures collected from various paving projects across the state based on performance thresholds developed in RP 261. Chapter 6 provides cost analysis to assess economic savings associated with using rejuvenators and high RAP content in asphalt mixtures. Chapter 7 provides a summary of the main findings of this study and associated recommendations. The appendices provide additional information and figures that were cited and discussed in this report.

# 2. Literature Review

## Introduction

Many State Departments of Transportation (DOTs) allow incorporation of RAP into asphalt mixtures. The use of RAP promotes and integrates sustainable solutions into civil engineering practices. The amount of virgin aggregates and asphalt binders can be reduced by using RAP in asphalt mixtures which offers environmental benefits and cost savings. The benefits of using RAP in asphalt mixtures include, but are not limited to, reducing production energy consumption and associated emissions, preserving natural resources, and reducing production cost (Yin et al., 2017). To ensure the pavement performance is not compromised due to the use of RAP, rejuvenators or recycling agents are often recommended in asphalt mixtures containing RAP. Recycling agents, also known as rejuvenators, are organic or petroleum products with chemical and physical properties that help to restore the rheological properties of aged asphalt binders. Rejuvenators were first introduced in 1960's as a pavement preservation practice. RAP binders are often aged and stiffer with less flexibility and ductility due to the oxidization process experienced by the pavement during service. The asphalt binder is often modeled as a colloidal material with two phases, asphaltene and maltene. The ratio of maltenes to asphaltenes is reduced due to oxidative aging, which results in stiffer asphalt pavements that are susceptible to cracking (Kaseer et al. 2019a). Rejuvenators contain maltenes to restore the original ratio of maltenes to asphaltenes which improves the flexibility and resistance of asphalt mixtures to cracking.

Recently, the pavement community and industry are moving towards a Balanced Mix Design (BMD) approach to complement current Superpave method for asphalt mixture design with improved performance (Meroni et al. 2020 and Lombardo et al. 2020). The current Superpave mix design is primarily based on volumetrics. As the pavement community gravitates towards incorporating more recycled materials, the Superpave process fails to consider the interactions between virgin materials and recycling agents (NCAT 2019). The goal of BMD is to design a mix that meets the specifications according to the purpose the pavement will serve. BMD includes specific tests to evaluate pavement performance against specific distresses, most commonly resistance to rutting and cracking. The goal of BMD is to define testing criteria that are simple, affordable, and accurate enough to ensure acceptable performance (Taylor and West 2019). BMD should accurately simulate realistic climate and aging conditions, loading expectations, temperature variation, and other factors.

## Rejuvenators

The American Standards for Testing and Materials (ASTM) developed a standard (ASTM D4552) that classifies recycling agents into six groups based on their viscosity at 60°C as shown in Table 1Table 1. In 2014, the National Center for Asphalt Technology (NCAT) divided the rejuvenators into five categories: 1) paraffinic oil (refined used lubricating oil), 2) aromatic extract (refined crude oil products with polar aromatic oil components), 3) naphthenic oils (engineered hydrocarbons for asphalt modification), 4) triglycerides and fatty acids (derived from vegetable oils), and 5) tall oils (byproducts from the paper industry) as shown in Table 2 (NCAT 2014).

		RA1	RA1	RA5	RA5	RA25	RA25	RA75	RA75	RA250	RA250	RA500	RA500
	ASTM												
Test	Test Method	Min.	Max.	Min.	Max.	Min.	Max.	Min.	Max.	Min.	Max.	Min.	Max.
Viscosity at 140°F, cSt	D 2170 or D2171	50	175	176	900	901	4500	4501	12500	12501	37500	37501	60000
Flash point, COC, °F	D92	425		425		425		425		425		425	
Saturates, wt%	D2007		30		30		30		30		30		30
Tests on													
residue from	D 2872 or				1								
RTFO or TFO	D 1754				1								
oven 325 °F					1								
Viscosity Ratio	-		3		3		3		3		3		3
Wt. change ± %	-		4		4		4		4		4		4
Specific gravity	D70 or D 1298	Report		Report		Report		Report		Report		Report	

Table 1. Physical Properties of Recycling Agents (ASTM D4552)

#### Table 2: Recycling Agent Types (NCAT 2014)

Category	Examples	Description
Paraffinic Oils	<ul> <li>Waste Engine Oil (WEO)</li> <li>Waste Engine Oil Bottoms (WEOB)</li> <li>Valero VP 165<sup>®</sup></li> <li>Storbit<sup>®</sup></li> </ul>	Refined used lubricating oils
Aromatic Extracts	<ul> <li>Hydrolene<sup>®</sup></li> <li>Reclamite<sup>®</sup></li> <li>Cyclogen L<sup>®</sup></li> <li>ValAro 130A<sup>®</sup></li> </ul>	Refined crude oil products with polar aromatic oil components
Nathenic Oils	<ul> <li>SonneWarmix RJ™</li> <li>Erogon Hyprene<sup>®</sup></li> </ul>	Engineered hydrocarbons for asphalt modification
Triglycerides & Fatty Acids	<ul> <li>Waste Vegetable Oil</li> <li>Waste Vegetable Grease</li> <li>Brown Grease</li> <li>Delta S*</li> </ul>	Derived from vegetable oils *Has other key chemical elements in addition to triglycerides and fatty acids.
Tall Oils	<ul> <li>Sylvaroad™ RP1000</li> <li>Hydrogreen<sup>®</sup></li> </ul>	Paper Industry byproducts Some chemical family as liquid antistrip agents and emulsifiers

#### Cracking Performance of Asphalt Mixtures with RAP and Rejuvenators

Many research studies examined the effect of recycling agents and rejuvenators on the performance of RAP mixtures. Zaumanis et al. (2013, 2015) evaluated the effect of different rejuvenators in restoring the

performance of 100 percent RAP mixtures. They examined included plant oils, waste-derived oils, engineered products, and traditional and nontraditional refinery base oil. They evaluated the low temperature cracking performance of the test mixtures using the creep compliance and tensile strength tests as shown in Figure 1 and Figure 2. The results demonstrated that all mixtures showed improvements in their low temperature cracking performance when compared to the control RAP mixture. The waste vegetable oil improved the creep compliance of the RAP mixture to that of the virgin mix (i.e., 0 percent RAP). The results of tensile strength, conducted at -10°C, showed that certain rejuvenators such as aromatic extract increased the strength when compared to the control RAP mixture (i.e., without rejuvenator) while others including waste engine oil reduced the tensile strength.





Figure 1. Effect of Rejuvenators on Creep Compliance and Tensile Strength at -10°C (Zaumanis et al. 2013)



Figure 2. Creep Compliance and Tensile Strength at -10 °C (Zaumanis et al. 2015)

Nabizadeh et al. (2017) studied the effect of various rejuvenators on asphalt mixtures and Fine Aggregate Matrix (FAM) with 65 percent RAP. The study evaluated three unique rejuvenators including petroleum-tech based, green-tech based, and agriculture-tech based rejuvenators. They compared the performance of asphalt mixtures and FAM prepared with the selected rejuvenators to that of the control mix. The control mix contained 65 percent RAP and 35 percent virgin aggregates. All mixes had nominal maximum aggregate size of 12.5 mm and used PG 64-34 binder. The FAM mixtures used the same gradation as the original asphalt mixtures without aggregates larger than 1.18 mm. The results presented in Figure 3 and Table 3 showed that the petroleum-tech based rejuvenator or soybean oil (CR3), green-tech based rejuvenator or tall oil (CR2). All mixtures exhibited better cracking resistance as compared to the control mix (C).



Figure 3. Effect of Different Recycling Agents on IDT Load-Displacement Curve (Nabizadeh et al. 2017)

Mixture ID	Upward Slope	Ipward Slope Downward Slope at Inflection Point		Flexibility Index (FI)
С	1.472	-1.11	942	8.5
CR1	0.593	-0.50	919	18.4
CR2	0.735	-0.63	685	10.9
CR3	1.276	-0.67	878	13.1

Table 3. Semi-Circular Bending Fracture Test Results (Nabizadeh et al. 2017)

Kaseer et al. (2020) evaluated the performance of asphalt mixtures with high RAP content and rejuvenators. They evaluated the performance of both field cores and laboratory mixes. The field mixtures were extracted from five different states in the United States to cover different climate and mixture properties. In addition, laboratory mixes were prepared using raw materials collected from the source of each field project. Each field project had two sections, one designed with the maximum allowed percentage of RAP without recycling agent per the DOT specifications of that state, and the other designed similarly with the addition of rejuvenator. In the case of Texas field and laboratory mixtures, they used a replacement binder ratio of 28 percent and two different recycling agents: tall oil at two different doses and one dose of aromatic extract. The results of this study indicated that for both Short-Term Oven Aged (STOA) and Long-Term Oven Aged (LTOA) the rejuvenated lab mixtures showed a better cracking performance as shown in in Figure 4 (Kaseer et al. 2020). However, the rejuvenated field cores exhibited a lower cracking resistance compared to the cores without recycling agents.



Figure 4. Effect of Rejuvenators on Flexibility Index and Cracking Resistance Index (Kaseer et al. 2020)

An ongoing research study by the National Road Research Alliance (NRRA) is currently evaluating the effect of rejuvenators on performance of asphalt mixtures. They examined the effect of seven different rejuvenators including petroleum-based and bio-based rejuvenators. They prepared and tested a total of ten mixtures. Nine mixtures had 40 percent RAP, and one mixture had 30 percent RAP. These mixtures

were prepared with different rejuvenators. They compared the performance of Reheated Plant Mixtures (RPM) to Long-Term Aging (LTA) mixtures. The preliminary results (Figure 5) showed that the IDEAL-CT<sub>Index</sub> decreased with the increase of aging. Therefore, the cracking resistance is expected to decrease with aging. Both RPM and LTA mixtures with bio-based rejuvenators (e.g., Soybean, Ingevity and Kraton) were able to provide better cracking resistance compared to the control mix (i.e., 40 percent RAP). In addition, it was found that certain rejuvenators had a negative impact on LTA mixtures as the cracking performance decreased compared to the control mixture after aging. They indicated that such mixtures were more susceptible to aging.



Figure 5. Effect of Rejuvenator and Aging on Cracking Performance (NRRA 2022)

Zaumanis et al. (2014) evaluated the rutting performance of rejuvenated asphalt mixtures using the HWTT. They examined six unique rejuvenators and compared the rutting performance of the virgin mixture and RAP mixture to that of the rejuvenated mixtures. The virgin mixture was prepared by burning the binder off RAP aggregates and then mixing the aggregates with a virgin binder. The results presented in Figure 6 demonstrated that the virgin mixture had the lowest rutting resistance. This could be attributed to higher binder content (5.94 percent), loss of fine materials during burning process, or moisture damage. Conversely, the RAP mixture had the highest rutting resistance due to the presence of the aged RAP binder. The results also showed that the use of rejuvenators helped to soften the materials which may resulted in increased rut depth; however, all mixtures with rejuvenators performed better that the virgin mixture and mixtures with aromatic extract, waste engine oil (WEO), and organic oil passed the rutting threshold (12.5 mm after 20,000 passes) (Zaumanis et al. 2014).



Figure 6. Effect of Rejuvenators on Rutting Performance (Zaumanis et al. 2014)

Bajaj et al. (2020) evaluated the rutting performance of asphalt mixtures prepared with six unique rejuvenators. The mixtures also included a control mix without RAP and one RAP mixture without rejuvenator. Figure 7 shows the rut depth with loading cycles. The results demonstrated that the RAP mixture without rejuvenators had the lowest rut depth which indicates better rutting resistance. All mixtures with rejuvenators passed the rutting criteria of 12.5 mm after 7500 loading cycles except the rejuvenated mixture with paraffinic recycling agent P (Bajaj et al. 2020).





Kaseer et al. (2020) also evaluated the rutting performance of asphalt mixtures with rejuvenators. Asphalt mixtures were prepared in the laboratory using materials from two different states: Wisconsin and Delaware. They evaluated the rutting performance of mixtures containing two percentages of RAP (i.e., 31 and 50 percent) and modified vegetable oil rejuvenator at 5.5 and 9 percent. Figure 8 shows the number of passes for the mixtures to reach a rut depth of 12.5 mm. The results demonstrated that the

addition of recycling agents increased the rutting of the mixture; however, all mixtures passed the rutting threshold (minimum of 5000 passes for binder Performance Grade PG 58-XX before reaching 12.5 mm). Also, they found that the test mixtures were susceptible to moisture damage. On the other hand, the APA test (conducted in dry conditions) showed that the rutting resistance improved compared to HWTT results with the use of rejuvenators.



Figure 8. HWTT and APA Rutting Performance of Rejuvenated Mixtures (Kaseer et al. 2020)

#### Effect of Rejuvenators on Binder Performance Grade (PG)

Zaumanis et al. (2014) studied the effect of incorporating rejuvenators into asphalt mixtures on the binder performance grade (PG) of the rejuvenated asphalt binders in the mix. They extracted and recovered asphalt binders from rejuvenated asphalt mixtures and determined their PG. The results shown in Figure 9 indicated that all rejuvenated mixtures had improved low-temperature PG (PGL) compared to the RAP binder (-12 °C). The target PG was reached in all rejuvenated mixtures except mixtures with WEO rejuvenator which required a higher dose to reach the targeted PGL. None of the examined rejuvenators decreased the high-temperature PG (PGH) as compared to that of the virgin binder (Zaumanis et al. 2014).

Ali et al. (2016) investigated the impact of rejuvenator on RAP mixtures. They used five unique rejuvenators in asphalt mixtures containing 25 and 45 percent RAP at the manufacturer's recommended doses. They examined the effect of the rejuvenators on the extracted and recovered binder PG. The results demonstrated that all rejuvenators lowered the PGH. In addition, the paraffinic oil was the most effective rejuvenator in lowering the binder PG to that of the virgin binder. Other studies (Samara et al. 2022 and NRRA 2022) evaluated the effect of rejuvenators on the binder PG. They evaluated asphalt mixtures with four unique rejuvenators including crude tall oil, modified vegetable oil, corn oil and aromatic extract oil. The results on the extracted and recovered binder showed that all rejuvenators lowered the PGH by at least one grade. Furthermore, all rejuvenators lowered the PGL. They also concluded that increasing the rejuvenator dose resulted in lower PGH.



Figure 9. Effect of Rejuvenators on Performance Grade (Zaumanis et al. 2014)

#### Effect of Rejuvenators on Asphalt Mixture Workability

Limited studies were conducted on the effect of rejuvenators on the workability of asphalt mixtures. Zaumanis et al. (2014) evaluated the workability of rejuvenated mixtures using different rejuvenators. They recorded the number of gyrations to reach 8 percent air voids. The results demonstrated that the virgin mixture had the highest workability while the RAP mixture had the lowest workability. The workability of all mixtures with rejuvenators was enhanced by the addition of rejuvenators as shown in Figure 10.

Another study evaluated the effect of rejuvenators on workability through characterizing binder properties. Majidifard et al. (2019) investigated the effect of rejuvenators on the workability of asphalt mixtures by measuring binder stiffness. They examined the workability of the binder extracted and recovered from rejuvenated mixtures with waste cooking oil. The study concluded that the waste cooking oil was able to reduce the stiffness of the RAP binder and thus improve the mixture workability.





#### **Rejuvenator Dose Selection**

Several studies have been conducted to establish a method to select a proper dose of rejuvenators. Arámbula-Mercado et al. (2017) evaluated the selection of proper recycling agent dose and incorporation method into asphalt mixtures with high RAP and Recycled Asphalt Shingles (RAS). In their study, they evaluated three methods for dose selection and their effect on the performance of asphalt mixtures with high RAP/RAS content. The three methods included 1) restoring PGL and verifying PGH, 2) achieving a temperature difference in Bending Beam Rheometer (BBR),  $\Delta T_c$  of -5 °C, and 3) restoring the PGH. For the first method, they used a PG 70-22 binder and determined an optimum dose of 4.5 percent of weight of total binder, which was found to meet the PGL and then verified the PGH as shown in Figure 11 (Arámbula-Mercado et al. 2017). The second method was to achieve a maximum  $\Delta T_c$  of -5 °C after 20 hours of PAV aging. They found that a dose of 12.5 percent was enough to reach the target  $\Delta T_c$  for PG 58-32 and 14.5 percent for PG 68-32. The third selection method suggested that an optimum rejuvenator dose of 6 percent to reach continuous the PGH (i.e., 70 °C).

The results of the three selection methods demonstrated that the first method improved the stiffness and phase angle, but such improvement decreased with long-term aging. The second method showed that the stiffness and phase angle were improved as well; however, the stiffness reduction was excessive and had a negative effect on the mixture rutting performance. The third method provided better results over the other two methods. The researchers demonstrated that the rejuvenator dose should be carefully selected as the rutting resistance could be reduced while the cracking resistance improved. The dose selection is affected by many factors such as binder source and grade, aging level of recycled materials, and their proportions in the mix.



Figure 11. Performance Grade blending Chart (Arámbula-Mercado et al. 2017)

Zaumanis et al. (2014) also proposed a procedure to determine the optimum rejuvenator dose based on Superpave PG specifications. They used extracted RAP binder with PG 94-12 which had penetration of 1.9 mm. Their target PG for the region was PG 64-22. Six rejuvenators were evaluated in their study. Two rejuvenators were petroleum products and the other four were organic products. The petroleum products were aromatic extract and waste engine oil. The organic products included waste vegetable oil, organic oil, waste vegetable grease, and distilled tall oil. They evaluated two doses for organic products (i.e., 6 percent and 12 percent) and two doses for petroleum products (i.e., 12 percent and 18 percent). They

conducted several binder tests including penetration test at 25°C, Dynamic Shear Rheometer (DSR), and Bending Beam Rheometer (BBR) on extracted and recovered asphalt binders. The results demonstrated that both PGH and PGL reduced linearly, while the penetration increased exponentially with the increase of the rejuvenation dose as shown in Figures 12 and 13. Overall, organic-based products required a lower dose than petroleum-based products.



Figure 12. High, Low, and intermediate PG for Rejuvenated Extracted Binder (Zaumanis et al. 2014)



Figure 13. Minimum Rejuvenator Dose (Zaumanis et al. 2014)

Sánchez et al. (2020) examined three methods to select the optimum dose of a palm oil rejuvenator. They conducted several tests including penetration, softening point, and PGH. The PGH resulted in the lowest dose, while the softening point required the highest dose. The optimum dose was 3, 4.5 and 5.5 percent based on the PGH, penetration, and softening point, respectively (Sánchez et al. 2020).

Other researchers evaluated the optimum rejuvenator dose through the balanced mix design approach. S. Im et al. (2016) examined dose selection for three recycling agents including 1) Hydrogreen, 2) Road Science, and 3) Arizona Chemical. They determined the rejuvenator dose range per binder weight, then determined the optimum asphalt content and optimized the content for each rejuvenator used in their study. First, they determined a range of rejuvenator content based on the binder PG and Glover-Rowe (G-R) parameter. Based on the binder test results, they established the rejuvenator dose range and validation table for the three recycling agents as well as the manufacturer recommended dose as presented in Table 4. The dose range based on the binder PG varied from about 0.7 to 4.8 percent for high and low temperatures, respectively as shown in Figure 14. They also found that the rejuvenator content determined by the G-R parameter had a lower range (i.e., 0.7 to 2.6 percent). Based on the validation table (Table 4), they determined optimum dose ranges of 1.7 to 4.8 percent, 2.6 to 3.6 percent, and 1.8 to 3.7 percent per weight of binder for Hydrogreen, Road Science, and Arizona Chemical, respectively. To confirm the optimum dose for each rejuvenator, the study conducted a balanced mix design approach by examining the rutting and cracking of asphalt mixtures within the dose range. The optimum dose of each rejuvenator was determined based on a maximum rut depth of 9.5 mm. The results demonstrated that optimum dose was 3.2, 2.2, and 3.0 percent for Hydrogreen, Road Science, and Arizona Chemical, respectively. The Hydrogreen rejuvenator required a higher dose compared to the other two rejuvenators (i.e., Road Science and Arizona Chemical).

-	Based on PG	Based on PG	Based on aging	Based on aging	Selected Range	Selected Range	Dose recommended by manufacturer
Rejuvenator	High	Low	Damage	Significant	Min	Max	-
	temp.	temp.	onset	Cracking			
			Min				
	max	Min		Min			
Hydrogreen	4.8%	1.7%	1.0%	1.7%	1.7%	4.8%	2.9%
Road Science	3.6%	0.7%	0.8%	2.6%	2.6%	3.6%	2.0%
Arizona Chemical	3.7%	1.1%	0.7%	1.8%	1.8%	3.7%	2.3%

Table 4. Rejuvenator Content Range and Dose Validation (S. Im e	et al. 2016)
---	--------------



#### Figure 14. High and Low Temperature Performance Grade for Rejuvenated Mixtures (S. Im et al. 2016)

#### **Economic Benefits of Rejuvenators in RAP Mixtures**

Many studies evaluated the economic benefits of incorporating rejuvenators along with high RAP content in asphalt mixtures. Martin et al. (2019) studied the likelihood of cost savings of using rejuvenators when RAP content increased from 20 to 40 percent. They considered two scenarios including low economic incentive and high economic incentive. The low economic scenario assumed that the recycling agent and RAP are relatively high in price and the amount of binder available in the RAP is low, while the high economic incentive assumed that the amount of binder available in the RAP is high and the price of recycling agent and RAP are relatively low. They found that increasing the RAP content from 20 to 40 percent resulted in 4.9 percent savings in the production cost for the low economic incentive and 17 percent savings for the high economic incentive assumption. In addition, they found that the cost saving associated with 40 percent RAP is about 12 and 35 percent for the low and high economic incentive, respectively.

S. Im et al. (2014) conducted cost analysis on the benefits of using rejuvenators with asphalt mixtures containing RAP content of 19 percent. They calculated the cost (per ton) of virgin mix (i.e., 0 percent RAP), RAP mixture containing 19 percent of RAP, and rejuvenated RAP mixture containing 19 percent RAP. The outcome of their study demonstrated that a ton of virgin mix cost \$47.2 while a ton of a similar mix containing 19 percent RAP cost \$38.9 which resulted in cost reduction of 21 percent. However, the rejuvenated RAP mixture roughly increased the cost of the RAP mix by \$0.3 per ton which resulted in cost reduction of 20 percent compared to the virgin mix. Zaumanis et al. (2014) evaluated the cost of asphalt mixtures prepared with and without RAP. The study indicated that the location of where the mixture was made derived the cost as some locations may or may not have RAP materials accessible where other locations may have extra RAP materials. Their study assumed that only the material cost will be impacted by rejuvenators and other costs would remain constant. The cost saving associated with using 100 percent RAP and rejuvenators resulted in cost reduction of about 50 to 70 percent as shown in Figure 15.



Figure 15. Cost Saving Potential Associated with Incorporating RAP and Rejuvenator Zaumanis et al. 2014)

## **Incorporating RAP in Asphalt Mixtures**

Transportation agencies incorporate RAP into asphalt mixtures for its environmental and economic benefits. A survey conducted by the National Asphalt Pavement Association (NAPA) demonstrated the favorable impact of RAP on decreasing the cost of pavement construction and greenhouse gas emissions (NAPA 2021). Figure 16 shows the amount of RAP and the average percentage of RAP used by different transportation agencies from 2009 to 2021. The average percent of RAP increased from 21.1 percent in 2019 to about 21.9 percent in 2021 for all industry sectors. Figure 17 shows the average percentage of RAP used in various states. The number of states using 20 percent RAP or grater increased to 32 states in 2021 compared to only 10 states in 2009. Idaho is among the leading states by allowing higher RAP content (up to 30 percent). Table 5 shows the amount of RAP mixes that incorporate softer binders and/or recycling agents in each state. Most states (28 out of 32) using 20 percent RAP or more reported using rejuvenator and/or softer binders in their asphalt mixtures. The rest of states (four) don't use any rejuvenators and/or softer binders in the asphalt mixtures with 20 percent or more RAP (NAPA 2021).



Figure 16. RAP Use by Sector (Million Tons) and Average Percent RAP Used by Sector (NAPA 2021)



Figure 17. Average Estimated RAP Content in Asphalt Mixes by State (NAPA 2021)

	Reported Tons	Reported Tons	Estimated Tons	Estimated Tons
State	Stockpiled (Million)	Stockpiled (Million)	Stockpiled (Million)	Stockpiled (Million)
	in 2020	in 2021	in 2020	in 2021
Alabama	0.76	1.26	2.13	1.62
Alaska	*	*	*	*
American Samoa	NCR	NCR	NCR	NCR
Arizona	0.58	1.02	1.02	2.13
Arkansas	0.45	0.32	0.93	0.7
California	2.08	0.99	4.33	2.92
Colorado	0.86	0.31	1.57	0.77
Connecticut	*	*	*	*
Delaware	*	*	*	*
District of Columbia	*	*	*	*
Florida	3.62	2.04	5.43	5.21
Georgia	3.31	2.25	6.07	5.03
Guam	NCR	NCR	NCR	NCR
Hawaii	*	0.13	*	0.24
Idaho	0.65	0.59	1.56	1.39
Illinois	2	1.16	3.43	2.39
Indiana	2.35	3.71	4.07	5.05
lowa	0.53	0.65	1.45	1.83
Kansas	0.79	0.8	1.15	1.31
Kentucky	0.58	0.96	1.36	1.98
Louisiana	0.05	0.21	0.35	1.02
Maine	0.37	0.29	0.37	0.29
Maryland	2,65	2.27	3.63	4.49
Massachusetts	0.69	0.92	1.56	3.67
Michigan	14.98	2,28	22.17	3.77
Minnesota	3.96	1.88	6.62	2
Mississippi	0.61	0.46	0.63	0.74
Missouri	0.42	0.46	1.76	1.47
Montana	*	*	*	*
Nebraska	0.12	*	0.9	*
Nevada	0.12	0.29	0.34	0.65
New Hampshire	0.32	0.29	0.46	03
New Jersey	2.3	9.59	4 69	26.89
New Mexico	*	*	*	*
New York	0.89	0.65	2 79	2 65
North Carolina	5.6	4 39	6.78	6 35
North Dakota	*	*	*	*
No. Mariana Isl	NCR	NCR	NCR	NCR
Ohio	3 81	3 09	5.81	3.46
Oklahoma	1.2	1 21	1 61	1 21
Oregon	0.94	0.69	2 3/	2.05
Pennsylvania	0.94	0.88	2.54	3 00
Puorto Pico	NCP	NCP	2.37	NCP
Puerto Kico Rhodo Island	*	*	*	*
South Carolina	1.42	1 69	2 11	1 69
South Dakota	*	*	5.11	1.08
Toppessoo	1 95	1.67	1 22	2 50
rennessee	2.85	1.07	4.32	2.59

Table 5. Percent of RAI	P Mixes Using Softer Binde	r and/or Recycling A	Agents by State (NAPA 2021)
-------------------------	----------------------------	----------------------	-----------------------------

State	Reported Tons Stockpiled (Million) in 2020	Reported Tons Stockpiled (Million) in 2021	Estimated Tons Stockpiled (Million) in 2020	Estimated Tons Stockpiled (Million) in 2021
Texas	0.99	2.14	6.69	11
U.S. Virgin Islands	NCR	NCR	NCR	NCR
Utah	0.46	1.08	0.52	1.45
Vermont	*	*	*	*
Virginia	2.56	2.37	3.68	4.15
Washington	1.15	0.73	1.22	0.98
West Virginia	0.34	0.36	4.33	0.65
Wisconsin	2.14	2.7	2.67	4
Wyoming	*	*	*	*
Total	71.48	59.82	135.3	137.45

Table 5. Percent of RAP Mixes Using Softer Binder and/or Recycling Agents by State (NAPA 2021)(Continued)

NCR No Companies Responding for the State Survey

\* Fewer than 3 Companies Reporting

Total Includes Values from State with Fewer than 3 Companies

Several researchers examined the amount of RAP that can be used in asphalt mixes without detrimental effects on performance (McDaniel et al. 2012; Beeson et al. 2011; Sondag et al. 2002). McDaniel et al. (2012) demonstrated that asphalt mixtures with up to 50 percent RAP can meet the Superpave design criteria. They reported that asphalt mixtures with more than 20 percent RAP increased the stiffness which could impact the cracking resistance at low temperature (McDaniel et al. 2012). Beeson et al. (2011) demonstrated that 22 percent of RAP can be used without changing the PGL of the tested binder. In addition, the virgin binder PG should be reduced by one grade for up to 40 percent RAP. On the other hand, Mogawer et al. (2020) demonstrated that using more than 15 percent RAP impacts the PGL significantly. Sondag et al. (2002) also documented that Iowa DOT requires the virgin binder PG to be reduced by one grade for RAP exceeding 20 percent. ITD limits the maximum RAP content to 30 percent. For mixtures with less than 17 percent RAP, no binder PG adjustment is required; however, either the blending chart is required or lowering the high and low designated temperatures by one grade for mixtures with more than 17 percent RAP. Tables 6 and 7 summarize the ITD specifications for the binder grade adjustment and typical adjusted binder grades (ITD 2018). ITD RP 280 provides additional information about incorporating RAP into asphalt mixtures including characterization of RAP, mix design considerations of RAP mixtures, amount of RAP in asphalt mixtures, laboratory, and field evaluation of mixes with high RAP contents, and state of practice at different DOTs (Kassem et al. 2021).
Level	RAP binder content by weight of the total binder in the mixture, percent	Binder Grade Adjustment to compensate for the stiffness of the asphalt binder in the RAP
1	0 to 17	No binder grade adjustment is made.
2	> 17 to 30	The selected binder grade adjustment for the binder grade specified on the plans is one grade lower for the high and the low temperatures designated. Or determine the asphalt binder grade adjustment using a blending chart. Note: See AASHTO M 323 for recommended blending chart procedure.

#### Table 6. Grade Adjustment for RAP (ITD Specification 2018)

#### Table 7. Typical Adjusted Binder Grades (ITD Specification 2018)

Binder Grade Specified in	Adjusted Binder Grade	Adjusted Binder Grade	
Contract	(Level2)	(Level1)	
58-28	58-34	No Adjustment Needed	
58-34	No Adjustment Needed	No Adjustment Needed	
64-28	58-34	No Adjustment Needed	
64-34	58-34	No Adjustment Needed	
70-28	64-34	No Adjustment Needed	
76-28	70-34	No Adjustment Needed	

# **Balanced Mix Design Approach**

Zhou et al. (2006) proposed a method to improve the mix design of asphalt mixtures that relies on balancing the cracking and rutting performance, which is known as a balanced or engineered mix design (BMD). Asphalt mixtures prepared with low binder content have better resistance to rutting; however, they are prone to cracking. On the other hand, asphalt mixtures prepared with high binder content have better resistance to cracking; however, they may be prone to rutting. Therefore, a procedure was introduced for determining the optimum binder content that would balance rutting and cracking

resistance (Zhou and Scullion 2006). This procedure is proposed for asphalt mixtures used in wearing courses (i.e., surface layer). Figure 18 shows an example of balancing rutting and cracking resistance.

The most common tests used in BMD to assess cracking performance are Indirect Tensile Asphalt Cracking Test (IDEAL-CT) and the Semi-Circular Bending Illinois Flexibility Index Test (SCB-IFIT), while the Hamburg Wheel Tracking Test (HWTT) and Asphalt Pavement Analyzer (APA) are used to evaluate the rutting performance (Kassem at al. 2019). Other tests used in BMD also include Texas Overlay to assess cracking performance. Based on previous ITD research studies (Kassem et al. 2019 and Kassem et al. 2021), this study used IDEAL-CT and HWTT to assess the cracking and rutting performance, respectively. In addition, HWTT can be used to assess moisture susceptibility of the asphalt mixtures since it is conducted in wet conditions (Kassem et el. 2019).

ITD RP 261 proposed three performance thresholds for the IDEAL-CT: good cracking resistance (IDEAL-CT > 73.7), fair cracking resistance ( $26.4 \le IDEAL-CT \le 73.7$ ), and poor cracking resistance (IDEAL-CT < 26.4). The proposed thresholds for some performance indicators were comparable to the ones proposed by other researchers (Kassem et al. 2019). In addition, ITD RP 261 proposed a maximum rut depth of 10 mm for HWTT after 15,000 passes or 12.5 mm for HWTT after 20,000 passes to ensure adequate resistance to rutting. Also, the HWTT thresholds can be used to ensure adequate resistance to moisture damage. Similar thresholds were also used and adopted by several transportation agencies (Kassem et al. 2019).

Table 8 summarizes the most promising intermediate temperature cracking performance indicators and their associate testing standards (Alkuime et al. 2020). ITD RP 261 provides comprehensive review of cracking and rutting tests used by various states as well as various performance indicators and proposed thresholds (Kassem et al. 2019). The following section provides an overview of cracking and rutting tests selected in this study and proposed thresholds from the literature.



Figure 18. An example of balancing rutting and cracking resistance (Zhou et al. 2006)

 Table 8. Intermediate temperature cracking most promising performance indicators and its associate testing standards after (Alkuime et al. 2020)

No.	Name	Symbol	Equation	Standard No.
1	Total Fracture Energy	G <sub>fracture</sub>	$G_{Fracture}^{Total} = rac{w_{Fracture}^{Total}}{crackfacearea}$	ASTM-D6931-07 & ASTM D8225-19
2	Cracking Resistance Index	CRI	$CRI = \frac{G_{Fracture}^{Total}}{P_{peak}}$	ASTM-D6931-07 & ASTM D8225-19
3	Flexibility Index	FI	$FI = \frac{G_{Fracture}^{Total}}{\left m_{inflection}^{Post-Peak}\right }$	ASTM-D6931-07 & ASTM D8225-19
4	IDEAL-CT <sub>Index</sub>	IDEAL- CT <sub>Index</sub>	$IDEAL - CT_{Index} = \frac{G_{Fracture}^{Total}}{\left m_{75\%Post-peak}\right } \times \frac{t}{62} \times \varepsilon_{v \ tolerance}$	ASTM-D6931-07 & ASTM D8225-19
5	N <sub>flex</sub> Factor	N <sub>flex</sub>	$N_{flex} = \frac{Toughness}{m_{inflection}^{Post-Peak}}$	ASTM-D6931-07 & ASTM D8225-19
6	IDT Modulus	IDT <sub>Modulus</sub>	$IDT_{Modulus} = \frac{\sigma_{Tensile}^{IDT}}{L_{Peak\ Load}}$	ASTM-D6931-07 & ASTM D8225-19
7	Weibull <sub>CRI</sub>	Weibull <sub>CRI</sub>	$Weibull_{CRI} = \left(\frac{\eta}{\beta}\right)  imes \log[A]$	ASTM-D6931-07 & ASTM D8225-19

## Indirect Tensile Asphalt Cracking Test

#### IDEAL-Cracking Test Index [IDEAL-CT<sub>Index</sub>]

IDEAL-CT<sub>Index</sub> is a cracking resistance index recently developed by Zhou et al. (2017). IDEAL-CT<sub>Index</sub> utilizes the load-displacement curve obtained from the IDT test. This index is a function of the total fracture energy, 75 percent post-peak slope m<sub>75</sub>, and the strain tolerance parameter (Equation 1). This index is measured in accordance with ASTM D8225-19 "Determination of Cracking Tolerance Index of Asphalt Mixture Using the Indirect Tensile Cracking Test at Intermediate Temperature". Several researchers evaluated this index as a cracking resistance measure. Zhou et al. (2017) found the index to be sensitive to the key components of asphalt mixtures such as, but not limited to, binder grade, RAP content, binder content, and air voids. Dong et al. (2019) found the IDEAL-CT<sub>Index</sub> to be sensitive to binder content and the use of emulsion. Bennert et. al (2018) evaluated the index as a quality control tool in New Jersey, and they found that IDEAL-CT<sub>Index</sub> correlated very well with overlay tester (OT) lower variability.

$$IDEAL - CT_{Index} = \frac{G_{Total \, Fracture}}{\left| m_{75\% \, Post - peak} \right|} \times \frac{t}{62} \times \varepsilon_{v \, tolerance} \tag{1}$$

where:

IDEAL-CT <sub>Index</sub>	= Cracking test index
G <sub>Total Fracture</sub>	= Total fracture energy (J/m <sup>2</sup> )
M75% Post-peak	= Post-peak slope at 75 percent of the peak load
t	= Specimen thickness (mm)
E <sub>v tolerance</sub>	= Strain tolerance

Since the development of IDEAL-CT<sub>Index</sub>, several researchers and DOTs have proposed thresholds for cracking performance. Zhou et al. (2020) proposed an IDEAL-CT<sub>Index</sub> criteria based on a strong correlation with the OT. They proposed a minimum criterion of 90 for Superpave mixes, 55 for TxDOT dense-graded mixes, and 135 for stone matrix asphalt mixtures (SMA). West et al. (2021) suggested a minimum threshold of 40 and 80 for Superpave, SMA mixes, respectively. ARDOT adopted a minimum preliminary threshold of 50. Kassem et al. (2019) proposed three performance thresholds for the IDEAL-CT: good cracking resistance (IDEAL-CT > 73.7), fair cracking resistance ( $26.4 \le IDEAL-CT \le 73.7$ ), and poor cracking resistance (IDEAL-CT < 26.4). Diefenderfer et al. (2019) proposed CT<sub>Index</sub> of 80 as an initial minimum criterion for VDOT. Similarly Cross et al. (2019) proposed a minimum threshold of 80 for Oklahoma mixes. Yin et al. (2021) demonstrated that TDOT considers a minimum of 100 for interstate and controlled access

state routes and a minimum of 50 for non-controlled access state routes with less than 10,000 Average Daily Traffic (ADT) and 75 for an ADT greater than 10,000.

#### Weibull<sub>CRI</sub>

The Weibull<sub>CRI</sub> is another cracking resistance indicator that was developed in RP 261 (Kassem et al. 2019). Similar to the IDEAL-CT<sub>Index</sub>, this indicator utilizes the IDT load-displacement curve. However, the Weibull<sub>CRI</sub> describes the entire load-displacement curve, while other monotonic cracking indicators (e.g., IDEAL- $CT_{Index}$ ) uses one or more elements of the load-displacement curve. The Weibull<sub>CRI</sub> uses the Weibull probability density function's fitting parameters to calculate the cracking resistance index. The Weibull<sub>CRI</sub> is calculated using Equation 2. Kassem et al. (2019) found Weibull<sub>CRI</sub> to have less variability in the test results compared to IDEAL- $CT_{Index}$ .

Weibull<sub>CRI</sub> = 
$$\left(\frac{\eta}{\beta}\right) \times \log(A)$$
 (2)

where:

η	= Scale parameter
β	= Shape parameter (Weibull slope)
A	= Scaling factor equals to the area under the load-displacement curve

## IDT Strength

The  $IDT_{strength}$  is also another performance indicator that is used in the literature (Kassem et al. 2019). This indicator is calculated by dividing the IDT peak load by the specimen geometry. The  $IDT_{strength}$  is calculated using Equation 3. Additional cracking resistance indices are described in Table 8.

$$IDT_{strenght} = \frac{2000 \times P_{Peak}}{\pi \times t \times D}$$
(3)

where:

<b>IDT</b> <sub>strength</sub>	= Tensile strength (kPa) determined from IDT test
P <sub>Peak</sub>	= Peak load (N)
t	= Specimen thickness (mm)
D	= Specimen diameter (mm)

#### **Creep-Compliance and Strength Test as low Temperature**

The creep-compliance and strength test are two parameters used to evaluate the low-temperature cracking performance (thermal cracking) of asphalt mixtures. The test is conducted in accordance with AASHTO T322 "Standard Method of Test for Determining the Creep Compliance and Strength of Hot Mix Asphalt (HMA) Using the Indirect Tensile Test Device". The creep compliance is calculated using Equation 4. Further discussion of the test is provided in Chapter 3 of this report. This test was used in several studies to evaluate the low-temperature cracking performance of asphalt mixtures. Krcmarik et al. (2016) demonstrated that mixtures with stiffer binder resulted in a higher low-temperature IDT strength which indicates higher thermal cracking susceptibility. Another study by Zaumanis et al. (2015), concluded that mixtures with higher creep-compliance and lower thermal tensile strength tend to have better cracking resistance than mixtures with lower creep and higher strength.

$$D(t)_{avg} = \frac{\Delta X_{tm} \times D_{avg} \times b_{avg}}{P_{avg} \times GL} \times C_{Cmpl}$$

$$C_{Cmpl} = 0.6354 \times \left(\frac{X}{Y}\right)^{-1} - 0.332$$
(4)

D(t)	= Creep compliance at time t (kPa)
GL	= Gauge length in (mm)
D <sub>avg</sub>	= Average diameter of the test specimen (mm)
B <sub>avg</sub>	= Average thickness of the test specimen (mm)
Pavg	= Average creep load (kN)
$\Delta X_{tm}$	= Trimmed mean of the normalized horizontal deformations
(X/Y)	= Absolute value normalized trimmed mean of the horizontal deformation ratio

#### Hamburg Wheel-Tracking Test

The Hamburg Wheel-Tracking Test (HWTT) is used to assess the rutting performance as well as moisture susceptibility of asphalt mixtures. The HWTT has steel wheel rollers that move back and forth over cylindrical asphalt specimens. The wheels are 17 mm wide and apply 705 N force. The test samples are submerged in a water bath at a controlled temperature of 50 °C. The rut depth is measured along the

roller path during the test, and generally the test is performed for 20,000 passes. The test is conducted in accordance with AASHTO T324 "Standard Method of Test for Hamburg Wheel-Track Testing of Compacted Asphalt Mixtures". Table 9 provides information on the rutting performance standards along with HWTT rutting performance thresholds established by various DOTs including Texas Department of Transportation (TxDOT), Washington State Department of Transportation (WSDOT), Colorado Department of Transportation (CODOT), Louisiana Department of Transportation (LADOT), and Montana Department of Transportation (MTDOT). ITD RP 261 proposed a maximum rut depth of 10 mm for HWTT after 15,000 passes or 12.5 mm for HWTT after 20,000 passes to ensure adequate resistance to rutting. Also, the HWTT thresholds can be used to ensure adequate resistance to moisture damage. In addition, other HWTT parameters such as Stripping Inflection Point (SIP) can also be calculated to evaluate moisture damage (Yin et al. 2014).



Figure 19. Hamburg wheel tracking test setup (Kassem et al. 2011)



Figure 20. APA Jr. for Hamburg wheel tracking test

DOT	Test procedure	PG grading/Mixture Type	Rutting performance threshold (minimum # of passes)
TXDOT	Tex-242-F	-	Limits, @12.5mm rut depth tested at 50 $^\circ \! C$
TXDOT	Tex-242-F	<=PG 64	10000
TXDOT	Tex-242-F	PG 70	15000
TXDOT	Tex-242-F	=>PG 76	20000
WSDOT	AASHTO T 324	-	15,000 Passes @10 mm rut depth tested at 50 $^\circ\!C$
CODOT	CP-L 5112	-	10,000 @ 4 mm rut depth tested at 50 $^\circ\!C$
LADOT	AASHTO T 324	Incidental Paving and ATB	10 mm @ 10,000 passes tested at 50 $^\circ\!\mathcal{C}$
LADOT	AASHTO T 324	Wearing and Binder Course level 1	10 mm @ 20,000 passes tested at 50 $^\circ\!C$
LADOT	AASHTO T 324	Wearing and Binder Course level 2	6 mm @ 20,000 passes tested at 50 $^\circ\!C$
MTDOT	MT 334-14	Mix design	13 mm @ 10,000 passes
MTDOT	MT 334-14	Plant	13 mm @ 15,000 passes
ITD	AASHTO T 324	-	12.5 mm @ 20,000 passes tested at 50 °C

Table 9. HWTT rutting performance threshold (after: Kassem at al. 2019)

# 3. Material Description and Experimental Design

Chapter 3 provides information about the testing materials (e.g., RAP, virgin aggregates, binders, rejuvenators). In addition, it documents the methods and procedures used by the researchers to evaluate the performance of the asphalt mixtures including intermediate temperature cracking performance, thermal cracking performance, rutting performance, and moisture susceptibility. Also, it discusses the preparation of test laboratory mixtures, loose mixtures obtained from the field, as well as the testing program.

# **Material Description**

## **Reclaimed Asphalt Pavement**

The researchers obtained RAP materials from four different sources and acquired virgin aggregates from two sources. The first source of RAP (i.e., RAP No. 1) was an "artificial RAP". This RAP was prepared in the lab using loose mixtures obtained from an ITD paving project that is described later in this section. The second source of RAP (i.e., RAP No. 2) and third source of RAP (i.e., RAP No. 3) were obtained from two different asphalt plants in Lewiston, Idaho. The last source of RAP (i.e., RAP No. 4) was obtained from an asphalt plant in Boise, Idaho. The first source of virgin aggregate was river gravel obtained from a quarry in Lewiston, Idaho.

The first source of RAP (i.e., RAP No. 1) was an "artificial RAP" that was aged in the laboratory. The mixture was obtained from a paving project in District 2, and the materials were sampled and delivered by ITD to the laboratory at University of Idaho in 50-lb boxes. The Nominal Maximum Aggregate Size (NMAS) of this mixture was 12.5 mm, the binder grade was PG 64-28, and the binder content was 5.3 percent. The loose mixture was placed in an oven at 135 °C for 3 days to simulate long-term field aging. This method was used by researchers in a previous study and was found to simulate field aging (Sirin et al. 2020). Figure 21 shows the RAP gradations for all examined RAP (i.e., RAP No. 1 through RAP No. 4).

Figure 22 shows the RAP binder content from different sources. RAP No. 2 and RAP No. 3 had an asphalt content of 5.37 percent and 4.3 percent, respectively. The first three sources of RAP materials (RAP No. 1, 2, and 3) were used in the laboratory testing at the University of Idaho. RAP No. 2 and 3 were fractionated into two different sizes: coarse (i.e., retained on Sieve No. 4) and fine (i.e., passing Sieve No. 4). RAP No. 4 had an asphalt content of 5.3 percent, and similar to RAP No. 2 and RAP No. 3, it was fractionated into coarse and fine piles. RAP No. 4 was used in the laboratory evaluation at Boise State University. Appendix A provides more information about the RAP materials of different sources acquired and used in this study.



Figure 21. RAP Gradations





## Rejuvenators

In this project, seven unique commercially available rejuvenators were used. Five rejuvenators (R1 through R5) were used at the University of Idaho (UI) and the other two rejuvenators (i.e., R6 and R7) were used at Boise State University (BSU). Figures 23 and 24 show the examined rejuvenators in this study.

The first rejuvenator (R1) is categorized as a tall oil. This rejuvenator is derived from a crude tall oil which is a by-product of the paper industry and tends to have a clear yellowish color. The first rejuvenator (R1) was obtained from a company in the United Kingdom. The second rejuvenator (R2) is categorized as an aromatic extract which is a refined crude oil product obtained from a company in the U.S. This rejuvenator is a dark brown color with a petroleum odor. The third rejuvenator (R3) is considered as a bio-based forestry product which was also obtained from another company in the U.S. This rejuvenator tends to have a dark brown color. The fourth rejuvenator (R4) is an engineered product by a company in the U.S. This rejuvenator tends to have a dark brown color. The fourth rejuvenator (R4) is an engineered product by a company in the U.S. This rejuvenator tends to have a dark brown color with a petroleum odor too. The fifth rejuvenator (R5) is considered as a triglycerides and fatty acids rejuvenator, and it is derived from waste vegetable oil obtained from a company in the U.S. This rejuvenator tends to have a obtained from a company in the U.S. too. The sixth (R6) and seventh (R7) rejuvenators were obtained from a company in the U.S. too. The sixth rejuvenator (R6) is a bio-based oil, and it has an orange hue oil. The seventh rejuvenator (R7) is a petroleum-based oil and has a clear liquid look as shown in Figure 24. Table 10 summarizes the examined doses of various rejuvenators. These doses were within the recommended values by the manufacturers.



Figure 23. Rejuvenators evaluated at UI



Figure 24. Rejuvenators evaluated at BSU

<b>Rejuvenator No.</b>	<b>Examined Doses</b>	Dose Description
R1	3.5%, 5%, and 7%	By weight of total binder
R2	6%, 10%, and 12%	By weight of reclaimed binder
R3	12.5% and 15%	By weight of reclaimed binder
R4	1% and 2%	By weight of RAP
R5	12% and 16%	By weight of reclaimed binder
R6	6.6% and 8.3%*	By weight of reclaimed binder
R7	11.3%+	By weight of reclaimed binder

#### Table 10. Rejuvenator Test Doses

Notes:

The doses of R6 and R7 were further checked using the blending charts, see Appendix B <sup>+</sup>Only one dose was evaluated.

\*Only 8.3 percent was used to evaluate both cracking and rutting performance, while 6.6 percent was used to evaluate the cracking performance

#### Laboratory-Mixed Laboratory-Compacted (LMLC) Test Specimens

The researchers prepared Laboratory-Mixed, Laboratory-Compacted (LMLC) asphalt mixtures with different RAP materials and rejuvenators. The laboratory experiments at UI included three different sources of RAP, five rejuvenators, three binder grades, two binder contents, different RAP contents, and different rejuvenator doses as discussed in detail later in this section. Similarly, the team at BSU prepared LMLC with one source of RAP, two rejuvenators, two binder grades, one binder content, different RAP contents, and different rejuvenator doses as discussed also later in this section.

The LMLC mixtures were prepared and tested to evaluate the cracking and rutting performance of asphalt mixtures with RAP and rejuvenators. The testing matrix at UI included LMLC mixes that were prepared using three different sources of RAP (i.e., RAP No. 1, RAP No. 2, and RAP No. 3). Each mixture was prepared using different RAP content. For RAP No. 1, different RAP contents were included (i.e., 0, 25, 50, 75 percent), while mixtures with RAP No. 2 examined the RAP contents of 0, 25, 50, 70 percent. Mixtures with RAP No. 3 included higher RAP content of 70 percent and control mix (i.e., 0 percent RAP). Furthermore, different rejuvenator doses were examined at each RAP content. Table 11 summarizes the testing matrix at UI, and Chapter 4 provides detailed information and performance results on examined mixtures. The mix designs of asphalt mixtures prepared and tested in this study are provided in Appendix B.

Similarly, the testing matrix at BSU included LMLC mixes that were prepared using one source of RAP. Each mixture was prepared using different RAP contents (i.e., 0, 25, 50, and 70 percent). In addition, the mixtures were prepared using two different rejuvenators at two different doses. Table 12 summarizes the testing matrix at BSU. The mixing and compaction temperatures for the LMLC specimens were obtained from the job mix formula. Prior to compaction, all mixes were short-term aged for four hours at a temperature of 135 °C in accordance with AASHTO R30-02. The LMLC specimens were compacted using a SuperPave Gyratory compactor to a target air void of 7±0.5 percent.

RAP %	0	25	50	≥70	-
RAP Source	1	2	3	-	-
Air Void %	7%	-	-	-	-
Binder Grade	PG 70-28	PG 64-28	PG 58-34	PG 58-28*	-
Binder Content %	OBC	OBC+0.5%	-	-	-
Rejuvenators	R1	R2	R3	R4	R5

Table 11. Laboratory-Mixed, Laboratory-Compacted Testing Matrix at UI

\*Only used with the 3<sup>rd</sup> source of RAP.

RAP %	0	25	50	70	
RAP Source	4	-	-	-	
Air Void %	7%	-	-	-	
Binder Grade	PG 70-28	PG 64-34	-	-	
Binder Content %	OBC	-	-	-	
Rejuvenators	R6	R7	-	-	

## Plant-Mixed Laboratory-Compacted (PMLC) Test Specimens

In this project, the researchers prepared Plant-Mixed Laboratory-Compacted (PMLC) test specimens obtained from new ITD paving projects. Loose asphalt mixtures from 23 projects were obtained and delivered to UI to evaluate the cracking and rutting performance of asphalt mixtures currently produced

and used in the state. These projects were distributed across all six districts of the state (District 1 to District 6). The loose mixtures were delivered in boxes, and each box weighed 50 lbs. Also, the boxes received were labeled with the necessary information and mix composition (e.g., binder content, project number, compaction temperature, etc.). The main properties of PMLC are summarized in Table 13, which include mix design, binder grade, RAP content, and RAP binder replacement. These mixtures included two different mix types (SP3 and SP5), four different virgin binder grades (PG 58-34, PG 64-28, PG 64-34, and PG 70-28), different percent of RAP content (ranging from 0 to 35 percent by weight of the mix), and 10 different percent of RAP binder replacement (e.g., 8, 15, 17, 20, 22, 25.4, 26.3, 27.3, 29.3, and 29.7 percent). Appendix D provides the mix design for the loose mixtures evaluated in this study. The PMLC specimen were reheated at the compaction temperature specified in the job mix formula for two hours and shoveled to avoid segregation. All PMLC specimens were compacted at a target air void of 7±0.5 percent using a SuperPave gyratory compactor.

#	District	Mix Type	Specified Binder PG	Virgin Binder PG	Binder Content Pb (%)	RAP (%)	RAP Binder Replacement (RBR %)	NMAS mm	Theoretical Specific Gravity (G <sub>mm</sub> )	Dust Proportion	Effective Binder Content P <sub>be</sub>	Project Key No.
1	D5 &6	SP-3	PG 58-34	PG 58-34	5	30	29.4	19	2.423	1.2	4.44	20590
2	D4	SP-3	PG 70-28	PG 70-28	6.2	0	NA	9.5	2.377	1.4	5.12	19863
3	D4	SP-3	PG 70-28	PG 64-34	5.6	30	26.3	12.5	2.396	1.1	4.8	18737
4	D4	SP-3	PG 64-34	NA	5.8	17	15	19	2.420	1.2	4.56	19404
5	D4	SP-2	PG58-28	PG58-34	5.7	35	29.7	12.5	2.402	1.2	4.8	19312
6	D4	SP-3	PG 64-34	NA	5.8	17	15	19	2.409	1.2	4.56	19699
7	D4	SP-3	PG70-28	PG 64-34	5.6	30	26.3	12.5	2.402	1.1	4.8	20180
8	D4	SP-3	PG 64-34	NA	5.4	NA	NA	12.5	2.392	1.3	4.74	19130
9	D3	SP3	PG 64-34	PG 58-34	5.5	20	20	12.5	2.497	1.3	4.75	20508
10	D2	SP-3	PG 64-28	PG 58-34	5.7	30	25	12.5	2.491	1.4	4.73	20436
11	D3	SP-3	PG 70-28	PG 64-34	5.33	31	29.3	12.5	2.418	0.8	4.7	13932
12	D3	SP-3	PG 64-34	PG 58-34	5.34	31	29.3	12.5	2.382	1.2	4.53	20714
13	D6	SP-5	PG 64-34	PG 58-34	4.9	29	NA	12.5	2.458	0.9	4.5	19812
14	D3	SP-3	PG 64-28	PG 58-34	5.3	32	NA	12.5	2.441	1.4	4.33	19112
15	D2	SP-3	PG 64-28	PG 58-34	5.2	30	25.4	12.5	2.602	1.4	4.2	19261
16	D2	SP-3	PG 64-28	NA	5.6	10	8	12.5	2.555	1.4	4.66	20193
17	D4	SP-5	NA	NA	5.2	NA	NA	12.5	2.285	NA	NA	20559
18	D4	SP-5	PG 70-28	NA	5.7	19	17	19	2.298	1.3	4.44	19086
19	D5	SP-3	PG 64-34	PG 58-34	5	20	27.3	19	2.466	1.3	3.94	20051
20	D2	SP-3	PG 70-28	NA	5.9	NA	NA	12.5	2.536	1.4	4.76	19373
21	D4	SP-3	PG 64-34	NA	5.8	17%	15	19	2.407	1.2	4.56	18742
22	D5	SP-3	PG 58-34	NA	4.8	NA	22	19	2.467	0.96	3.96	20051
23	*								2.423			20003

Table 13. PMLC Properties

<sup>\*</sup>Note: Project No. 23 (key 20003) was the only one without proper identification

# **Laboratory Testing**

The researchers conducted various laboratory tests to evaluate the cracking and rutting performance of the test mixtures. The cracking tests included the Indirect Tensile (IDT) Strength Test in accordance with ASTM D8225 *"Standard Test Method for Determination of Cracking Tolerance Index of Asphalt Mixture Using the Indirect Tensile Cracking Test at Intermediate Temperature"* to evaluate the intermediate cracking resistance. In addition, the researchers evaluate the low temperature cracking performance of selected mixtures in accordance with AASHTO T 322 *"Standard Method of Test for Determining the Creep Compliance and Strength of Hot Mix Asphalt (HMA) Using the Indirect Tensile Test Device"*. Furthermore, the team examined the rutting performance of the test mixtures in accordance with AASHTO T 324 *"Standard Method of Test for Hamburg Wheel-Track Testing of Compacted Asphalt Mixtures"*. This section describes the laboratory tests conducted in this study.

## **Evaluation of Cracking Performance**

The researchers evaluated the intermediate temperature cracking performance of the test asphalt mixtures using the Indirect Tensile (IDT) Strength Test in accordance with ASTM D8225. The IDT was conducted at UI using a servo-hydraulic Material Testing System (MTS-810) as shown in Figure 25. The test specimens were placed inside an environmental chamber at a temperature of 25 °C for two hours for condition before testing. The IDT conducted at BSU used a counter-top IDT apparatus at room temperature. The IDT test specimens are 6 inches (150 mm) in diameter and 2.45 inches (62 mm) thick and compacted to have 7 plus/minus 0.5 percent air voids. The IDT test is conducted at a constant compressive axial loading rate of 2 inches per minute (50 plus/minus 5 mm per minute) until failure. Figure 25 shows the test setup at UI.



Figure 25. IDT Test Setup

The researchers evaluated low temperature cracking (i.e., thermal cracking) of selected asphalt mixtures using the creep-compliance and strength test in accordance with AASHTO T 322. The creep compliance test was conducted using a servo-hydraulic Material Testing System equipped with an environmental chamber as shown in Figure 26b. The creep-compliance test specimens are 6 inches (150 mm) in diameter and 1.7 inches (43 mm) thick and compacted to have 7 plus/minus 0.5 percent air voids (Figure 26). The test is conducted at three different temperatures (i.e., -20, -10, and 0 °C), and the specimens are conditioned at the test temperatures for three hours before testing. In this test, a vertical static load is applied on the test specimen for 100 sec to produce a horizontal deformation between 0.00125 to 0.0190 mm to ensure that the test specimen is in the linear viscoelastic range. Both horizontal and vertical deformations are recorded during the test. Once the creep compliance test is performed at the middle testing temperatures, the tensile strength test is conducted. The tensile strength test is performed at the middle testing temperature (i.e., -10 °C) by applying a vertical load rate of 12 mm/min until failure.



Figure 26. Creep-Compliance Test specimen

## **Evaluation of Rutting Performance**

The researchers evaluated the rutting performance of the mixtures by conducting the Hamburg Wheel Tracking Test (HWTT) in accordance with AASHTO T324. Figure 27 shows the Asphalt Pavement Analyzer Jr. (APA Jr.) at UI. Four test specimens were used from each mixture for the HWTT testing. The HWTT test specimens are 6 inches (150 mm) in diameter and 2.36 inches (60 mm) thick and compacted to have 7 plus/minus 0.5 percent air voids. The specimens were sawed and placed in the testing molds as shown in

Figure 27a. The specimens were conditioned in a water bath at a temperature of 50 °C for 30 minutes before the test started. The HWTT wheels apply 705 N load directly on the surface of the test specimens at a constant rate of 52 pass/minute. The test is terminated after 20,000 passes or after an average rut depth of 12.5 mm is achieved. The rut depth is measured at 11 different locations along the pass of the HWTT wheels on the test specimens.



Figure 27. APA Jr. and HWTT Test Specimen and Setup

# 4. Evaluation of the Performance of Rejuvenators Incorporated in Mixtures with High RAP Content

This chapter presents and discusses the results of the laboratory testing program conducted to evaluate the cracking and rutting performance as well as the low-temperature cracking of mixtures with different characteristics including RAP source and content, rejuvenator type and content, and binder grade and content. The cracking resistance was evaluated for all mixtures using the IDT test and calculations of IDEAL-CT<sub>index</sub>. The IDEAL-CT<sub>index</sub> is proposed by Zhou et al. (2017) and is calculated from the IDT load-displacement curve as discussed in Chapter 2 using Equation 1. The IDEAL-CT<sub>index</sub> is a function of the fracture energy (Gf) and the post peak slope (m<sub>75</sub>) at which the load after the peak equals to three quarters of the maximum load. The IDEAL-CT<sub>index</sub> is an indicator that measures the resistance of asphalt mixtures to intermediate-temperature cracking. The higher the IDEAL-CT<sub>index</sub> the higher the resistance to cracking. In addition, the researchers evaluated the low-temperature cracking performance of mixtures at low temperatures. The test is conducted in accordance with AASHTO T 322 as discussed in Chapter 2 using Equation 4. Safi et al. (2018) concluded that the lower the compliance the stiffer the mix and thus more prone to thermal cracking and vice versa. At least three replicates were tested for each cracking test (i.e., IDT and creep compliance) to minimize the variation of the test results.

The rutting performance was evaluated in this study using Hamburg Wheel Tracking Test (HWTT) conducted using the Asphalt Pavement Analyzer JR. (APA Jr.). The rut depth is measured at eleven different locations along the wheel pass and the average rut depth is recorded. The test is conducted for 20,000 passes or until 12.5 mm rut depth is achieved as discussed in Chapter 2.

The statistical analysis of the test results for cracking and rutting evaluation was conducted using Tukey's Honestly Significant Difference (Tukey's HSD). The Tukey's HSD is a one-way analysis of variance (ANOVA) and is performed at 95 percent confidence interval (i.e.,  $\alpha = 0.05$ ). It can identify test means with significant difference. In this study, the researchers used Minitab software (Minitab 2019) to conduct the statistical analysis. In this chapter, the researchers discussed the results for mixtures prepared with various RAP sources separately. RAP materials from four different sources were utilized in this study as discussed in Chapter 3.

# **Evaluation of Asphalt Mixtures Prepared with RAP No. 1**

The source of RAP No. 1 was a loose mixture (PMLC) obtained from an ITD paving project. The loose mixture was aged in the laboratory by placing the mixture in an oven at 135 °C for 3 days to simulate aged RAP materials as discussed in Chapter 3. Therefore RAP No. 1 was an artificial RAP. The researchers conducted this control experiment using artificial RAP to eliminate the effect of RAP variability (e.g., gradation, binder content, etc.) on the test results when other parameters were evaluated such as

rejuvenator type and content. Different amounts of RAP (i.e., 0, 25, 50, and 75 percent), four different rejuvenators (R1, R2, R3, and R4) and two different doses from each rejuvenator were evaluated. The examined doses were based on the manufacturers' recommendations and literature as discussed in Chapter 3 in Table 10. The recommended dose for R1 is between 3.5 to 7.0 percent by weight of total binder. The recommended dose of R2 depends on RAP content (i.e., for less than 30 percent RAP, 5 to 7 percent by weight of reclaimed binder and for more or equal to 30 percent RAP, 9 to 12 percent by weight of reclaimed binder). For the third rejuvenator (R3), the recommended dose is between 12.5 to 15 percent by weight of reclaimed binder. For rejuvenator No. 4 (R4), the recommended dose is between 1 to 3 percent of weight of RAP, while it is between 12 to 16 percent by weight of reclaimed binder for R5. For R6 and R7, the examined doses were 6.6 and 8.3 percent by weight of reclaimed binder for R6 and 11.3 percent by weight of reclaimed binder for R7.

#### **Effect of RAP Content**

Figure 28 shows the effect of RAP content (i.e., 0, 25, 50, and 75 percent) on cracking performance using IDEAL-CT<sub>index</sub>. Figure 28 also includes performance thresholds proposed in ITD RP 261 (Kassem et al. 2019) where the red dotted line represents the higher threshold (IDEAL-CT<sub>Index</sub> of 73.7), and the solid blue line shows the minimum threshold (IDEAL-CT<sub>index</sub> of 26.4). Mixtures with IDEAL-CT<sub>index</sub> greater than 73.7 are expected to exhibit good cracking resistance, while mixtures with IDEAL-CT<sub>Index</sub> less than 26.4 are expected to have poor resistance to cracking. Mixtures with IDEAL-CT<sub>Index</sub> between 26.4 and 73.7 are expected to exhibit moderate or fair cracking performance. Tukey's HSD was also conducted to compare the IDEAL-CTIndex values for the examined mixtures. The statistical analysis results (Tukey's HSD groups) are included in the form of capital letters on each bar. If the mixtures share the same capital letters (e.g., A, B, C, D, etc.) then, there was no significant difference between the means.

The results of Figure 28 demonstrate that the control mix (i.e., 0 percent RAP) had higher IDEAL- $CT_{index}$ which indicates good cracking performance. The addition of 25 percent RAP decreased the IDEAL-CT<sub>index</sub> but still the mixture is expected to exhibit good cracking performance, and there was no statistically significant difference in the IDEAL-CT<sub>Index</sub> results between mixtures without RAP (0 percent RAP) and with 25 percent RAP. Mixtures with 50 percent and 75 percent RAP showed lower IDEAL-CT<sub>index</sub> and are expected to exhibit poor cracking performance (average IDEAL-CT<sub>index</sub> was less than 26.4). In addition, mixtures with 50 percent and 75 percent RAP share the same letter "B" which indicate that there was no statistically significant difference in the IDEAL-CT<sub>Index</sub> results between these two mixtures. However, there was statistically significant difference in IDEAL-CT<sub>Index</sub> results between these mixtures (i.e., mixtures with 50 percent and 75 percent RAP) and mixtures without RAP and 25 percent RAP.

Figures 29 and 30 show the IDT<sub>Strength</sub> and Weibull<sub>CRI</sub> results for mixtures prepared at different RAP contents using RAP No. 1. The results showed that the IDT<sub>strength</sub> increased with the increase of RAP content which

demonstrates that the mixtures are stiffer with the addition of RAP. Meanwhile, mixtures with 70 percent RAP had slightly lower stiffness compared to 50 percent; however, such difference was not statistically significant. Furthermore, the results Weibull<sub>CRI</sub> were consistent with that of IDEAL-CT<sub>Index</sub> and the cracking performance decreased from good (Weibull<sub>CRI</sub> > 4.7) to poor (Weibull<sub>CRI</sub> < 3.6) with increasing RAP content. Therefore, the researchers focused on the results of IDEAL-CT<sub>Index</sub> in this section and provided the IDT<sub>Strength</sub> and Weibull<sub>CRI</sub> results in Appendix E.

The results of this section clearly demonstrated the detrimental effect of increasing RAP content on the cracking performance of asphalt mixtures prepared with RAP No. 1. Increasing the amount of RAP in asphalt mixtures resulted in stiffer mixtures with reduced resistance to cracking as expected. Next, the researchers evaluated the use of rejuvenators to improve the cracking resistance of these mixtures.



Figure 28. Effect of RAP Content on Cracking Performance for RAP No. 1



Figure 29. Effect of RAP Content of RAP No. 1 on IDT<sub>Strength</sub>



Figure 30. Effect of RAP Content of RAP No. 1 on Weibull  $_{\mbox{\tiny CRI}}$ 

#### **Effect of Rejuvenator Type**

The researchers evaluated the use of rejuvenators to improve the cracking resistance of mixtures prepared with different RAP contents and the results were compared to the control mix (i.e., without RAP). Figures 31, 32, and 33 show the effect of the first dose (lower end) of different types of rejuvenators for mixtures with RAP content of 25, 50, and 75 percent, respectively. For RAP No. 1, the researchers evaluated four different rejuvenators 1) tall oil (R1) at 3.5 percent by weight of total binder for 25 and 50 percent RAP and 5 percent for 75 percent RAP; 2) aromatic extract (R2) at 6 percent by weight of the reclaimed binder for the 25 percent RAP and 10 percent by weight of reclaimed binder for 50 percent and 75 percent RAP, 3) bio-based forestry oil (R3) at 12.5 percent by weight of reclaimed binder, and 4) engineered product (R4) at 1 percent of weight of RAP. These doses were discussed in Chapter 3.

As shown earlier from Figure 28, mixtures with 25 percent RAP showed good cracking performance based on proposed IDEAL-CT<sub>index</sub> thresholds. The results of Figure 31 demonstrated that adding rejuvenators didn't significantly impact the cracking performance except mixtures with R2 (25 percent plus R2) compared to mixtures with 25 percent RAP. The use of R2 resulted in reduced cracking resistance. For mixtures with 50 percent RAP (Figure 32), the addition of rejuvenators R2 and R4 showed improvements to cracking resistance; however, R1 and R3 didn't have significant improvement on cracking resistance compared to mixtures with 50 percent RAP. Unlike mixtures with 25 percent RAP, R2 improved the cracking performance from poor to good and this is attributed to increased rejuvenator dose as specified for mixtures with 30 percent RAP or higher. The dose was increased as specified for R2 (i.e., for less than 30 percent RAP, 5 to 7 percent by weight of reclaimed binder and for more or equal to 30 percent RAP, 9 to 12 percent by weight of reclaimed binder. M4 also exhibited improved the cracking resistance from poor to good cracking performance. Rejuvenators R1 and R3 improved the cracking resistance slightly but there was no statistically significant difference in the results compared to that of mixtures with 50 percent RAP.



Figure 31. Effect of Rejuvenator Type on Cracking Performance at 25% RAP No. 1



Figure 32. Effect of Rejuvenator Type on Cracking Performance at 50% RAP No. 1

Figure 33 shows the effect of rejuvenator type on the cracking performance for mixture prepared with 75 percent RAP. The results demonstrated that the use of R1, R2 and R4 improved the cracking resistance from poor to fair; however, the results of IDEAL-CT<sub>Index</sub> for R2 and R4 were not statistically significant different from the control mix (i.e., 75 percent RAP). Also, rejuvenator R3 improved the performance slightly but such improvement was not statistically significant.



Figure 33. Effect of Rejuvenator Type on Cracking Performance at 75% RAP No. 1

## **Effect of Rejuvenator Dose**

Figure 34 shows the effect rejuvenator dose on the IDEAL-CT<sub>Index</sub> results for mixtures prepared with 75 percent RAP. The researchers increased the dose for three rejuvenators (i.e., R2, R3, and R4). Two doses were evaluated for R2 (i.e., 10 and 12 percent by weight of reclaimed binder), two doses for R3 (i.e., 12.5 and 15 percent by weight of reclaimed binder), and two doses for R4 (i.e., 1 and 2 percent of weight of RAP). Increasing the rejuvenator dose for R2 improved the cracking performance, while it showed slight improvement for R3. On the other hand, the second dose of R4 adversely impacted the cracking resistance; however, such detrimental effect was not significant. These results clearly demonstrate that increasing the rejuvenator dose doesn't necessarily improve the cracking resistance. In some cases, it could adversely impact the cracking performance.



Figure 34. Effect of Rejuvenator Dose on Cracking Performance for RAP No. 1

# **Evaluation of RAP No. 2 Mixtures**

RAP No.2 was obtained from an asphalt plant in Lewiston, Idaho. The researchers prepared and tested asphalt mixtures in the laboratory prepared using RAP No. 2 to evaluate the cracking performance at various RAP contents (i.e., 0 percent [control], 25 percent, 50 percent, and 70 percent), three different binder grades (i.e., PG 58-34, PG 64-28, and PG 70-28), two different binder contents (i.e., optimum binder content [OBC] and OBC+0.5 percent), five different rejuvenators (i.e., R1, R2, R3, R4, and R5), and two different doses of each rejuvenator. For R1, the researchers evaluated two doses of 5 percent and 7 percent by weight of total binder. For R2, two doses of 6 percent and 7 percent by weight of reclaimed binder for mixtures with 25 percent RAP, and 10 percent and 12 percent by weight of reclaimed binder for mixtures with 50 and 70 percent RAP. Also, two doses of 12.5 percent and 15 percent by weight of reclaimed for R3. For R4, the researchers evaluated two doses of 1 percent and 2 percent by weight of RAP incorporated in the mixture. Finally, two doses of 12 percent and 16 percent by weight of reclaimed binder were examined for R5. Furthermore, the researchers conducted the creep compliance test to evaluate the low-temperature cracking as well as rutting testing for selected mixtures that exhibited good cracking performance.

## **Effect of RAP Content**

Figure 35 shows the results of IDEAL- $CT_{Index}$  for test mixtures prepared with different RAP contents of RAP No. 2. The results demonstrated that IDEAL- $CT_{Index}$  decreased with the increase of RAP content. This trend is consistent with the results of RAP No. 1 (artificial RAP). The mixture without RAP (0 percent) had higher

IDEAL- $CT_{Index}$  and fair cracking resistance (26.4 < IDEAL- $CT_{Index}$  > 73.7) compared to mixtures prepared with RAP No. 2. Mixture with 25, 50, and 70 percent RAP which had poor cracking resistance (26.4 > IDEAL- $CT_{Index}$ ). In addition, there was a statistically significant difference in IDEAL- $CT_{Index}$  results of control mixture compared to mixtures with 50 and 70 percent RAP. Also, there was no statistically significant difference in IDEAL- $CT_{Index}$  for mixtures prepared with various RAP contents (i.e., 25, 50, and 70 percent RAP).

Figures 36 and 37 show the IDT<sub>Strength</sub> and Weibull<sub>CRI</sub> results for mixtures prepared at different RAP contents using RAP No. 2, respectively. The results showed that the IDT<sub>Strength</sub> consistently increased with the increase of RAP content which demonstrated that the mixtures became stiffer with the addition of RAP similar to the results of RAP No. 1. Furthermore, the results of Weibull<sub>CRI</sub> were consistent with that of IDEAL-CT<sub>Index</sub> and the cracking performance decreased from fair (3.6 < Weibull<sub>CRI</sub> > 4.7) to poor (Weibull<sub>CRI</sub> < 3.6) with increasing RAP content. In this section, the researchers focused on the results of IDEAL-CT<sub>Index</sub> and provided the IDT<sub>Strength</sub> and Weibull<sub>CRI</sub> results in Appendix E.



Figure 35. Effect of RAP Content on Cracking Performance for RAP No. 2



Figure 36. Effect of RAP Content of RAP No. 2 on IDT<sub>Strength</sub>



Figure 37. Effect of RAP Content of RAP No. 2 on Weibull $_{\mbox{\tiny CRI}}$ 

## Effect of Rejuvenator Type

The researchers examined the effect of rejuvenator type at different doses on IDEAL-CT<sub>index</sub>. Figures 38, 40, and 41 show the effect of rejuvenators (i.e., R1, R2, R3, R4, and R5) at the first dose of each rejuvenator at 25, 50, and 70 percent RAP, respectively. The first dose was 5 percent by weight of total binder for R1, 10 percent by weight of reclaimed binder for R2, 12.5 percent by weight of reclaimed Binder weight for R3, 1 percent by weight of RAP for R4, and 12 percent by weight of reclaimed binder for R5. In addition, the researchers evaluated the mixtures at two different binder contents (i.e., OBC and OBC+0.5 percent). For mixtures with 25 percent of RAP No. 2 (Figure 38), the use of different rejuvenators results in improved cracking performance of higher IDEAL-CT<sub>index</sub> compared to mixtures with 25 percent RAP. Also, increasing the binder content by 0.5 percent above the optimum binder content increased the IDEAL-CT<sub>index</sub>. Meanwhile, the statistical analysis demonstrated that such improvement was not statistically significant. Also, increasing the binder content by 0.5 percent above the optimum binder content for mixtures without RAP (0 percent RAP) resulted in good cracking resistance (IDEAL-CT<sub>index</sub> > 73.7) compared to mixtures without RAP at optimum binder content which had fair cracking resistance (26.4<IDEAL-CT<sub>index</sub> > 73.7) and the difference in IDEAL-CT<sub>index</sub> results was statistically significant.

Similarly, all rejuvenators improved the cracking resistance for mixtures prepared with 50 percent RAP; however, such improvement was not statistically significant as shown in Figure 39. Meanwhile, R3 and R4 provided fair cracking performance comparable to the mixture without RAP (i.e., 0 percent RAP). Figure 40 shows the cracking performance at higher RAP content of 70 percent. All mixtures with rejuvenators had better cracking resistance (i.e., higher IDEAL-CT<sub>Index</sub>) compared to the control mixture (i.e., mixture with 70 percent RAP without rejuvenators). Mixtures with R4 exhibited statistically significant improvements compared to the control mixture. Also, increasing the binder content by 0.5 percent at 70 percent RAP increased IDEAL-CT<sub>Index</sub> and improved the cracking performance from poor to fair. Next, the researchers evaluated the effect of different rejuvenator doses at various RAP contents as discussed in the following section.



Figure 38. Effect of Rejuvenator Type on Cracking Performance at 25% RAP No. 2



Figure 39. Effect of Rejuvenator Type on Cracking Performance at 50% RAP No. 2



Figure 40. Effect of Rejuvenator Type on Cracking Performance at 70% RAP No. 2

## **Effect of Rejuvenator Dose**

The researchers evaluated the effect of various rejuvenator doses at different RAP contents. The researchers increased the dose for five rejuvenators (i.e., R1, R2, R3, R4, and R5). Two doses were evaluated for each rejuvenator as follows: R1 (5 percent and 7 percent by weight of total binder), R2 (6 percent and 7 percent by weight of reclaimed binder for mixtures with 25 percent RAP, and 10 percent and 12 percent by weight of reclaimed binder for mixtures with 50 and 70 percent RAP ), R3 (12.5 and 15 percent by weight of reclaimed binder), R4 (i.e., 1 and 2 percent of weight of RAP), and R5 (12 and 16 percent by weight of reclaimed binder). Figures 41, 42, and 43 show the IDEAL-CT<sub>index</sub> results for mixtures prepared with 25, 50, and 70 percent RAP, respectively. The following findings were found.

- Mixtures prepared with 25 percent RAP showed improved performance at higher rejuvenator dose (i.e., R1+, R2+, R3+, and R4+) compared to the control mixture with 25 percent RAP at optimum binder content (OBC). Also, the difference in IDEAL-CT<sub>Index</sub> results was statistically significant for R1 at higher dose (R1+) and R3 at higher dose (R3+).
- Mixtures prepared with 50 percent RAP showed improved performance at higher rejuvenator dose compared to the first dose and compared to the control mixture with 50 percent without rejuvenators. The use of R1+, R3+, and R4+ improved the cracking resistance from poor cracking performance (26.4 > IDEAL-CT<sub>Index</sub>) for the control mixture (i.e., 50 percent RAP) to moderate cracking performance (IDEAL-CT<sub>Index</sub> between 26.4 and 73.7). The IDEAL-CT<sub>Index</sub> of R4 at a higher dose was statistically significantly different compared to the control mixture and provided the best cracking performance among all examined rejuvenators.

Mixtures prepared with 70 percent RAP showed improved performance at higher rejuvenator dose (i.e., R1+, R3+, R4+, and R5+) compared to the first dose and compared to the control mixture prepared with 70 percent RAP, for all rejuvenators except for R2. Also, the results demonstrated that R4 at higher dose exhibited the best performance among all examined rejuvenators at different doses, even better than the mixture without RAP. However, this mixture was relatively wet (i.e., soft) and it failed in rutting as discussed later in this chapter. These results demonstrated the importance of the balanced mix design approach to satisfy both cracking and rutting criteria. Also, R1 and R5 improved the cracking resistance significantly when compared to the control mix (70 percent RAP) and provided comparable cracking performance to the mixture without RAP.



#### Figure 41. Effect of Rejuvenator Dose on Cracking Performance at 25% on RAP No. 2



Figure 42. Effect of Rejuvenator Dose on Cracking Performance at 50% on RAP No. 2



#### Figure 43. Effect of Rejuvenator Dose on Cracking Performance at 70% on RAP No. 2

# Effect of Rejuvenator on Cracking Performance of Mixtures with Different Binder Grades

Based on the results of the previous section, R1 and R5 were selected for further evaluation at higher dose with different binder grades (i.e., PG 58-34, PG 64-28, and PG 70-28). The objective of this evaluation was to assess the positive impact of these two rejuvenators at higher doses for mixtures prepared with different binder grades. It should be noted that R4 also provided good cracking performance but produced relatively softer mixtures susceptible to rutting; therefore, it was not included in this additional evaluation.

Figure 44 shows the IDEAL- $CT_{Index}$  results for mixtures prepared without RAP (0 percent RAP) and 70 percent RAP, with and without rejuvenators (I.e., R1 and R5 at higher doses) with different binder grades. In addition, Figures 45 and 46 show the IDT<sub>Strength</sub> and Weibull<sub>CRI</sub>, respectively. The main findings of the results can be summarized below.

- Mixtures prepared with 70 percent RAP had lower IDEAL-CT<sub>Index</sub> values compared to the ones without RAP (0 percent RAP) irrespective of the binder grade. Mixtures with RAP had poor cracking resistance (26.4>IDEAL-CT<sub>Index</sub>) compared to the control mixture with moderate cracking resistance (IDEAL-CT<sub>Index</sub> between 26.4 and 73.7).
- The binder grade did not affect the IDEAL-CT<sub>Index</sub> for mixtures without RAP (0 percent RAP) and those prepared with 70 percent RAP. However, the IDT<sub>Strength</sub> for mixtures with PG 64-28 and PG 70-28 was higher compared to the ones for PG 58-34 for the mixtures without RAP. The results of IDT<sub>Strength</sub> were comparable for mixtures of different binder grades at higher RAP content (70 percent RAP) which demonstrates stiffer mixtures.
- The use of R5 resulted in significantly improved the cracking performance (higher IDEAL-CT<sub>index</sub>) compared to mixtures with 70 percent RAP without rejuvenators irrespective of the binder grade. In addition, the use of R5 resulted in comparable cracking performance to the mixtures without RAP at the corresponding binder grade. The IDT<sub>strength</sub> results further demonstrated that R5 was effective in reducing the mixture stiffness irrespective of the binder grade too.
- Mixtures prepared with 70 percent RAP and rejuvenator R1 showed improved cracking performance compared to mixtures at the same RAP content without rejuvenator. Also, there was significant improvement in IDEAL-CT<sub>index</sub> results for mixtures with PG 58-34 and PG 64-28. Therefore, R1 provided better results with softer binders which could be attributed to the compatibility between asphalt binders and rejuvenators. Also, the use of R1 resulted in comparable cracking performance to the mixtures without RAP for PG 58-34 and PG 64-28.
- The results of Weibull<sub>CRI</sub> shown in Figure 46 had the same trend as the IDEAL-CT<sub>index</sub> results, and there was good agreement between both performance indicators.
- The results of this section clearly demonstrated the effectiveness of rejuvenators in improving the cracking performance at higher RAP content which offers significant environmental and economic benefits. The associated cost savings are discussed in Chapter 6.



Figure 44. Effect of Rejuvenator on Cracking Performance for Different Binder Grades



Figure 45. Effect of Rejuvenator on IDT<sub>Strength</sub> for Different Binder Grades





#### **Evaluation of Rutting Performance**

To ensure that mixtures prepared with RAP and rejuvenators attain good resistance to rutting, the researchers evaluated the rutting performance using the HWTT. Test mixtures prepared with RAP and rejuvenators that exhibited improved cracking performance were further evaluated for rutting performance. Mixtures without RAP (i.e., 0 percent RAP) and with 70 percent RAP, prepared without rejuvenators and with rejuvenators (i.e., R1, R4, and R5) at different rejuvenator doses (i.e., 7 percent by weight of total binder for R1, 1 percent and 2 percent by wight of RAP for R4, and 12 percent and 16 percent by weight of reclaimed binder for R5) were tested.

Figure 47 shows the HWTT rut depth results. Mixtures prepared with R4 at two different doses (i.e., R4 and R4+) failed HWTT. Mixtures with R4 at 1 percent by weight of RAP accumulated rut depth of 12.5 mm after 7060 passes, while mixtures with R4 at 2 percent by weight of RAP reached 12.5 mm rut depth after only 2470 passes. Rejuvenator R4 over softened the asphalt mixtures, and this was observed during the laboratory preparation of these mixtures. It should be noted that R4, which is an engineered commercial product, had the lowest viscosity among all test rejuvenators. All remaining test mixtures, including the ones prepared with R1 and R5, experienced very low rutting (rut depth less than 3 mm after 20,000 passes) and passed the test criteria of 12.5 mm after 20,000 passes. Mixtures prepared with R1 and R5 had comparable rutting performance to that of mixtures with RAP (70 percent RAP) and mixtures without RAP and rejuvenators. Figures 48 and 49 further illustrate the rutting performance of asphalt mixtures prepared with R4 at 1 percent dose and 2 percent dose, respectively.

The results of this section further demonstrated the importance of the balanced mix design approach to produce asphalt mixtures with adequate resistance to cracking and rutting. Some mixtures prepared with certain rejuvenators (R4 as an example) had good cracking resistance but failed the rutting resistance criteria; therefore, it is important to ensure that mixtures prepared with RAP and rejuvenators achieve balanced performance.



Figure 47. Evaluation of HWTT Rutting Performance of Asphalt Mixtures with RAP No. 2


Figure 48. HWTT Test Specimens with 70% RAP and Rejuvenator 4 at 1 Percent Dose



Figure 49. HWTT Test Specimens with 70% RAP and Rejuvenator 4 at 2 Percent Dose

## **Evaluation of RAP No. 3 Mixtures**

RAP No.3 was obtained from a second asphalt plant in Lewiston, Idaho. The researchers evaluated the cracking performance of asphalt mixtures prepared without RAP (0 percent RAP), 70 percent RAP, two different rejuvenators (i.e., R1 and R5) at optimum binder content (5.8 percent of PG 58-28). For rejuvenator R1, the dose was 7 percent by weight of total binder and 16 percent by wight of reclaimed binder for R5. These doses were found to provide good cracking performance based on the testing results

of mixtures prepared with RAP No. 2. In addition, the researchers evaluated the rutting performance and moisture susceptibility of test mixture using the HWTT. Furthermore, the researchers investigated the low temperature cracking performance in accordance with AASHTO T 322 *"Standard Method of Test for Determining the Creep Compliance and Strength of Hot Mix Asphalt (HMA) Using the Indirect Tensile Test Device"*.

### Effect of RAP and Rejuvenator Type

Similar to mixtures prepared with RAP No. 1 and RAP No. 2, incorporating 70 percent RAP of RAP No. 3, resulted in stiffer mixtures with lower IDEAL- $CT_{Index}$  compared to the control mixture without RAP. Mixtures prepared with 70 percent RAP showed poor cracking resistance (26.4 > IDEAL- $CT_{Index}$ ) while mixture without RAP had moderate performance (IDEAL- $CT_{Index}$  between 26.4 and 73.7) as shown in Figure 50. The use of R1 and R5 increased IDEAL- $CT_{Index}$  which indicates improved cracking performance. Both rejuvenators produced mixtures with good cracking performance (IDEAL- $CT_{Index}$  between 26.4 and 73.7) compared to the control mixture which exhibited moderate cracking resistance (IDEAL- $CT_{Index}$  between 26.4 and 73.7). The results of Figure 50 further demonstrated that increasing the binder content by 0.5 percent at 70 percent RAP (70 percent + 0.5BC) increased IDEAL- $CT_{Index}$  compared to mixtures prepared at optimum binder content and 70 percent RAP, and such improvement was statistically significant. Meanwhile, mixtures with no RAP at a higher binder content (0.5 percent above optimum binder content) provided better cracking performance compared to mixtures with no RAP at optimum binder content.

Figures 51 and 52 show the results of  $IDT_{Strength}$  and Weibull<sub>CRI</sub>, respectively. Mixtures prepared with high RAP content (i.e., 70 percent RAP) had higher  $IDT_{Strength}$  compared to mixtures without RAP (0 percent RAP); however, such increased stiffness was associated with reduced cracking resistance (lower Weibull<sub>CRI</sub> and IDEAL-CT<sub>Index</sub>). The results of loose mixtures presented in Chapter 5 showed that this an opposite correlation between  $IDT_{Strength}$  and both  $IDEAL-CT_{Index}$  and Weibull<sub>CRI</sub>. Furthermore, the results of Weibull<sub>CRI</sub> were consistent with that of  $IDEAL-CT_{Index}$ . Increasing the binder content and the use of rejuvenators (i.e., R1 and R5) were found to be very effective in improving the cracking resistance based on the Weibull<sub>CRI</sub> results.



Figure 50. Effect of RAP Content and Rejuvenator Type on IDEAL-CT<sub>Index</sub> for RAP No. 3



Figure 51. Effect of RAP Content and Rejuvenator Type on IDT<sub>Strength</sub> for RAP No. 3



Figure 52. Effect of RAP Content and Rejuvenator Type on Weibull<sub>CRI</sub> for RAP No. 3

#### **Evaluation of Low-Temperature Cracking**

Researchers further evaluated the low-temperature or thermal cracking of mixtures prepared with RAP No. 3 and rejuvenators. They evaluated mixtures without RAP (0 percent RAP), 70 percent RAP, and 70 percent RAP with rejuvenators R1 and R5. Previous study by Safi et al. (2018) showed that stiffer mixtures tend to have lower deformation and thus lower compliance. Figures 53 through 55 show the creep compliance for the test mixtures at different temperatures of -20 °C, -10 °C, 0 °C, respectively. Figure 56 shows the creep compliance at all test temperatures. Also, Figure 57 shows the IDT<sub>Strength</sub> for the same mixtures measured at -10 °C after the completion of creep compliance testing. The main findings of these results can be summarized below.

- Mixtures with 70 percent RAP had the lowest creep compliance compared to mixtures without RAP (i.e., 0 percent RAP), and mixtures prepared with 70 percent RAP with rejuvenators R1 and R5 (i.e., 70 percent+R1 and 70 percent+R5). Lower compliance values are associated with stiffer mixtures and less resistance to low-temperature or thermal cracking.
- Mixtures prepared with 70 percent RAP and R1 and R5 had the highest creep compliance and consequently would exhibit better resistance to thermal cracking compered to mixtures with 70 percent RAP without rejuvenators as well as mixtures without RAP and rejuvenators. Also, mixtures without RAP and rejuvenators had higher creep compliance than those of mixtures with 70 percent RAP without rejuvenators and lower creep compliance than mixtures with rejuvenators. These results demonstrated that the use of rejuvenators (i.e., R1 and R5) were

found effective in improving the thermal cracking resistance of asphalt mixtures with high RAP content (i.e., 70 percent).

- The results of IDT<sub>strength</sub> measured at -10 °C demonstrated that mixtures without RAP and rejuvenators had comparable IDT<sub>strength</sub> values with no statistically significant difference in the results.
- The use of rejuvenators reduced the IDT<sub>Strength</sub> compared to mixtures without rejuvenators (with and without RAP). Mixtures with R5 had the lowest IDT<sub>Strength</sub> compared to all other mixtures. In addition, the difference in IDT<sub>Strength</sub> results between mixtures with R5 and mixtures without rejuvenators was statistically significant.
- There is a good agreement between the results of IDT<sub>Strength</sub> (measured at -10 °C) and that of the creep compliance test. The results further demonstrated the favorable impact of rejuvenators on improving the performance of asphalt mixtures with high RAP content at low temperature.



Figure 53. Creep Compliance Results at -20 °C







Figure 55. Creep Compliance Results at 0 °C



Figure 56. Creep Compliance Results at Different Temperatures



Figure 57. Measured IDT<sub>Strength</sub> of Test Mixtures at -10 °C

### **Evaluation of Rutting Performance**

Mixtures prepared with 70 percent of RAP No. 3 and rejuvenators R1 and R5 all passed HWTT rutting criteria. The rut depth values after 20,000 passes were way below the rutting threshold of 12.5 mm as shown in Figure 58. Furthermore, there was no sign of moisture damage. In addition, the results showed that there was no statistically significant difference in the rut depth results between mixtures with rejuvenators and high RAP content of 70 percent and the control mixture with no RAP. These results demonstrated that the use of rejuvenators R1 and R5 at high RAP content of 70 percent produced mixtures with good resistance to intermediate-temperature as well as low-temperature cracking. In addition, these mixtures exhibited good resistance to rutting and moisture damage.



Figure 58. Evaluation of Rutting Performance of Mixtures with RAP No. 3

## **Evaluation of RAP No. 4 Mixtures**

The researchers evaluated the performance of asphalt mixtures prepared with a fourth source of RAP (i.e., RAP No. 4). This RAP was obtained from an asphalt plant in Boise, Idaho. The asphalt mixtures were prepared at different RAP contents (i.e., 0, 25, 50 and 70 percent) as well as rejuvenator type and doses. Two rejuvenators (i.e., R6 and R7) were used at 6.6 percent and 8.3 percent by weight of RAP binder for R6 and 11.3 percent by weight of RAP binder for R7. Rejuvenator R6 is a bio-based product while R7 is petroleum-based product as discussed in Chapter 3.

The researchers conducted the IDT and HWTT to assess the intermediate temperature cracking and rutting performance, respectively and this section discusses the results of these tests.

### **Effect of RAP Content**

Figure 59 shows the effect of RAP content on IDEAL-CT<sub>Index</sub>. Mixtures without RAP had higher IDEAL-CT<sub>Index</sub> and it decreased with the increase of RAP content. There is a statistically significant difference in the IDEAL-CT<sub>Index</sub> results for mixtures prepared without and with RAP. Based on the results of Figure 59, mixtures without RAP had good cracking performance (IDEAL-CT<sub>Index</sub> > 73.7), mixtures with 25 and 50 percent RAP had moderate cracking performance (IDEAL-CT<sub>Index</sub> between 26.4 and 73.7), while mixtures with 70 percent RAP had poor cracking resistance (26.4>IDEAL-CT<sub>Index</sub>). These results are consistent with the other sources of RAP (i.e., RAP No. 1, 2, and 3) where increasing RAP content resulted in decreased cracking performance.



Figure 59. Effect of RAP Content on Cracking Performance of Mixtures Prepared with RAP No. 4

## Effect of Rejuvenator Type

Figure 60 shows the effect of rejuvenator type and first dose on the IDEAL-CT<sub>Index</sub> results. Mixtures prepared with different RAP contents (i.e., 0, 25, 50, and 70 percent) and two types of rejuvenators (i.e., R6 and R7). The following findings can be summarized from these results.

- The use of rejuvenators R6 and R7 didn't affect the IDEAL-CT<sub>Index</sub> significantly at 25 percent RAP. It should be noted that mixtures with 25 percent RAP had good cracking performance. These results are consistent with the results of other RAP sources where the use of rejuvenators at low RAP content didn't improve the cracking performance.
- The use of softer binder PG 64-34 at 25 percent RAP did not affect IDEAL-CT<sub>Index</sub> compared to mixtures prepared with 25 percent RAP and PG 70-28 and there was no statistically significant difference in the results.
- Similar to mixtures prepared with 25 percent RAP, the IDEAL-CT<sub>Index</sub> results of mixtures prepared with 50 percent RAP did not exhibit improved cracking performance when rejuvenators R6 and R7 were used at the lower dose (i.e., 6.6 percent by weight of reclaimed binder for R6 and 11.3 percent by weight of reclaimed binder for R7). It should be noted that the results of other RAP sources demonstrated that some rejuvenators were effective at 50 percent RAP which might be attributed to the compatibility between asphalt binders and rejuvenators.
- The IDEAL-CT<sub>Index</sub> for mixtures prepared with 70 percent RAP further demonstrated that the use of both rejuvenators, R6 and R7 improved the cracking resistance from poor cracking performance to moderate performance; however, there was no statistically significant difference mixtures with 70 percent RAP with and without rejuvenator.



Figure 60. Effect of Rejuvenator Type on 25% RAP No. 4

### **Effect of Rejuvenator Dose**

The researchers examined a higher dose of 8.3 percent by weight of reclaimed binder for R6 only. Figure 61 shows the IDEAL-CT<sub>Index</sub> results of the tested mixtures at different doses of R6 (i.e., 6.6 and 8.3 percent by weight of reclaimed binder) and different RAP contents (i.e., 0, 25, 50, and 70 percent RAP). The following observations can be drawn from the results shown in Figure 61.

- The increase of rejuvenator dose from at 6.6 to 8.3 percent by weight of RAP binder for R6 at 25 and 50 percent RAP increased the cracking performance; however, the IDEAL-CT<sub>Index</sub> was comparable to mixtures prepared with 70 percent RAP without rejuvenators. In other words, there was no favorable effect for R6 on the cracking performance at both 25 and 50 percent RAP at different doses.
- For mixtures prepared with 70 percent RAP and R6 at 6.6 percent by weight of reclaimed binder showed improved cracking resistance compared to mixtures with the same RAP content (i.e., 70 percent). The IDEAL-CT<sub>Index</sub> increased from 25, which indicates poor cracking resistance (26.4 >IDEAL-CT<sub>Index</sub>) to IDEAL-CT<sub>Index</sub> of 39, which indicates fair or moderate cracking resistance (IDEAL-CT<sub>Index</sub> between 26.4 and 73.7). The increase of dose from 6.6 to 8.3 percent of R6 had an inverse effect on the cracking resistance (decreased IDEAL-CT<sub>Index</sub>); however, such reduction was not statistically significant. These results demonstrated the need to evaluate different doses of the same rejuvenator to select the one that provides optimum performance at the mixture level. Also, an increase in rejuvenator's dose does not necessarily translate into improved mixture performance.



Figure 61. Effect of Rejuvenator Dose on 25% RAP No. 4

### **Evaluation of Rutting Performance**

The researchers evaluated the rutting performance of all test mixtures prepared using RAP No. 4. Figure 62 shows the effect of RAP content on HWTT rut depth. Test mixtures prepared with RAP had less rutting compared to mixtures without RAP; however, all the mixtures had rut depths way below the threshold of 12.5 mm after 20,000 passes. Furthermore, mixtures prepared with rejuvenators also satisfied the rutting performance as shown in Figure 63.



Figure 62. Evaluation of Rutting Performance on RAP No. 4



Figure 63. Effect of Rejuvenator Type and Dose on Rutting Performance of RAP No. 4

## 5. Evaluation of Field Mixes

Chapter 5 evaluates the cracking and rutting performance of loose mixtures collected from new paving projects. These loose mixtures were collected from different districts in Idaho and have different properties (mix design, binder grade, binder content, RAP, etc.) as discussed in Chapter 3. The researchers calculated different cracking and rutting performance parameters and compared to performance thresholds proposed and used in previous ITD studies (Kassem et al. 2019 and Kassem et al. 2021). Furthermore, the researchers examined the coefficient of variation of various cracking performance indicators of the field projects along with their correlations.

#### **Cracking and Rutting Performance Thresholds**

Kassem et al. (2019) proposed monotonic performance thresholds for various performance indicators to assess the cracking and rutting performance for asphalt mixtures produced in Idaho. The cracking performance indicators included IDEAL-CT<sub>Index</sub>, Weibull<sub>CRI</sub>, N<sub>flex</sub> factor, Cracking Resistance Index (CRI), and Flexibility Index from IDT test (FI[IDT]). They proposed a minimum and maximum threshold for each cracking performance indicator. Table 14 summarizes the recommended thresholds for each indicator that were proposed by Kassem et al. (2019). Mixtures with cracking performance below the minimum thresholds are expected to exhibit poor cracking resistance, while mixtures with cracking performance above the maximum thresholds are expected to exhibit good cracking resistance. Mixtures with performance between the minimum and maximum thresholds are expected to show fair or moderate cracking resistance.

	Poor	Fair/Moderate	Good
IDEAL-CT <sub>Index</sub>	<26.4	$26.4 \leq IDEAL-CT_{Index} \leq 73.7$	>73.7
Weibull <sub>cri</sub>	<3.60	3.60 ≤ WeibullCRI ≤ 4.70	>4.70
N <sub>flex</sub>	<0.40	0.40 ≤ Nflex≤ 0.70	>0.70
CRI (IDT)	<466	466 ≤ CRI (IDT) ≤ 614	>614
FI (IDT)	<11.4	11.4 ≤ FI (IDT) ≤ 22.6	>22.6

Table 14. Proposed Thresholds for Monotonic Cracking Performance Indices (after Kassem at al. 2019)

Kassem et al. (2019) reviewed the HWTT rutting depth thresholds used in different DOTs (i.e., TxDOT, WSDOT, CODOT, LADOT, and MTDOT). All DOTs performed the HWTT test at 50°C. ITD RP 261 proposed a maximum rut depth of 10 mm after 15,000 passes or 12.5 mm after 20,000 passes to ensure adequate

resistance to both rutting and moisture damage (Kassem et al. 2019). TxDOT specifies a max rut depth of 12.5 mm at different number of HWTT passes based on the binder grade (i.e.,  $\leq$  PG 64 required a minimum of 10,000 passes, PG 70 required a minimum of 15,000, and  $\geq$  PG 76 required a minimum of 20,000). WSDOT specifies a maximum rut depth of 10 mm at 15,000 passes. In addition, CODOT specifies a maximum rut depth of 4 mm at 10,000 passes while LADOT specifies a maximum rut of 10 mm at different number of passes based on the mixture type (i.e., Incidental Paving and ATB required at least 10,000 passes, while Wearing and Binder Course required at least 20,000 passes). MTDOT specifies a maximum rut depth of 13 mm and a minimum number of passes based on the mix design (i.e., the mix design required a minimum of 10,000 passes and the plant mix required a minimum of 15,000 passes). Table 9, in Chapter 2, summarizes the HWTT rutting performance thresholds used by different DOTs.

#### **Evaluation of Cracking Resistance of Field Mixes**

IDEAL-CT<sub>Index</sub> is one of the monotonic cracking performance indicators that is obtained from the IDT loaddisplacement curve. This index can be used to evaluate the cracking resistance for a given asphalt mixture. Mixtures with higher IDEAL-CT<sub>Index</sub> exhibit higher cracking resistance and vice versa. The IDEAL-CT<sub>Index</sub> is calculated using Equation 1. The researchers calculated IDEAL-CT<sub>Index</sub> for all PMLC specimens. Figure 64 shows the results of the IDEAL-CT<sub>Index</sub> for 23 different mixes that were collected from different districts. Most of the PMLC mixes (18 out of 23) in Figure 64 exceeded the maximum threshold of 73.7 which indicate that these mixtures are expected to exhibit good cracking resistance in the field based on the proposed threshold by Kassem e al. (2019). Four projects were within the moderate cracking performance range including Project No.7, No.12, No.14, and No. 23. Only one project (Project No. 11) had an IDEAL-CT<sub>Index</sub> less than the minimum threshold of 26.4.

It was observed that some mixtures with good cracking resistance used a higher binder content or lowered down the binder grade if they included high RAP content. For instance, Project No. 10 and Project No. 15 had a good cracking resistance with 5.7 percent and 5.2 percent binder content, respectively. Both projects had 30 percent RAP and lowered the binder PG by one grade to PG 58-34. In addition, Project No. 21 found to have a good resistance to cracking had 5.8 percent binder content, 17 percent RAP, and the specified binder grade of PG 64-28. Project No. 2 had the highest binder content of 6.2 percent, 0 percent RAP, and PG76-28 among other projects which provided a good cracking resistance. On other hand, Project No. 7 had a moderate cracking resistance and had PG 64-34, 30 percent RAP, and 5.6 percent binder content. Moreover, Project No. 12 used 32 percent RAP, 5.3 percent binder content, and PG 58-34 and showed moderate cracking resistance which could be attributed to RAP sources as some may have been aged more than others. Project No. 11 with poor cracking resistance had RAP content of 31 percent, binder content of 5.3 percent, and PG 64-34. The researchers observed that this mix (i.e., Project No. 11) was relatively drier and has more fines compared to other projects.

The Tukey's HSD analysis classified the mixes into five statistical groups (A, B, C, D, and E). The statistical analysis showed that there was a statistically significant difference between Project No. 11 compared to

Project No. 4, No. 8, No. 10, No. 15, No.16, No. 17, No. 19, and No. 21, while there was no significant difference between Project No. 11 and the remaining projects.





Weibull<sub>CRI</sub> is another cracking performance indicator that was examined in this study. Weibull<sub>CRI</sub> is calculated from the IDT test by fitting the entire load-displacement curve as discussed in Chapter 2. The higher the Weibull<sub>CRI</sub> value the better cracking resistance and vice versa. Figure 65 shows the Weibull<sub>CRI</sub> values for the examined 23 mixes. The figure includes the maximum (i.e., 4.7) and minimum (i.e., 3.6) thresholds for the Weibull<sub>CRI</sub> cracking performance indicator. The results demonstrated that 18 field projects were above the maximum threshold and are expected to exhibit good resistance to cracking. Four projects (i.e., Project No.7, No.12, No.14, and No. 23) are expected to show fair cracking resistance, while only one project (i.e., Project no. 11) is expected to exhibit poor cracking resistance (Weibull<sub>CRI</sub> < 3.6). These results are consistent with the results of IDEAL-CT<sub>index</sub>. However, the Tukey's HSD analysis classified the mixes into eight statistical groups (A, B, C, D, E, F, G, and H). This could be due to the lower coefficient of variation for Weibull<sub>CRI</sub> compared to IDEAL-CT<sub>Index</sub> as discussed later in this section. A higher number of statistical groups helps to distinguish between more mixes in terms of cracking resistance. The statistical analysis showed that there was a statistically significant difference between Project No. 11 and all the projects except those with fair cracking resistance (i.e., Project No.7, No.12, No.14, and No. 23). These results demonstrated that Weibull<sub>CRI</sub> was able to distinguish between more projects in terms of cracking resistance compared to IDEAL-CT<sub>Index</sub>. Based on the cracking performance results of IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub> and the corresponding thresholds proposed in RP 261, there is no concern with the expected cracking performance of most asphalt mixtures currently produced in Idaho.



Figure 65. WeibullCRI for PMLC Mixes

Figure 66 illustrates the results of IDT<sub>Strength</sub> for the field projects. The IDT<sub>Strength</sub> is calculated from the IDT load-displacement curve as discussed in Chapter 2. Mixtures with higher IDT<sub>Strength</sub> are often stiffer compared to those with lower IDT<sub>Strength</sub> and vice versa. Stiffer mixtures often exhibit lower resistance to cracking. Based on the IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub> cracking resistance for different groups (i.e., good, fair, and poor), the group of mixtures with good cracking resistance had an IDT<sub>Strength</sub> ranging from 872 kPa to 1004 kPa, while mixtures with fair cracking resistance had an IDT<sub>Strength</sub> ranging from 774 kPa to 1160 kPa. The mixture (Project No. 11) that is expected to exhibit poor cracking resistance had relatively higher IDT<sub>strength</sub> of 1156 kPa. The IDT<sub>strength</sub> had an opposite correlation with both IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub>, as discussed later in this chapter; however, such correlation is not strong. Similar to Weibull<sub>CRI</sub>, the Tukey's HSD analysis classified the mixes, based on IDT<sub>Strength</sub>, into eight statistical groups (A, B, C, D, E, F, G, and H). Also, there was a statistically significant difference between Project No. 11 and some projects with good cracking performance including Project No. 1, No. 2, No. 4, No. 6, No. 8, No. 9, No. 10, No. 13, No. 16, No. 19, No. 20, No. 21, and No. 22.



Figure 66.IDT<sub>Strength</sub> for PMLC Mixes

## **Evaluation of Correlation and Coefficient of Variation for Cracking Performance** Indicators

The researchers evaluated various cracking performance indicators including IDEAL- $CT_{Index}$ , Weibull<sub>CRI</sub>, N<sub>flex</sub> factor, Cracking Resistance Index (CRI), and Flexibility Index from IDT test (FI[IDT]), Fracture Energy (Gf), IDT<sub>Strength</sub>, and IDT<sub>Modulus</sub>. All these performance indicators can be calculated from the same IDT load-displacement curve and conducted using the same testing protocol. The formulas used to calculate these indicators are summarized in Table 8 of Chapter 2.

The researchers evaluated the correlation among the above-mentioned cracking performance indicators. The Pearson coefficient is a statistical tool used in this study to examine the correlation between various cracking performance indicators. The Pearson product moment correlation coefficient (r) is used to assess the linear relationship between two indicators (Salkind 2010). The value of "r" ranges between -1 and +1. The magnitude of "r" describes the strength of the correlation between two parameters, while the sign (negative or positive) indicates whether such relationship is direct (+) or inverse (-). Higher magnitude of "r" demonstrates stronger correlation. Table 15 presents the value of r between various cracking performance indicators. The results clearly demonstrated that IDEAL-CT<sub>index</sub>, Weibull<sub>CRI</sub>, N<sub>Flex</sub>, CRI, and FI had direct strong correlations (r > 0.90). Also, the results showed that both IDT<sub>Strength</sub> and IDT<sub>Modulus</sub> had an inverse correlation with most indicators (i.e., that IDEAL-CT<sub>index</sub>, Weibull<sub>CRI</sub>, N<sub>Flex</sub>, CRI, and FI), except between each other. Fracture Energy (Gf) had poor correlations (r < 0.5) with all other examined performance indicators. The Pearson correlation results in Table 15 also demonstrated that WeibullCRI and IDEAL-CT<sub>index</sub> had a strong correlation (r = 0.964); however, Weibull<sub>CRI</sub> had lower variability in the test

results (average COV = 6.6 percent) compared to IDEAL- $CT_{index}$  (average COV = 18.8 percent) as discussed in the section below. These results are very consistent with the results of ITD RP 280 (Kassem et al. 2021).

Spearman Coefficient	Weibull <sub>cri</sub> (IDT)	IDEAL - CT (IDT)	N <sub>flex</sub> (IDT)	CRI (IDT)	FI (IDT)	IDT <sub>Strength</sub>	<b>IDT<sub>Modulus</sub></b>	Fracture Energy (Gf)
Weibull <sub>CRI</sub> (IDT)	1							
IDEAL -CT (IDT)	0.964	1						
N <sub>flex</sub> (IDT)	0.971	0.983	1					
CRI (IDT)	0.989	0.963	0.983	1				
FI (IDT)	0.960	0.944	0.955	0.958	1			
IDT strength	-0.539	-0.547	-0.643	-0.632	-0.572	1		
IDTModulus	-0.741	-0.731	-0.808	-0.82	-0.756	0.937	1	
Fracture Energy (Gf)	0.494	0.428	0.341	0.392	0.411	0.445	0.147	1

Table 15. Pearson Coefficient (r) for Cracking Performance Indicators

Excellent Correlation (rs ≥ 0.9) Good Correlation (0.7 < rs < 0.9) Fair Correlation (0.5 < rs ≤ 0.7) Poor Correlation (0.1 < rs ≤ 0.5)

Figure 67 shows the coefficient of variation (COV) for each cracking performance indicator based on the results of 23 loose mixtures. The COV indicates that variability in test results in relation to the mean of the values. It is calculated as the ratio of the standard deviation to the mean. The results of COV presented in Figure 67 demonstrate that CRI, Weibull<sub>CRI</sub>, Nflex, FI, and IDEAL-CT<sub>Index</sub> had a COV of 6.0, 6.6, 12.5, 14.7, and 18.8 percent, respectively, while Gf, IDT<sub>Strength</sub>, and IDT<sub>Modulus</sub> had a COV of 4.2, 6.3, and 11.2 percent, respectively. Weibull<sub>CRI</sub> had lower COV compared to IDEAL-CT<sub>Index</sub> which is consistent with previous findings by Kassem et al. (2019) and Kassem et al. (2021).



Figure 67. Coefficient of Variation of Cracking Performance Indicators for PMLC Mixes

#### **Evaluation of Rutting Resistance of Field Mixes**

Figure 68 shows the HWTT rut depth for the examined 23 field projects. Four PMLC replicates were prepared from each mixture. A total of 92 PMLC specimens were prepared to evaluate the rutting resistance of the field mixtures. The HWTT measures the rut depth at 11 different locations along the pass of the HWTT wheels on the test specimens. The test is conducted at 50°C in wet conditions where the test HWTT specimens are submerged in water; therefore, the test can be used to assess both rutting resistance as well as moisture susceptibility of asphalt mixtures. The results of the HWTT rutting depth summarized in Figure 68 demonstrate that the rut depth after 20,000 passes for all field mixtures ranged from 1.12 mm to 4.41 mm. Therefore, all the mixtures had a rut depth way below the maximum threshold of 12.5 mm after 20,000 passes, and there was no statistically significant difference in rut depth among various projects except between Project No. 11 and Project No. 12. Project No. 11 had a RAP content of 31 percent, 5.3 percent binder content, and PG 64-34. The researchers observed that this mix (i.e., Project No. 11) was relatively drier and has more fines compared to other projects. It had the maximum rut depth of 4.41 mm. Project No. 12 had RAP content of 32 percent, binder content of 5.3 percent and PG 58-34, and had the minimum rut depth of 1.18 mm. Meanwhile, Project No. 11 and Project No. 12 were way below 12.5 mm after 20,000 passes. In addition, Project No. 6, No. 9, No. 10, No. 13, No. 17, No. 20, No. 22, and No. 23 had an average rut depth from 3 mm to 4 mm. Project No. 6, No. 13, No. 9, and No. 10 had RAP content ranging from 17 percent to 30 percent with binder content ranging from 4.9 percent to 5.8 percent, while Project No. 17, No. 20, and No. 22 had binder content ranging from 4.8 percent to 5.9 percent with 0 percent RAP. Project No. 1, No. 2, No. 3, No. 4, No. 5, No. 7, No. 8, No. 14, No. 15, No. 16, No. 18, No. 19, and No. 21 had an average rut depth between 2 mm to 3 mm. These mixtures had RAP content ranging from 0 percent to 35 percent while the binder content ranged from 4.9 percent to 6.2 percent. It should be noted that Project No. 11 was found to exhibit the lowest cracking resistance among all examined projects too. The HWTT results further demonstrated that there was no sign of moisture damage. Most of the examined projects had anti-strip agents of 0.5 percent except Project No. 1 and No. 13 where 0.75 percent of anti-strip agents was used. Project No. 9 and No. 16 did not have any anti-strip agents, while Project No. 8 had 0.25 percent anti-strip agents. Based on the results of this section, there is no concern on the rutting performance of asphalt mixtures currently produced in Idaho which is consistent with the findings of RP 261 and RP 280.



Figure 68. Hamburg Rutting Depth for PMLC Mixes

# 6. Cost Analysis

The researchers conducted cost analysis to assess economic savings associated with using rejuvenators with high RAP content in asphalt mixtures without compromising the performance (i.e., cracking and rutting resistance). The use of RAP can cut down the percent of virgin binder added, leading to cost savings. In addition, the current practice is to use one grade softer virgin binder if the RAP content between 17 percent to 30 percent and the blending chart to select the grade of softer virgin binder if RAP content exceeds 30 percent. The use of softer binder costs more due to its limited availability. Therefore, the use of rejuvenators may eliminate or reduce the need for softer binders.

In this study, the researchers evaluated mixtures with different sources of RAP. Some RAP sources had higher binder content (e.g., RAP No. 3 and RAP No. 4 had 5.37 and 5.3 percent, respectively) and others had lower binder content (e.g., RAP No. 3 had 4.3 percent). The researchers evaluated the cost of asphalt mixtures with and without RAP as well as RAP mixtures with and without rejuvenators. They examined the materials cost for asphalt mixtures prepared with three different RAP materials evaluated in this study (i.e., RAP No. 2, RAP No. 3, and RAP No. 4) incorporating the best performing rejuvenators for each RAP source. For RAP No. 2, the researchers studied the cost associated with using two rejuvenators (i.e., R1 and R5) with different RAP contents (i.e., 25, 50,70 percent) and at different binder grades for the 70 percent RAP mixtures. For RAP No. 3, they examined the cost associated with using two rejuvenators (i.e., R1 and R5) at a higher RAP content of 70 percent. In addition, the researchers examined the cost of mixtures with no RAP (i.e., 0 percent RAP) and mixtures with higher binder content. For RAP No. 4, the researchers studied the cost associated with using two rejuvenators (i.e., R1 and R5) at a higher RAP content of 70 percent. In addition, the researchers examined the cost of mixtures with no RAP (i.e., 0 percent RAP) and mixtures with higher binder content. For RAP No. 4, the researchers studied the cost associated with using two rejuvenators (i.e., R6 and R7) at different RAP content (i.e., 0, 25, 50, and 70 percent). The results of the cost analysis are presented and discussed for each RAP source separately in this chapter.

### **Materials Cost**

The researchers conducted a cost-benefit analysis based on the cost of rejuvenators, aggregates, asphalt binders obtained from the manufacturers or literature. Martin et al. (2020) estimated the material cost at around 45 to 50 percent, plant production cost at around 35 to 40 percent, and cost of field operations (i.e., hauling, laydown, and compaction) at around 15 to 20 percent of the total in-place HMA cost. In this study, the researchers examined the cost savings associated with materials only. The cost of rejuvenators is much higher than the cost of binders, virgin aggregates, and RAP materials as summarized in Table 16. Meanwhile, the percentage of rejuvenator used in asphalt mixtures is very small and taken as a small percent of asphalt binder by weight. The tall oil rejuvenator (R1) cost ranges from \$4,000 to \$4,900 per ton, waste vegetable oil rejuvenator (R5) is about \$3,800 per ton, the bio-based oil rejuvenator (R6) cost is about \$2,200 per ton, and the petroleum-based oil rejuvenator (R7) cost is about \$1,900 per ton. The cost of virgin binder ranges from \$750 to \$925 per ton depending on the binder grade. The virgin aggregate and RAP materials are much lower compared to the cost of rejuvenators and asphalt binders.

The cost per ton for virgin aggregates and RAP materials ranges from \$12 to \$15 and from \$5 to \$8, respectively.

		\$Pri	\$Price/ton		
Material Type	Material Description	Low	High		
Aggregate	Virgin Aggregates	12	15		
Aggregate	RAP Aggregates	5	8		
Virgin Binder	PG 58-28	750	-		
Virgin Binder	PG 58-34	875	-		
Virgin Binder	PG 64-28	800	-		
Virgin Binder	PG 64-34	925	-		
Virgin Binder	PG 70-28	825	-		
Rejuvenator (R1)	Tall oil	4000	4900		
Rejuvenator (R5)	Waste Vegetable Oil	3800	-		
Rejuvenator (R6)	Bio-Based Oil	2200	-		
Rejuvenator (R7)	Petroleum-Based Oil	1900	-		

#### **Table 16. Summary Materials Cost**

#### Cost Comparison Associated with RAP No. 2

The researchers calculated the materials cost for mixtures prepared with different percentages of RAP No.2, rejuvenators, and binder contents as shown in Figures 69 and Table 17. Figure 69 includes two "Y" axes, the one on the left for the cost in US dollar per ton of asphalt mixtures and the axis on the right is for the IDEAL-CT<sub>Index</sub>. The grey bars represent the IDEAL-CT<sub>Index</sub> and the colored bars represent the cost. The percentage of cost reduction or increase, with respect to the control mix, is written on the top of each bar corresponding to mixtures with different compositions. A negative sign indicates a cost reduction, while a positive sign demonstrates a cost increase. The following observations can be made based on the cost analysis:

- The control asphalt mixture (0 percent RAP) had an IDEAL-CT<sub>index</sub> of 39 which is moderate cracking resistance according to the cracking performance thresholds proposed in RP 261 (Kassem et al. 2019). Also, the control mixture costs about \$65 per ton.
- When the binder content was increased from 5.8 percent to 6.3 percent which represents 0.5 percent increase in OBC (i.e., 0 percent RAP + 0.5 BC), the IDEAL-CT<sub>Index</sub> was increased which indicates improved cracking resistance compared to the control mixture. The IDEAL-CT<sub>Index</sub> increased from 39 (moderate cracking resistance) to 102 (good cracking resistance). However, this resulted in a cost increase of 6.8 percent compared to the control mix.
- Asphalt mixtures prepared with 25 percent RAP (25 percent RAP) without rejuvenators resulted in a cost reduction of \$13 per ton which is approximately 21.6 percent reduction in the cost compared to the virgin mix (0 percent RAP). However, the performance of mixtures with 25 percent RAP had a lower IDEAL-CT<sub>Index</sub> (IDEAL-CT<sub>Index</sub> of 20) which demonstrated reduced cracking resistance compared to the control mixture (IDEAL-CT<sub>Index</sub> of 39).
- When the binder content was increased from 5.8 percent to 6.3 percent which represents 0.5 percent increase in OBC at 25 percent RAP (25 percent RAP + 0.5 BC), the cracking performance was improved (IDEAL-CT<sub>Index</sub> of 46) compared to the control mixture (IDEAL-CT<sub>Index</sub> of 39). In

addition, there was a net 14.8 percent reduction in the cost compared to the control mix (\$9 reduction per ton). These analyses indicate that increasing the binder content at a RAP content of 25 percent could be a cost-effective approach to incorporate RAP in the mix without compromising the performance (i.e., both cracking and rutting resistance). All mixtures passed the rutting requirements as discussed in Chapter 4. This further demonstrates the importance of implementing a balanced mix design approach when incorporating RAP in asphalt mixtures. This leads to cost savings as well as producing a mixture with comparable or improved performance.

- At 25 percent RAP, the use of tall oil (25 percent RAP + R1) resulted in improved IDEAL-CT<sub>index</sub> (IDEAL-CT<sub>Index</sub> of 52) compared to the mix with 25 percent RAP (IDEAL-CT<sub>Index</sub> of 20) and the control mix (IDEAL-CT<sub>Index</sub> of 39). However, this resulted in a 6.8 percent cost increase (\$5 increase per ton) compared to the control mix.
- At 25 percent RAP, the use of waste vegetable oil (25 percent RAP + R5) resulted in slightly higher IDEAL-CT<sub>Index</sub> (IDEAL-CT<sub>Index</sub> of 26) compared to mixture with 25 percent RAP (IDEAL-CT<sub>Index</sub> of 20); however, it was less than the control mixture (IDEAL-CT<sub>index</sub> of 39). In addition, there was an 8.5 percent cost reduction compared to the control mix.
- The cracking performance of mixtures prepared with 50 percent RAP and tall oil (50 percent RAP + R1) was comparable to the control mixture (i.e., IDEAL-CT<sub>Index</sub> of 33 compared to 39 for the control mixture). In addition, there was a cost reduction of 14.6 percent compared to the control mixtures. Mixtures prepared with 50 percent RAP and waste vegetable oil (50 percent RAP + R2) had lower IDEAL-CT<sub>Index</sub> (IDEAL-CT<sub>Index</sub> of 25) than the control mixture (IDEAL-CT<sub>Index</sub> of 39). In addition, increasing the binder content by 0.5 percent at 50 percent RAP (50 percent RAP+0.5BC) was not sufficient to improve the cracking resistance to that of the control mix. For the previous analysis, it can be concluded that using the tall oil is the most cost-effective alternative to improve the performance of asphalt mixtures prepared with 50 percent RAP compared to the other rejuvenator (waste vegetable oil) or increasing the binder content. Martin et al. (2019) reported 12 percent reduction in materials cost for mixtures prepared with 40 precent RAP which agrees with our study where there was 14.6 percent reduction in materials cost for mixtures with 50 percent RAP.
- Asphalt mixtures prepared with 70 percent RAP and rejuvenators exhibited higher cost savings than at 25 and 50 percent RAP. Asphalt mixtures with 70 percent RAP and tall oil (70 percent RAP + R1) had comparable or improved cracking performance (IDEAL-CT<sub>Index</sub> of 40) compared to the control mix (IDEAL-CT<sub>Index</sub> of 39). Furthermore, the use of tall oil resulted in the highest cost reduction of 30.4 percent compared to the cost of control mix. Similarly, asphalt mixtures with 70 percent RAP and waste vegetable oil (70 percent RAP + R2) also provided comparable or improved cracking performance (IDEAL-CT<sub>Index</sub> of 39) compared to the control mix (IDEAL-CT<sub>Index</sub> of 39). Meanwhile, the cost reduction (i.e., 23.1 percent) was less than that of the tall oil (i.e., 23.1 percent). Increasing the binder content by 0.5 percent (70 percent RAP + 0.5 BC) resulted in higher cost reduction (52.2 percent) and slightly lower cracking performance (IDEAL-CT<sub>Index</sub> of 30) to the control mix (IDEAL-CT<sub>index</sub> of 39) which demonstrates that the increasing the binder content could be very beneficial in some cases. This further emphasizes the effectiveness of the balanced mix design approach.



Figure 69. Cost Comparison of Mixtures with RAP No. 2 and Rejuvenators

Міх Туре	Cost (\$/ton)	Cost Difference (\$/ton)	Performance, IDEAL-CT <sub>Index</sub>
0% RAP	63	0	39
0% RAP +0.5 BC	67	4	102
25% RAP	50	-13	20
25%RAP +0.5 BC	54	-9	46
25% + R1	68	5	52
25% + R5	58	-5	26
50% RAP	36	-27	14
50%RAP +0.5 BC	41	-22	18
50% + R1	55	-8	33
50% + R5	53	-10	25
70% RAP	26	-37	14
70%RAP +0.5 BC	41	-22	30
70% + R1	44	-19	40
70% + R5	43	-20	39

Table 17. RAP No. 2 Cost Analysis Summary

### Cost Associated with RAP No. 3

The researchers also calculated the materials cost for mixtures prepared with 70 percent RAP, two rejuvenators, and binder contents as shown in Figures 70 and Table 18. The following observations can be made based on the cost analysis:

- The control asphalt mixture (0 percent RAP) had an IDEAL-CT<sub>Index</sub> of 49 which is moderate cracking resistance according to the cracking performance thresholds proposed in RP 261 (Kassem et al. 2019). Also, the control mixture costs about \$56 per ton.
- When the binder content was increased from 5.8 percent to 6.3 percent which represents 0.5 percent increase in OBC (i.e., 0 percent RAP + 0.5 BC), the IDEAL-CT<sub>index</sub> was increased from 43 (moderate cracking resistance) for the control mix to 102 (good cracking resistance) which is considered a significant improvement. However, this resulted in an increase in the cost by 7 percent compared to the control mix.
- The use of 70 percent RAP decreased the cracking resistance (IDEAL-CT<sub>index</sub> of 23) which indicated poor cracking resistance according to the thresholds presented in RP 261. Although the use of 70 percent RAP resulted in higher cost savings (i.e., 48 percent reduction), the cracking performance was not acceptable (IDEAL-CT<sub>index</sub> of 23 demonstrates poor cracking resistance).
- Increasing the binder content from 5.8 percent to 6.3 percent which represents 0.5 percent increase in OBC) combined with the use of 70 percent RAP (i.e., 70 percent RAP + 0.5 BC), resulted in improved cracking resistance (IDEAL-CT<sub>index</sub> of 72) compared to the control mix (IDEAL-CT<sub>index</sub> of 49). This approach resulted in a higher cost reduction of 42 percent.
- The use of both rejuvenators (i.e., tall oil and waste vegetable oil) improved the cracking resistance from poor performance for 70 percent RAP (IDEAL-CT<sub>Index</sub> of 23) to good performance (IDEAL-CT<sub>Index</sub> of 118 for tall oil and IDEAL-CT<sub>Index</sub> of 92 for waste vegetable oil) which is better than the control mix (IDEAL-CT<sub>Index</sub> of 49) at optimum binder content (0 percent RAP) and even to the virgin mix at higher binder content (0 percent RAP + 0.5 BC). The use of tall oil and waste vegetable oil resulted in cost reductions of 16 and 15 percent, respectively. These results further demonstrate the effectiveness of the use of rejuvenators at higher RAP contents which are consistent with the cost analysis of RAP No. 2.



Figure 70. Cost Comparison of Mixtures with RAP No. 3 and Rejuvenators

Mix Type	Cost (\$/ton)	Cost Difference (\$/ton)	Performance, IDEAL-CT <sub>Index</sub>
0% RAP	56	0	49
0% RAP +0.5 BC	60	4	102
70% RAP	29	-27	23
70%RAP +0.5 BC	33	-23	72
70% + R1	47	-9	118
70% + R5	48	-8	92

#### Table 18. RAP No. 3 Cost Analysis Summary

#### Cost Associated with RAP No. 4

The researchers also calculated the materials cost for mixtures prepared with different percentages of RAP No. 4 (i.e., 0, 25, 50, and 70 percent), two rejuvenators (i.e., R6 and R7), and one binder content as shown in Figures 71 and Table 19. In addition, the cost of mixtures prepared with 25 percent of RAP with softer binder (i.e., PG 64-34) was included. The following observations can be made based on the cost analysis:

• The control mix (i.e., 0 percent RAP) had an IDEAL-CT<sub>Index</sub> of 90 which illustrates good cracking resistance according to the cracking performance thresholds proposed in RP 261 (Kassem et al. 2019) and costs about \$57 per ton.

- Mixtures prepared with 25 percent RAP without rejuvenators resulted in 22 percent reduction in the cost; however, these mixtures had lower cracking resistance (IDEAL-CT<sub>Index</sub> of 59) compared to control mixture (IDEAL-CT<sub>Index</sub> of 90).
- The addition of rejuvenators R6 (at two doses of 6.6 and 8.3 percent by weight of reclaimed binder) and R7 (at 11.3 percent by weight of reclaimed binder) with 25 percent RAP resulted in a cost reduction of 19.4, 18.5, and 18.7 percent respectively. However, the use of R6 and R7 at 25 percent RAP resulted in reduced cracking performance compared to mixtures with 25 percent RAP without rejuvenators (IDEAL-CT<sub>index</sub> of 59). The IDEAL-CT<sub>index</sub> values for mixtures prepared with 25 percent RAP and 6.6 and 8.3 percent of R6 and 11.3 percent of R7 were 43, 56, and 54, respectively.
- The use of softer binder grade (i.e., PG 64-34) at 25 percent RAP didn't impact the cracking performance (IDEAL-CT<sub>Index</sub> of 59) compared to mixtures with 25 percent and PG 70-28 (IDEAL-CT<sub>Index</sub> of 60). Meanwhile, the cost increased by 6.4 percent as compared to the control mix (i.e., 0 percent) and 36 percent as compared to mixtures with PG 70-28.
- The use of 50 percent RAP resulted in a cost reduction of 43 percent compared to the control mix (0 percent RAP); however, this was associated with reduced cracking resistance (IDEAL-CT<sub>Index</sub> of 90 for the control mixture compared to IDEAL-CT<sub>Index</sub> of 59 for mixtures with 37 percent RAP).
- The use of R6 at a lower dose of 6.6 percent and a higher dose of 8.3 percent resulted in cost reduction of 38.2 and 38.0 percent compared to the control mixture without RAP or rejuvenators. However, this was associated with reduced cracking resistance (i.e., IDEAL-CT<sub>Index</sub> of 36 and 44 for the lower and higher doses of R6, respectively compared to IDEAL-CT<sub>Index</sub> of 90 for the control mixture). Similarly, mixtures prepared with 50 percent RAP and rejuvenator R7 showed 37.6 percent reduction in the total cost but with reduced IDEAL-CT<sub>Index</sub> of 33 compared to the control mixture of 90. These results showed that the use of both R6 and R7 were not effective at 50 percent RAP.
- The use of 70 percent RAP provided the maximum cost reduction of 60.9 percent; however, the cracking performance decreased from good (IDEAL-CT<sub>Index</sub> of 90 for the control mixture) to poor (IDEAL-CT<sub>Index</sub> of 24). The use of R6 at two doses (6.6 and 8.3 percent), and R7 at 70 percent RAP reduced the cost by 55.1, 46.0, and 45.4 percent, respectively. Meanwhile, the IDEAL-CT<sub>Index</sub> values for mixtures prepared with R6 and R7 at 70 percent RAP were much lower compared to the control mixture without RAP but were slightly higher compared to mixtures with 70 percent RAP. The use of 6.6 percent of R6 improved the cracking performance from poor for mixture with 70 percent RAP (IDEAL-CT<sub>Index</sub> of 24) to good (IDEAL-CT<sub>Index</sub> of 39).



Figure 71. Cost Comparison of Mixtures with RAP No. 4 and Rejuvenators

Міх Туре	Cost (\$/ton)	Cost Difference (\$/ton)	Performance, IDEAL-CT <sub>Index</sub>	
0% RAP	57	0	90	
25% RAP	44	-13	59	
25% RAP+ (R6) 6.6%	45.54	-11	43	
25% RAP+ (R6) 8.3%	46	-11	56	
25% RAP+ (R7)	46	-11	54	
25% RAP+ Softer Binder (PG 64-34)	60	3	60	
50% RAP	32	-25	37	
50% RAP+ (R6) 6.6%	35	-22	36	
50% RAP+ (R6) 8.3%	35	-22	44	
50% RAP+ (R7)	35	-22	33	
70% RAP	22	-35	24	
70% RAP+ (R6) 6.6%	25.37	-32	39	
70% RAP+ (R6) 8.3%	31	-26	33	
70% RAP+ (R7)	31	-26	29	

#### Table 19. RAP No. 4 Cost Analysis Summary

# 7. Conclusions and Recommendations

The researchers examined the performance of asphalt mixtures with different percentages of RAP. In addition, they evaluated the use of rejuvenators to improve the performance of asphalt mixtures through a balanced mix design approach that produces asphalt mixtures with sufficient resistance to cracking and rutting. The researchers prepared and tested mixtures prepared with different sources of RAP (i.e., RAP No. 1, RAP No. 2, RAP No. 3, and RAP No. 4) as well as different RAP contents (e.g., 0, 25, 50, 70 percent RAP). Seven unique commercially available rejuvenators (i.e., R1 though R7) were acquired and included in the testing program. These rejuvenators were tall oil, aromatic extract, bio-based forestry, engineered product, triglycerides and fatty acids product, bio-based oil, petroleum-based oil for R1 through R7, respectively. Asphalt mixtures prepared employing different doses of each rejuvenator were examined as discussed in Chapter 3. Furthermore, the researchers prepared and tested asphalt mixtures at different binder contents (i.e., optimum binder content [OBC] and OBC+0.5 percent) and binder grades (e.g., PG 58-28, PG58-34, PG, PG 64-28, PG 64-34, and PG 70-28).

Also, the researchers prepared Plant-Mixed Laboratory-Compacted (PMLC) test specimens obtained from new ITD paving projects. Loose asphalt mixtures from 23 projects were obtained and tested to evaluate the cracking and rutting performance of asphalt mixtures currently produced and used in the state. These projects were distributed across the six districts of the state (District 1 to District 6). The mixtures included different mix designs, binder grades, percent of RAP content, and percent of RAP binder replacement as discussed in detail in Chapter 3. The researchers evaluated the performance of the test mixtures based on the performance thresholds developed in RP 261 and the ones proposed in the respective standards and literature.

The researcher conducted several laboratory tests to evaluate the cracking and rutting performance of asphalt mixtures examined in this study. The cracking tests included the Indirect Tensile (IDT) Strength Test in accordance with ASTM D8225 to evaluate the intermediate cracking resistance. In addition, the researchers evaluated the low temperature cracking performance of selected mixtures by measuring the creep compliance and strength of the test mixtures at low temperature in accordance with AASHTO T 322. In addition, the researchers examined the rutting performance of the test mixtures in accordance with AASHTO T 324. The main findings from this study are summarized below.

#### Evaluation of the Performance of Asphalt Mixtures Prepared with RAP and Rejuvenators

- The cracking resistance decreased with the increase of RAP content. Both IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub> decreased with the increase of RAP content which demonstrates reduced cracking resistance. Meanwhile, the IDT<sub>strength</sub> increased with the increase of RAP content which demonstrates that the mixtures become stiffer with the addition of RAP.
- The use of rejuvenators in mixtures with low RAP content (e.g., 25 percent), especially for mixtures with good cracking performance, didn't improve the cracking resistance (i.e., did not

increase IDEAL-CT<sub>Index</sub>). In fact, it was observed that the addition of some rejuvenators (i.e., R4 and R6) could be detrimental to the cracking resistance at low RAP content for mixtures with good cracking resistance at low RAP content.

- Mixtures prepared with 70 percent RAP had statistically significant lower IDEAL-CT<sub>Index</sub> values compared to the ones without RAP (0 percent RAP) irrespective of the binder grade.
- The favorable effect of rejuvenators in asphalt mixtures is observed in mixtures with higher RAP content (e.g., 70 percent) for different RAP sources evaluated in this study. In some cases (e.g., RAP No. 2 and RAP No. 3), it was possible to produce mixtures prepared with 70 percent RAP and rejuvenators that provided comparable cracking performance to the mixture without RAP.
- The use of rejuvenator R1 (tall oil) and rejuvenator R5 (waste vegetable oil) with mixtures with high RAP contents provided the best performance compared to other rejuvenators examined in this study, and these mixtures had comparable cracking performance to the virgin mix (i.e., 0 percent RAP).
- The binder PG did not affect the IDEAL-CT<sub>Index</sub> for mixtures without RAP (0 percent RAP) and those prepared with 70 percent RAP. However, the IDT<sub>Strength</sub> for mixtures prepared with PG 64-28 and PG 70-28 was higher compared to the ones for PG 58-34 for the mixtures without RAP. The results of IDT<sub>Strength</sub> were comparable for mixtures of different binder PG at higher RAP content (70 percent RAP) which demonstrates stiffer mixtures. Also, the rejuvenator R1 provided better results with softer binders, likely due to better compatibility between that product and the binder used.
- The rejuvenator R4 (engineered product) at a higher dose improved the cracking performance of mixtures with RAP; however, these mixtures failed the rutting criteria prematurely (i.e., the mixtures were over softened). These results demonstrated the importance of following a balanced mix design (BMD) approach to satisfy both cracking and rutting criteria.
- Depending on the RAP source and aging conditions, rejuvenators could improve the cracking performance significantly.
- Increasing the binder content was found to increase the cracking resistance for some mixtures with or without RAP. This further emphasizes the effectiveness of following BMD approach.
- The use of rejuvenators R1 and R5 could slightly increase the rut depth compared to the control mixtures (i.e., 70 percent RAP); however, such increase is not statistically significant. All mixtures prepared with these rejuvenators passed the rutting criteria.
- All mixtures with and without rejuvenators prepared with RAP No. 4 provided good rutting resistance and there was no sign of moisture damage or stripping and the rut depth for all mixture was under 4 mm. Meanwhile, the use of RAP in the mixtures tends to decrease the rut depth.
- The use of rejuvenator R1 (tall oil) and rejuvenator R5 (waste vegetable oil) provided higher creep compliance compared to the virgin mixture (0 percent RAP) and control mixture (70 percent RAP) which demonstrated improved cracking resistance at a low temperature.
- Some rejuvenators (e.g., R1 and R5) were highly effective in improving cracking performance of asphalt mixtures with higher RAP content, which offers significant environmental and economic benefits. Furthermore, these rejuvenators enhanced the thermal cracking performance.

There was a good agreement between the results of IDT<sub>strength</sub> (measured at -10 °C) and that of • the creep compliance test. The results further demonstrated the favorable impact of rejuvenators on improving the performance of asphalt mixtures with high RAP content at low temperature.

#### **Evaluation of Cracking and Rutting Performance of Plant Mixtures**

- The researchers examined the cracking and rutting performance of 23 loose mixtures collected from new paving projects. These loose mixtures were collected from different districts in Idaho and have different properties.
- Most of the field mixes (18 out of 23) exceeded the proposed IDEAL-CT<sub>Index</sub> threshold of 73.7 which indicates that these mixtures are expected to exhibit good cracking resistance in the field. Four projects were within the moderate cracking performance range, while only one project had an IDEAL-CT<sub>Index</sub> less than the minimum proposed threshold of 26.4 which demonstrates poor cracking resistance.
- Based on the cracking performance results of IDEAL-CT<sub>Index</sub> and Weibull<sub>CRI</sub> and the corresponding thresholds proposed in RP 261, there is no concern with the expected cracking performance of most asphalt mixtures currently produced in Idaho.
- The results demonstrated that IDEAL-CT<sub>index</sub>, Weibull<sub>CRI</sub>, N<sub>Flex</sub>, CRI, and FI had direct strong correlations (r > 0.90). Also, the results showed that both IDT<sub>strength</sub> and IDT<sub>Modulus</sub> had an inverse correlation with most indicators (i.e., that IDEAL-CT<sub>index</sub>, Weibull<sub>CRI</sub>, N<sub>Flex</sub>, CRI, and FI), except between each other.
- The Pearson correlation results demonstrated that Weibull<sub>CRI</sub> and IDEAL-CT<sub>Index</sub> had a strong correlation (r = 0.964); however, Weibull<sub>CRI</sub> had lower variability in the test results (average COV = 6.6 percent) compared to IDEAL- $CT_{index}$  (average COV = 18.8 percent) which is consistent with the results of ITD RP 280 (Kassem et al. 2021).
- The HWTT rut depth for the field mixes ranged from 1.12 mm to 4.41 mm after 20,000 passes. Therefore, all the mixtures had a rut depth way below the maximum threshold of 12.5 mm after 20,000 passes.
- The HWTT results further demonstrated that there was no sign of moisture damage. Most of the examined projects had anti-strip agents of 0.5 percent except Project No. 1 and No. 13 where 0.75 percent was used.
- Based on the HWTT results, there is no concern on the rutting performance of asphalt mixtures currently produced in Idaho which is consistent with the findings of RP 261 and RP 280.

#### Economic Benefits of Incorporating RAP and Rejuvenators in Asphalt Mixtures

- The researchers conducted cost analysis to assess economic savings associated with using • rejuvenators with high RAP content in asphalt mixtures without compromising the performance (i.e., cracking and rutting resistance).
- At 25 percent RAP, increasing the binder content was more effective than using rejuvenators in terms of cracking performance and associated cost reduction. This leads to cost savings as well as producing mixtures with comparable or improved performance.
- At 50 percent RAP, the use of rejuvenator R1 (tall oil) was the most cost-effective alternative to improve performance as compared to the other rejuvenators including R2 (waste vegetable oil)

or increasing the binder content. In addition, R1 provided comparable cracking performance to that of the virgin mixture.

• At a higher percentage of RAP (e.g., 70 percent), the use of rejuvenators (especially R1) was very effective in improving the cracking resistance with associated cost savings.

#### **Recommendations and Implementation**

The results of this study demonstrated that the use of rejuvenators in asphalt mixtures with RAP is beneficial and could offer environmental benefits and cost savings. However, it is more cost effective to incorporate rejuvenators in mixtures with high RAP content (e.g., 50 or 70 percent). The cracking performance of mixtures with certain rejuvenators and high RAP content could be comparable to that of the virgin mixture (0 percent RAP) with additional cost savings. At a low RAP content, increasing the binder content could be more effective in improving the cracking resistance and reducing the cost of asphalt mixtures as compared to that of the virgin mixtures. In addition, the use of rejuvenators R1 (tall oil) and R2 (waste vegetable oil) were found effective in improving the cracking resistance and providing cost savings compared to other rejuvenators. Furthermore, different doses of rejuvenators should be evaluated at the mixture level to select the optimum dose based on cracking performance.

Based on the cracking and rutting assessment of 23 PMLC projects, it is recommended to implement the performance thresholds for cracking and rutting proposed in RP 261 and further evaluated in RP 280 and this study (RP 292) in assessing the performance of asphalt mixtures produced in the state as well as designing new mixtures using a BMD approach. The results of this study clearly demonstrate the importance of implementing a BMD approach to optimize the design of asphalt mixtures prepared with RAP and rejuvenators to provide adequate performance in terms of cracking and rutting resistance. Furthermore, the balanced mix design should be supplemented by conducting a cost-benefit analysis to compare different alternatives that provide acceptable performance.

It is recommended to construct trial sections in the field. These sections should be constructed using asphalt mixtures designed based on a BMD approach. The performance thresholds proposed in RP 261 and evaluated in this study can be adopted to ensure adequate resistance to cracking, rutting as well as moisture damage. The asphalt mixtures used in these sections should be prepared with RAP and rejuvenators. The performance of these sections should be monitored overtime to evaluate the need to revise or adjust the performance thresholds proposed for the BMD approach. In addition, field cores should be extracted and tested in the laboratory to examine the correlation between laboratory tests and field performance.

# 8. Cited Work

- Ali, Ayman W., Yusuf A. Mehta, Aaron Nolan, Caitlin Purdy, and Thomas Bennert. 2016. "Investigation of the impacts of aging and RAP percentages on effectiveness of asphalt binder rejuvenators." *Construction and Building Materials* 110 (2016): 211-217. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2016.02.013</u>.
- Alkuime, Hamza, Emad Kassem, Fouad MS Bayomy, and Richard J. Nielsen. 2020. "Development of a new performance indicator to evaluate resistance of asphalt mixes to cracking." *Journal of Transportation Engineering, Part B: Pavements* 146, no. 4 (2020): 04020072. Accessed March 15, 2021. <u>https://ascelibrary.org/doi/full/10.1061/JPEODX.0000224</u>.
- Arámbula-Mercado, Edith, Fawaz Kaseer, Amy Epps Martin, Fan Yin, and Lorena Garcia Cucalon. 2018.
   "Evaluation of recycling agent dosage selection and incorporation methods for asphalt mixtures with high RAP and RAS contents." *Construction and Building Materials* 158 (2018): 432-442. Accessed March 18, 2021. https://doi.org/10.1016/j.conbuildmat.2017.10.024.
- Bajaj, Akash, Amy Epps Martin, Gayle King, Charles Glover, Fawaz Kaseer, and Edith Arámbula-Mercado.
   2020. "Evaluation and classification of recycling agents for asphalt binders." *Construction and Building Materials* 260 (2020): 119864. Accessed March 20, 2021.
   <a href="https://doi.org/10.1016/j.conbuildmat.2020.119864">https://doi.org/10.1016/j.conbuildmat.2020.119864</a>.
- Beeson, Matt, Mike Prather, and Gerald A. Huber. 2011. "Characterization of reclaimed asphalt pavement in Indiana: Changing INDOT specifications for RAP." Presented at Transportation Research Board 90th Annual Meeting (No. 11-1055), Washington, DC, 2011. Accessed March 13, 2021
- Bennert, Thomas, Navneet Garg, Christopher Ericson, and Nick Cytowicz. 2022. "Evaluation of Test Methods to Identify Asphalt Binders Prone to Surface-initiated Cracking." *Transportation Research Record* (2022). Accessed October 08, 2022. <u>https://doi.org/10.1177/03611981221119191</u>.
- Diefenderfer, Stacey D., and Benjamin F. Bowers. 2019. "Initial approach to performance (balanced) mix design: The Virginia experience." *Transportation Research Record* 2673, no. 2 (2019): 335-345. Accessed March 20, 2021. <u>https://doi.org/10.1177/0361198118823732</u>
- Dong, Wenlong, and Stephane Charmot. 2019. "Proposed tests for cold recycling balanced mixture design with measured impact of varying emulsion and cement contents." *Journal of Materials in Civil Engineering* 31, no. 2 (2019): 04018387. Accessed March 20, 2021. <u>https://doi.org/10.1061/(ASCE)MT.1943-5533.0002611</u>.
- Federal Highway Administration. 2019. Implement Balanced Asphalt Mix Design in Oklahoma. Cross, Stephen A., and Joshua Li. FHWA-OK-19-01 2305. Accessed October 13, 2021. <u>https://rosap.ntl.bts.gov/view/dot/57426</u>.

- Idaho Transportation Department. 2018. 2018 ITD Standard Specifications for Highway Construction. Accessed Dec 17, 2022. <u>https://apps.itd.idaho.gov/apps/manuals/specbook/specbook18.pdf</u>.
- Idaho Transportation Department. 2022. RP261 *Development and evaluation of performance measures to augment asphalt mix design in Idaho*. No. FHWA-ID-19-261. Accessed Jan 17, 2022. <u>https://apps.itd.idaho.gov/apps/research/Completed/RP261.pdf</u>.
- Idaho Transportation Department. 2022. RP280 *Development of Gyratory Stability Index to Evaluate Variation of RAP Content and Rutting Resistance of Asphalt Mixtures.* No. FHWA-ID-21-280. Accessed Jan 21, 2023. <u>https://apps.itd.idaho.gov/apps/research/Completed/RP280.pdf</u>
- Im, Soohyok, Fujie Zhou, Robert Lee, and Tom Scullion. 2014. "Impacts of rejuvenators on performance and engineering properties of asphalt mixtures containing recycled materials." *Construction and Building Materials* 53 (2014): 596-603. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2013.12.025</u>.
- Im, Soohyok, Pravat Karki, and Fujie Zhou. 2016. "Development of new mix design method for asphalt mixtures containing RAP and rejuvenators." *Construction and Building Materials* 115 (2016): 727-734. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2016.04.081</u>.
- Kaseer, Fawaz, Amy Epps Martin, and Edith Arámbula-Mercado. 2019. "Use of recycling agents in asphalt mixtures with high recycled materials contents in the United States: A literature review." *Construction and Building Materials* 211 (2019): 974-987. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2019.03.286</u>.
- Kaseer, Fawaz, Edith Arámbula-Mercado, Lorena Garcia Cucalon, and Amy Epps Martin. 2020.
  "Performance of asphalt mixtures with high recycled materials content and recycling agents." *International Journal of Pavement Engineering* 21, no. 7 (2020): 863-877. Accessed March 15, 2021. <u>https://doi.org/10.1080/10298436.2018.1511990</u>.
- Kassem, Emad, Eyad Masad, Robert Lytton, and Arif Chowdhury. 2011. "Influence of air voids on mechanical properties of asphalt mixtures." *Road Materials and Pavement Design* 12, no. 3 (2011): 493-524. Accessed March 15, 2021. <u>https://doi.org/10.1080/14680629.2011.9695258</u>.
- Kassem, Emad, Fouad Bayomy, S. J. Jung, Hamza Alkuime, and Fahmid Tousif. *Development and evaluation of performance measures to augment asphalt mix design in Idaho*. No. FHWA-ID-19-261.
   Idaho. Transportation Department, 2019. Accessed March 15, 2021.
   <a href="https://apps.itd.idaho.gov/apps/research/Completed/RP261.pdf">https://apps.itd.idaho.gov/apps/research/Completed/RP261.pdf</a>.
- Krcmarik, Michael, Sudhir Varma, M. Emin Kutay, and Anas Jamrah. 2016. "Development of predictive models for low-temperature indirect tensile strength of asphalt mixtures." *Journal of Materials in Civil Engineering* 28, no. 11 (2016): 04016139. Accessed March 15, 2021. <u>https://doi.org/10.1061/(ASCE)MT.1943-5533.0001655</u>.

- Lombardo, Jessica. "Why the Asphalt Industry Should Move to a Balanced Mix Design." Construction Pros, 2019. Accessed March 15, 2022. <u>https://www.forconstructionpros.com/asphalt/article/21102480/why-the-asphalt-industry-should-move-to-a-balanced-mix-design</u>.
- Majidifard, Hamed, Nader Tabatabaee, and William Buttlar. 2019."Investigating short-term and longterm binder performance of high-RAP mixtures containing waste cooking oil." *Journal of Traffic and Transportation Engineering (English Edition)* 6, no. 4 (2019): 396-406. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.jtte.2018.11.002</u>.
- Martin, Amy Epps, Fawaz Kaseer, Edith Arámbula-Mercado, Akash Bajaj, Lorena Garcia Cucalon, Fan Yin, Arif Chowdhury et al. 2020. *Evaluating the effects of recycling agents on asphalt mixtures with high RAS and RAP binder ratios*. National Academies of Science, Engineering, and Medicine. No. Project 09-58. 2020. Accessed March 15, 2021.
- McDaniel, Rebecca S., Ayesha Shah, Gerald A. Huber, and Audrey Copeland. 2012. "Effects of reclaimed asphalt pavement content and virgin binder grade on properties of plant produced mixtures." *Road Materials and Pavement Design* 13, no. sup1 (2012): 161-182. Accessed March 15, 2021. <u>https://doi.org/10.1080/14680629.2012.657066</u>.
- Meroni, Fabrizio, Gerardo W. Flintsch, Brian K. Diefenderfer, and Stacey D. Diefenderfer.2020.
   "Application of balanced mix design methodology to optimize surface mixes with high-RAP content." *Materials* 13, no. 24 (2020): 5638. Accessed March 15, 2021.
   <a href="https://doi.org/10.3390/ma13245638">https://doi.org/10.3390/ma13245638</a>.
- Minnesota Department of Transportation. 2002. Investigation of recycled asphalt pavement (RAP) mixtures. No. MN/RC-2002-15. Accessed Jan 17, 2022. <u>https://cts-d8resmod-prd.oit.umn.edu/pdf/mndot-2002-15.pdf</u>.
- Minnesota Department of Transportation. 2022. Asphalt mix rejuvenators synthesis. No. NRRA202002. Accessed Nov 23, 2023. <u>http://www.dot.state.mn.us/research/reports/2020/NRRA202002.pdf</u>.
- Nabizadeh, Hesamaddin, Hamzeh F. Haghshenas, Yong-Rak Kim, and Francisco Thiago Sacramento Aragão. 2017. "Effects of rejuvenators on high-RAP mixtures based on laboratory tests of asphalt concrete (AC) mixtures and fine aggregate matrix (FAM) mixtures." *Construction and Building Materials* 152 (2017): 65-73. Accessed March 15, 2021. https://doi.org/10.1016/j.conbuildmat.2017.06.101.
- NCAT (National Center for Asphalt Technology). "NCAT researchers explore multiple uses of rejuvenators." *Asphalt Technol. News* 26, no. 1 (2014): 7-8. Accessed May 12, 2021
- Safi, Fazal R., Kamal Hossain, Shenghua Wu, Imad L. Al-Qadi, and Hasan Ozer. 2018."A model to predict creep compliance of asphalt mixtures containing recycled materials." *Construction and Building*
*Materials* 184 (2018): 374-381. Accessed March 15, 2021. https://doi.org/10.1016/j.conbuildmat.2018.06.232.

- Samara, Mahmoud, Daniel Offenbacker, Yusuf Mehta, Ayman Ali, Mohamed Elshaer, and Christopher Decarlo. 2022."Performance Evaluation and Characterization of Extracted Recycled Asphalt Binder With Rejuvenators." *Transportation Research Record* 2676, no. 12 (2022): 714-727. Accessed January 8, 2023. <u>https://doi.org/10.1177/03611981221097091</u>.
- Sánchez, D. B., S. Caro, and A. E. Alvarez.2020. "Determining optimum doses of palm oil rejuvenators for recycled blends." In Advances in Materials and Pavement Performance Prediction II, pp. 339-342. CRC Press, 2020. Accessed March 15, 2021. <u>https://doi.org/10.1201/9781003027362</u>.
- Sirin, Okan, Md Ohiduzzaman, Emad Kassem, and Dalim K. Paul. 2020. "Comprehensive evaluation of long-term aging of asphalt mixtures in hot climatic condition." *Road Materials and Pavement Design* 21, no. 4 (2020): 927-949. Accessed March 15, 2021. https://doi.org/10.1080/14680629.2018.1531777.
- West, Randy, Fan Yin, Carolina Rodezno, and Adam Taylor. 2021. Balanced Mixture Design Implementation Support. No. WHRP 0092-20-04. Wisconsin. Dept. of Transportation. Research and Library Unit, 2021. Accessed Jan 17, 2022. <u>https://wisconsindot.gov/documents2/research/0092-20-04-final-report.pdf</u>.
- Williams, Brett A., J. Richard Willis, and Joseph Shacat. *Asphalt pavement industry survey on recycled materials and warm-mix asphalt usage: 2021*. No. IS 138 (12e). Accessed February 12, 2023.
- Yin, Fan, and Randy C. West. 2021. "Balanced Mix Design Resource Guide." National Asphalt Pavement Association No. IS-143. 2021. Accessed February 12, 2023. <u>https://www.eng.auburn.edu/research/centers/ncat/files/NAPA-IS-143-BMD-Resource-Guide-2021.pdf</u>.
- Yin, Fan, Fawaz Kaseer, Edith Arámbula-Mercado, and Amy Epps Martin. 2017. "Characterising the longterm rejuvenating effectiveness of recycling agents on asphalt blends and mixtures with high RAP and RAS contents." *Road Materials and Pavement Design* 18, no. sup4 (2017): 273-292. Accessed March 15, 2021. <u>https://doi.org/10.1080/14680629.2017.1389074</u>.
- Yin, Fan, Edith Arambula, Robert Lytton, Amy Epps Martin, and Lorena Garcia Cucalon. "Novel method for moisture susceptibility and rutting evaluation using Hamburg wheel tracking test." Transportation Research Record 2446, no. 1 (2014): 1-7. Accessed November 20,2023. <u>https://doi.org/10.3141/2446-01</u>

- Zaumanis, Martins, Rajib B. Mallick, and Robert Frank. 2015. "Evaluation of different recycling agents for restoring aged asphalt binder and performance of 100% recycled asphalt." *Materials and Structures* 48 (2015): 2475-2488. Accessed March 15, 2021. <u>https://doi.org/10.1617/s11527-014-0332-5</u>.
- Zaumanis, Martins, Rajib B. Mallick, and Robert Frank.2013. "Evaluation of rejuvenator's effectiveness with conventional mix testing for 100% reclaimed Asphalt pavement mixtures." *Transportation research record* 2370, no. 1 (2013): 17-25. Accessed March 15, 2021. <u>https://doi.org/10.3141/2370-03</u>.
- Zaumanis, Martins, Rajib B. Mallick, Lily Poulikakos, and Robert Frank. 2014. "Influence of six rejuvenators on the performance properties of Reclaimed Asphalt Pavement (RAP) binder and 100% recycled asphalt mixtures." *Construction and Building Materials* 71 (2014): 538-550. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2014.08.073</u>.
- Zhou, Fujie, Richard Steger, and Walaa Mogawer. 2021. "Development of a coherent framework for balanced mix design and production quality control and quality acceptance." *Construction and Building Materials* 287 (2021): 123020. Accessed October 20, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2021.123020</u>.
- Zhou, Fujie, Sheng Hu, and Dave Newcomb. 2020. "Development of a performance-related framework for production quality control with ideal cracking and rutting tests." *Construction and Building Materials* 261 (2020): 120549. Accessed March 15, 2021. <u>https://doi.org/10.1016/j.conbuildmat.2020.120549</u>.
- Zhou, Fujie, Sheng Hu, and Tom Scullion. Integrated asphalt (overlay) mixture design, balancing rutting and cracking requirements. No. FHWA/TX-06/0-5123-1. Texas Transportation Institute, Texas A & M University System, 2006. Accessed December 17, 2021. <u>https://static.tti.tamu.edu/tti.tamu.edu/documents/0-5123-1.pdf</u>.
- Zhou, Fujie. "Development of an IDEAL cracking test for asphalt mix design, quality control and quality assurance." *NCHRP-IDEA Program Project Final Report* 195 (2019). Accessed December 17, 2021. <u>https://doi.org/10.1080/14680629.2017.1389082</u>.

# **Appendix A. RAP Properties**

	Gyratory Model # / Compactor: Serial #	AFG2AS 8436	Job Mix	Formula	Spec
1	Percent	Asphalt by Weight of Total Mix	5.	3	
2	Percent A	sphalt by Weight of Aggregate	5.6	34	
3	1	/irgin Asphalt by Weight of Mix	4.0	08	
4	Virgin A	sphalt by Weight of Aggregate	4.3	32	
5		Percent Air Volds (Pa)	4.	0	4.0
6	Void	ds in Mineral Aggregate (VMA)	15	.6	14 min
7	Cor	mpacted Unit Weight Gmb, pcf	2.366	147.3	
8	Theoretics	al Maximum Density Gmm, pcf	2.465	153.4	
9	Percent E	ffective Asphalt Content (Pbe)	5.0	)4	
10	Pe	ercent Absorbed Asphalt (Pba)	0.3	32	
11	:	Specific Gravity of Binder (Gb)	1.0	28	
12	Percent	Gmm @ N Initial (7 Gyrations)	86	.8	≤ 89.0
13	Percent Grr	nm @ N Design (75 Gyrations)	96.	.0	96.0
14	Percent G	mm @ N Max (115 Gyrations)	97.	.3	≤ 98.0
15		Dust to Asphalt Ratio (DP)	1.1	1	0.6-1.4
16		Percent Passing #200 Sieve	5.6	6	2.0-10.0
17		Voids Filled w/ Asphalt (VFA)	74	1	65-75
18	Laboratory Mixin	g Temperature for Design (*F)	32	0	316-324
19	Laboratory Compaction	n Temperature for Design (*F)	29	9	295-303
20	Laboratory Sample We	ight for Volumetric Testing (g)	472	20	-
21	Ignition Oven (NCAT	T) Correction Factor @ 538 °F	0.3	0	
22	*Lo	s Angeles Abrasion (LAR) (%)	24	1	30 max
23		*Idaho Degradation ∆ % -200	3.5	5	5.0 max
24		Sand Equivalent	66	3	40 min
25		*Fracture Face Count (%)	98/9	96	75/60
26		Fine Aggregate Angularity (%)	47.	2	40 min
27	*Flat and Elongated Partic	des in Coarse Aggregates (%)	0.3	3	10 max
	Red	cycled Asphalt Pavement (RAP	) Propertie	s	
28	Percentage	of Asphalt in RAP (Wt. of Mix)	4.2	0	
29	Percentage of RAP	by Total Weight of Aggregate	30		
30	Percent of RAP Bin	der by Weight of Total Binder	24		30 max
31		RAP Contribution by Mix	1.2	6	-
32	R	AP Contribution by Aggregate	1.3	2	
33		RAP NCAT Correction Factor	0.3	6	

\*Composite blend including RAP

## Figure A.1: RAP No. 1

										283	Lumb .	20500
					Project	Number:	A020(8	590)	PI	oject Key I	vumber:	20590
Asphalt Mix Type:	19 mm	SP3 (De	sign ES	ALs, 1<1	0)		Mix	Producer	7			
JMF Identification:	20036-1	9mmSF	3-R30				Sp	ecified Oil	PG 58	-34		
Binder Supplier:	Idaho A	sphalt	Supply				A	ljusted Oil	PG 58	-34		
Anti-Strip:	Morlife	5000				Ag	. Sourc	e Number	Bu-26	s		
		1	Job Mix	Formula	Targets	and Vo	lumetrie	c Data		18.001		
Target Oil Content:	5.0		RAPP	ercent B	y Binder:	30		J	MF G mb	Specimen	Weight:	4675
Virgin Oil Added by Mix:	3.5		RAP	Oil Con	tribution:	1.5		J	MF Ncat	Correction	Factor:	0.40
Virgin Oil Added by Agg:	3.7	(	Compacti	on Temp	Range:	288-302	2°F	L	ab Com	paction Te	mp (F°):	292
Percent Anti-Strip:	0.75		Mixi	ng Temp	Range:	315-32	•F	Bi	nder Spe	ecific Gravi	ty (G ):	1.0293
State State State State	a state of		Aggrega	te, RAP	and Bin	der Spec	ific Gra	vities	and the second		1000 miles	Carlo The
Combined Gsb:	2.585	Effect	ive Speci	fic Gravil	ty (G se):	2.624		D6	RAP Bu	lk Specific	Gravity:	2.589
Fine Aggregate Gsb:	2.562	F	ine SSD	Specific	Gravity :	2.606		IF	RAP Bu	Ik Specific	Gravity:	2.605
Coarse Aggregate Gsb:	2.594	Coa	rse SSD	Specific	Gravity :	2.621	1	Fine Appa	rent Spe	cific Gravit	y (G 10):	2.679
Fine Agg. % Absorption	1.6	C	coarse Ag	19. % Ab.	sorbtion:	1.1	Coar	se Appare	nt Spec	ific Gravity	(G):	2.667
			IF RAP C	Correction	n Factor:	0.08			D6 RAP	<sup>o</sup> Correctio	n Factor	0.07
Markey Status Provide Land			Job Mix	Formula	Target	Volumet	ric Prop	erties	and the second			
				Design	Spec.						Design	Spec.
Maximum Theoretica	al Specifi	c Gravit	(G mm):	2.435	-			Bulk Spec	ific Grav	ity (G mb):	2.338	
	Percer	nt Air Vo	ids (Pa):	4.0	2.5-5.0		Voids I	n Mineral	Aggrega	te (VMA):	14.1	13.0 mir
Voids	Filled wit	h Aspha	It (VFA):	72	65-75			Dus	st Propor	tion (DP):	1.2	0.6-1.4
	Absorbe	d Apsha	alt (P ba):	0.59	-		Effe	ctive Aspl	alt Cont	ent (P be):	4.44	
Den	sity Perc	ent G mm	@ Nini:	97.6	≤89.0		De	nsity Perc	ent G mm	@ Ndes:	96	96.0
Densit	y Percen	t G mm	D Nmax:	97.3	≤98.0	A	verage	Rut Depth	@15,00	00 Passes	5.2	10.0 ma
Uncompacted Void C	ontent of	Fine Ag	gregate:	51	40 min				Sand E	quivalent:	60	40 min
racture Face Coarse Ag	gregate	(1 Face	2 Face):	98/98	75/60		F	lat and El	ongated	Particles:	0	10 max
						1.1.1	Average	DEAL-C	T Crack	ing Index:	123	80 min
Constraint Second Second	and the second	J	MF Targe	et Aggre	gate, Br	eakdown	, Prope	ortions	1 100	The second s	a second	
Blend Percentages:		A-Pile:	12.0	B-Pile:	21.0	C-Pile:	37.0	IF RAP	20.0	D6 RAP:	10.0	
and an interport of		lob Mix	Formula	Aggreg	ate Grad	dations &	Stock	pile Avera	iges	Contract of the		Same and
	1"	3/4"	1/2"	3/8"	#4	#8	#16	#30	#50	#100	#200	
JMF Specifications:		100	90-100	90 max		28-58					2.0-10.0	
JMF Target Gradation:	100	99	85	73	48	32	23	17	12	8.0	5.2	
Ncat C.F. Gradation:	100.0	98.3	84.4	72.7	49.2	30.7	23.2	16.9	11.9	7.2	4.2	
Agg. Correction Factor:	0.0	0.7	0.6	0.3	-1.2	1.3	-0.2	0.1	0.1	0.8	1.0	
A-Pile	100	91	21	5	2	1	1	1	1	1	0.6	
B-Pile	100	100	80	45	5	2	2	1	1	1	0.7	
Breakdown:	100	100	100	100	100	100	100	100	100	100	90.0	
C-Pile:	100	100	100	100	80	53	35	24	16	10	7.3	
IF Category 1 RAP:	100	100	96	88	62	43	33	28	22	14	7.6	
the second data where the second data was not as a second data where the second data was a second data where the											1.	

Figure A.2: RAP No. 2

	Gyratory Model # AFG2AS Compactor: Serial # 8732	Job Mix	Formula	Spec	
1	Percent Asphalt by Weight of Total Mix	5	.1		
2	Percent Asphalt by Weight of Aggregate	5	5.4		
3	Virgin Asphalt by Weight of Mix	3.	62	-	
4	Virgin Asphalt by Weight of Aggregate		83		
5	Percent Air Voids (Pa)		.0	4.0	
6	Voids in Mineral Aggregate (VMA)		4.7	14 min	
7	Compacted Unit Weight Gmb, pdf	2.485	154.7	-	
8	Theoretical Maximum Density Gmm, pdf	2.589	161.2	-	
9	Percent Effective Asphalt Content (Pbe)	4.	44		
10	Percent Absorbed Asphalt (Pba)	0.	69		
11	Specific Gravity of Binder (Gb)	1.0	030		
12	Percent Gmm (2) N Initiai (7 Gyrations)	86	3.4	≤ 89.0	
13	Percent Gmm @ N Design (75 Gyrations)	96	3.0	96.0	
L4	Percent Gmm @ N Max (115 Gyrations)	97	97.6		
15	Dust to Asphalt Ratio (DP)	1	1.3		
16	Percent Passing #200 Sieve		.8	2.0-10.0	
17	Voids Filled w/ Asphalt (VFA)		3	65-75	
18	Laboratory Mixing Temperature for Design (*F)		28	321-329	
19	Laboratory Compaction Temperature for Design (*F)	30	305		
20	Laboratory Sample Weight for Volumetric Testing (g)	49	50		
21	Ignition Oven (NCAT) Correction Factor @ 482 °F	1.	05	-	
2	*Los Angeles Abrasion (LAR) (%)	2	20		
23	*Idaho Degredation ∆ % -200	2	2.6		
24	Sand Equivalent	6	68		
25	*Fracture Face Count (%)	99	99/93		
6	Fine Aggregate Angularity (%)	48	48.0		
27	*Flat and Elongated Particles in Coarse Aggregates (%)	3	.1	10 Max	
	Recycled Asphalt Pavement (RA	P) Properti	es		
8	Percentage of Asphalt in RAP (Wt. of Mix)	4.	4.34		
9	Percentage of RAP by Total Weight of Aggregate	3	34		
0	Percent of RAP Binder by Weight of Total Binder	2	29		
81	RAP Contribution by Mix	1.	48		
12	RAP Contribution by Aggregate	1.	54	-	
3	RAP NCAT Correction Factor	1.	1.73		

#### Design Specifications: Blend 4 / 75 Gyrations @ N Design PG 70-28 (64-34 Adjusted Binder)

\*Composite blend including RAP

Figure A.3: RAP No. 3

## Table A.1: RAP No. 4

True RAP Grade:	PG 82-16
Available Asphalt Content (RAP):	5.3%

Sieve Size	% Passing
1"	100
3/1"	100
1/2"	94
3/8"	83
No. 4	57
No. 8	42
No. 16	33
No. 30	25
No. 50	17
No. 100	11
No. 200	7.2

# **Appendix B. Rejuvenator Dose using Blending Charts**

Testing	Property (units)	RAP	R6 (5%)	R6 (8%)	R7 (8%)	R7 (12%)
Org. High	G*/sinδ (1 kPa)	85.3	81.9	74.2	78.7	70.7
RTFO High	G*/sinδ (2.2 kPa)	89.0	84.1	81.2	81.2	75.6
PAV BBR	BBR, Stiffness (300 MPa)	-24.5	-31.2	-34.7	-36.2	-40.7
PAV BBR	BBR, m-value (0.3)	-16.2	-24.5	-31.7	-17.3	-20.0
True Grade		PG 85.3- 16.2	PG 81.9- 24.5	PG 74.2- 31.7	PG 78.7- 17.3	PG 70.7- 20
PG Grade (M320)		PG 82-16	PG 76-22	PG 70-28	PG 76-16	PG 70-16

Table B.1: Binder Blend Critical Temperatures (°C)

## Table B.2: Calculated Dose

DSR/BBR property	PG 70-28 value	RA type	RA dose (%)
Stiffness (BBR)	245	R6	3.6
Stiffness (BBR)	245	R7	-1.3
m-value (BBR)	0.32	R6	8.3
m-value (BBR)	0.32	R7	31.6
G*/sin(δ)	1.36	R6	8.3
G*/sin(δ)	1.36	R7	11.3
Log [G*/sin(δ)]	0.13354	R6	8.5
Log [G*/sin(δ)]	0.13354	R7	11.0
m-value (#2)	0.3 (standard minimum)	R6	6.6
m-value (#2)	0.3 (standard minimum)	R7	23.6

# **Appendix C. LMLC Mix Design Summary**

Design Specifications: Blend 1 / 75 Gyrations @ N Design	PG 64-28 (58-34 Adjusted Binder)
--	----------------------------------

	Gyratory Model # Compactor: Serial #	AFG2AS 8436	Job Mix F	ormula	Spec
1	Percen	t Asphalt by Weight of Total Mix	5.3	3	
2	Percent	Asphalt by Weight of Aggregate	5.6	4	
3		Virgin Asphalt by Weight of Mix	4.0	8	
4	Virgin	Asphalt by Weight of Aggregate	4.3	2	
5		Percent Air Voids (Pa)	4.0	)	4.0
6	Vo	olds in Mineral Aggregate (VMA)	15.	6	14 min
7	C	ompacted Unit Weight Gmb, pcf	2.366	147.3	-
8	Theoreti	cal Maximum Density Gmm, pcf	2.465	153.4	
9	Percent	Effective Asphalt Content (Pbe)	5.0	4	
10	1	Percent Absorbed Asphalt (Pba)	0.3	2	
11		Specific Gravity of Binder (Gb)	1.02	28	
12	Percen	t Gmm @ N Initial (7 Gyrations)	86.	8	≤ 89.0
13	Percent G	imm @ N Design (75 Gyrations)	96.	0	96.0
14	Percent	Gmm @ N Max (115 Gyrations)	97.3	3	≤ 98.0
15		Dust to Asphalt Ratio (DP)	1.1		0.6-1.4
16		Percent Passing #200 Sieve	5.6		2.0-10.0
17		Voids Filled w/ Asphalt (VFA)	74		65-75
18	Laboratory Mixi	ing Temperature for Design (*F)	320	)	316-324
19	Laboratory Compacti	ion Temperature for Design (*F)	299	)	295-303
20	Laboratory Sample W	/eight for Volumetric Testing (g)	472	0	
21	Ignition Oven (NC/	AT) Correction Factor @ 538 °F	0.30	0	
22	*L	os Angeles Abrasion (LAR) (%)	24		30 max
23		*Idaho Degradation ∆ % -200	3.5		5.0 max
24		Sand Equivalent	66		40 min
25		*Fracture Face Count (%)	98/9	6	75/60
26		Fine Aggregate Angularity (%)	47.2	2	40 min
27	*Flat and Elongated Part	ticles in Coarse Aggregates (%)	0.3		10 max
	R	ecycled Asphalt Pavement (RAP	) Properties	\$	
28	Percentage	e of Asphalt in RAP (Wt, of Mix)	4.20	)	
29	Percentage of RA	P by Total Weight of Aggregate	30		
30	Percent of RAP B	inder by Weight of Total Binder	24		30 max
31		RAP Contribution by Mix	1.20	3	-
32		RAP Contribution by Aggregate	1.32	2	
33		RAP NCAT Correction Factor	0.36	3	

\*Composite blend including RAP

## Figure C.1: Mix Design of LMLC with RAP No. 1

Asphalt Mix Type: 12.50 m	m SP3 (Desi	gn ESALs 1	1<10)		Mix	Producer			KRC	
JMF Identification:	2103	0			Spe	cified Oil		PG	58-28	
Binder Supplier:	Idaho As	phalt			Ad	iusted Oil			NR	
Anti-Strip:	Morelife	5000		Ac	g Source	Number		N	P169c	
	Job	<b>Wix Formula</b>	a Targets	and Volu	metric Da	rta	10.000			
Target Oil Content: 5.8	R	AP Percent E	By Binder:	1	7.5		JMF G mb	Specimer	Weight	4650
in Oil Added by Mix: 4.8		RAP Oil Co	ntribution:	1	.02		JMF Nca	t Correctio	n Factor	0.73
n Oil Added by Agg: 5.1	Com	paction Tem	p. Range:	272	2-281		Lab Con	npaction T	emp (F*)	275
Percent Anti-Strip: 0.05		Mixing Tem	p. Range:	293	-305		RAI	P Correctio	n Factor	0.53
	Aggr	egate, RAP	and Bind	er Specil	ic Graviti	es				
Combined G ab : 2.591	Effective \$	Specific Grav	vity (G):	2.657			RAP BL	ulk Specific	Gravity:	2.685
Fine Aggregate G st 2.559	Fine	SSD Specific	Gravity :	2.615	-	Fine App	arent Spe	cific Gravi	ty (G _):	2.711
Coarse Coarse G at: 2.581	Coarse	SSD Specific	Gravity :	2.629	Co	arse App	arent Spe	cific Gravi	ty (G):	2.710
Agg.% Absorbtion: 2.223	Coan	se Agg. % A	bsorbtion:	1.80		E	Binder Sp	ecific Grav	rity (G .):	1.0317
	Job I	<b>Mix Formula</b>	Target V	olumetric	Properti	es		-	3-1 -1 -1 -1	
		Design	Spec.			to be and			Design	Spec.
Maximum Theoretical Specific Gr	ravity (G):	2.433	-			Bulk Spe	cific Grav	ity (G _):	2.335	-
Percent Ai	r Voids (Pa):	4.0	2.5-5.0		Voids	in Minera	Aggrega	ate (Vma):	15.1	14.0 min
Voids Filled with A	sphalt (VFA)	74	65-75			Du	st Propo	tion (DP):	1.2	0.6-1.4
Absorbed A	pshalt (P ba):	1.0	-		Effe	ective Asp	halt Cont	tent (P pe):	4.9	-
Density Percent (	Gmm @ Nini:	86.9	≤ 89.0	Uncomp	acted Voi	d Content	of Fine A	Aggregate:	51.0	40.0
Density Percent Gi	mm @ Ndes:	96.0	96			Sa	nd Equiva	alent (SE):	57	40
Density Percent Gr	nm @ Nmax:	97.5	≤ 98.0			Flat and E	Songated	Particles:	0	10
racture Face Coarse Aggregate (1)	Face/2Face):	100/100	75/60	F	aut Depth	15,000	Passes	(Average):	NR	≤ 10.0 mm
					IDEAL-C	T Crackir	ng Index (	(Average):	NR	80
	JMF 1	arget Aggre	egate, Bre	akdown,	Proportio	vis				
lend Percentages: B-P	ile 40.0	C-Pile	40.0	RAP	20.0					
	Job Mix Form	nula Aggreg	gate Grada	tions & S	Stockpile	Averages	5	1.012	Sec.	
1* 3/4	1/2"	3/8"	#4	#8	#16	#30	#50	#100	#200	
JMF Specifications: - 10	90-100	90 max	-	28-58	-	-	-	-	2.0-10.0	6
NF Target Gradation: 100 10	0 90	79	55	38	25	18	13	9	6.1	
Ncat C.F. Gradation: 100 10	0 92	79	55	38	25	18	13	10	6.5	
g. Correction Factor: 0.0 0.	0 2.2	-0.1	-0.1	-0.3	-0.2	-0.2	0.0	0.6	0.4	
B-Pile 100 10	0 79	54	12	2	1	1	1	1	0.6	
C-Pile 100 10	0 100	100	93	70	46	32	22	15	10.7	
RAP 100 10	0 94	86	64	46	33	24	17	12	8.0	
0										

Figure C.2: Mix Design of LMLC with RAP No. 2

Desig	n Specifications: Blend 4 / 75 Gyrations @ N Design	n PG 70-28 (64-34	Adjusted Binder)
	Gyratory Model # AFG2AS Compactor: Serial # 8732	Job Mix Formula	Spec
1	Percent Asphalt by Weight of Total Mix	5.1	-
2	Percent Asphalt by Weight of Aggregate	5.4	
3	Virgin Asphalt by Weight of Mix	3.62	-
4	Virgin Asphalt by Weight of Aggregate	3.83	
5	Percent Air Voids (Pa)	4.0	4.0
6	Voids in Mineral Aggregate (VMA)	14.7	14 min
7	Compacted Unit Weight Gmb, pcf	2.485 154.7	
8	Theoretical Maximum Density Gmm, pcf	2.589 161.2	-
9	Percent Effective Asphalt Content (Pbe)	4.44	
10	Percent Absorbed Asphalt (Pba)	0.69	
11	Specific Gravity of Binder (Gb)	1.030	
12	Percent Gmm @ N Initial (7 Gyrations)	86.4	≤ 89.0
13	Percent Gmm @ N Design (75 Gyrations)	96.0	96.0
14	Percent Gmm @ N Max (115 Gyrations)	97.6	≤ 98.0
15	Dust to Asphalt Ratio (DP)	1.3	0.6-1.5
16	Percent Passing #200 Sieve	5.8	2.0-10.0
17	Voids Filled w/ Asphalt (VFA)	73	65-75
18	Laboratory Mixing Temperature for Design (°F)	328	321-329
19	Laboratory Compaction Temperature for Design (°F)	305	299-308
20	Laboratory Sample Weight for Volumetric Testing (g)	4950	
21	Ignition Oven (NCAT) Correction Factor @ 482 °F	1.05	
22	*Los Angeles Abrasion (LAR) (%)	20	30 max
23	*ldaho Degredation ∆ % -200	2.6	5.0 max
24	Sand Equivalent	68	40 min
25	*Fracture Face Count (%)	99/93	75/60
26	Fine Aggregate Angularity (%)	48.0	40 min
27	*Flat and Elongated Particles in Coarse Aggregates (%)	3.1	10 Max
	Recycled Asphalt Pavement (R/	P) Properties	
28	Percentage of Asphalt in RAP (Wt. of Mix)	4.34	
29	Percentage of RAP by Total Weight of Aggregate	34	
30	Percent of RAP Binder by Weight of Total Binder	29	30 Max
31	RAP Contribution by Mix	1.48	
32	RAP Contribution by Aggregate	1.54	-
the second se			1

\*Composite blend including RAP

Figure C.3: Mix Design of LMLC with RAP No. 3

Sieve Size	1 inch	³⁄₄ inch	½ inch	<sup>3</sup> / <sub>8</sub> inch	No.4	No.8	No.16	No.30	No.50	No.100	No.200
Virgin % Passing	100	100	94	76	50	37	24	18	14	10	5.3
RAP4 % Passing	100	100	94	83	57	42	33	25	17	11	7.2
25% RAP % Passing	100	100	94	76	50	37	24	18	14	10	5.3
50% RAP % Passing	100	100	94	76	50	37	24	18	14	10	5.3
70% RAP % Passing	100	100	94	76	50	37	24	18	14	10	5.3
Control Points		100	90			28					2.0
Control Points		100	100	90		58					10.0
Restricted Zone						39.1	25.6	19.1	15.5		
Restricted Zone						39.1	31.6	23.1	15.5		

Table C.1: Mix Design of LMLC with RAP No. 4

Mix Property	Job Mix Formula	Spec.
Percent Asphalt by Weight of Total Mix	5.3	-
Percent Asphalt by Weight of Aggregate	5.3	-
Percent Air Voids (Pa)	4.1	4.0
Voids in Mineral Aggregate (VMA)	14.7	14 min
Compacted Unit Weight Gmb, pcf	2.307	-
Theoretical Maximum Density Gmm, pcf	2.406	-
Percent Effective Asphalt Content (Pbe)	4.7	-
Percent Absorbed Asphalt (Pba)	0.54	-
Specific Gravity of Binder (Gb)	1.0331	-
Percent Gmm @ N Initial (7 gyrations)	86.7	<u>&lt;</u> 89.0
Percent Gmm @ N Design (75 gyrations)	95.9	96.0
Percent Gmm @ N Maximum (115 gyrations)	96.8	<u>&lt;</u> 98.0
Dust to Asphalt Ratio (DP)	1.1	0.8-1.2
Percent Passing #200 Sieve	5.0	3.0-6.0
Voids Filled with Asphalt (VFA)	72.1	-
Laboratory Mixing Temperature for Design (F)	336	327-337
Laboratory Compaction Temperature for Design (F)	314	306-316
Laboratory Sample Weight for Volumetric Testing (g)	4741	-
Ignition Oven (NCAT) Correction Factor @ 538 F	0.37	-
Los Angeles Abrasion (LAR) (%)	29	40 max
Sand Equivalent	64	45 min
Fracture Face Count (%)	98	75 min (2 Face)
Fine Aggregate Angularity (%)	46.1	45.0 min
Flat and Elongated Particles in Coarse Aggregates (%)	0	20 max (3:1)
Coarse Clay Lumps and Friable Particles	0	0.3 max
Fine Clay Lumps and Friable Particles	0	0.3 max
Percent Natural Sand	0	15 max
Coarse Sodium Sulfate Soundness	1.1	12 max

Figure C.4: Mix Design of LMLC with RAP No. 4

# Appendix D. PMLC Mix Design Summary

										283		
					Project	Number.	A020(	590)	Pr	oject Key I	Vumber:	20590
Asphalt Mix Type:	19 mm	SP3 (De	sign ES/	ALs, 1<1	0)		Mix	Producer:				
JMF Identification:	20036-	19mmSP	3-R30				Sp	ecified Oil:	PG 58	-34		
Binder Supplier:	Idaho /	Asphalt S	Supply				A	justed Oil:	PG 58	-34		
Anti-Strip:	Morlife	5000				Ag	g. Sourc	e Number:	Bu-26	s		
			Job Mix	Formula	Targets	and Vo	lumetri	c Data				
Target Oil Content:	5.0		RAPP	ercent B	Binder:	30		JA	AF G mb	Specimen	Weight:	4675
Virgin Oil Added by Mix:	3.5		RAP	Oil Con	tribution:	1.5		JI	MF Ncat	Correction	Factor:	0.40
Virgin Oil Added by Agg:	3.7	C	ompacti	on Temp	Range:	288-30	2 °F	L	ab Com	paction Te	mp (F°):	292
Percent Anti-Strip:	0.75		Mixi	ng Temp	Range:	315-32	9 °F	Bir	nder Spe	ecific Gravi	ity (G .):	1.0293
and the second states and	1.455	,	Aggrega	te, RAP	and Bin	der Spec	ific Gra	vities	12.40			- to the
Combined Gsb:	2.585	Effecti	ve Speci	fic Gravi	y (G):	2.624		D6	RAP Bu	Ik Specific	Gravity:	2.589
Fine Aggregate Gsb:	2.562	F	ine SSD	Specific	Gravity :	2.606		IF	RAP Bu	lk Specific	Gravity:	2.605
Coarse Aggregate Gsb:	2.594	Coar	se SSD	Specific	Gravity :	2.621	1	Fine Appar	ent Spe	cific Gravit	y (G):	2.679
Fine Agg. % Absorption	1.6	C	oarse Ac	19. % Ab.	sorbtion:	1.1	Coar	se Appare	nt Spec	ific Gravity	(G):	2.667
			FRAPO	Correction	Factor	0.08			D6 RAP	<sup>2</sup> Correctio	n Factor	0.07
Market States			lob Mix	Formula	Target	Volumet	ric Prop	erties	00101	00110000	in actor	0.01
				Design	Spec		1				Design	Spec
Maximum Theoretica	Specif	fic Gravity	(G):	2.435				Bulk Speci	fic Grav	ity (G):	2.338	
	Perce	nt Air Voi	ds (P .):	4.0	2.5-5.0		Voids	n Mineral /	Aaareaa	te (VMA):	14.1	13.0 mir
Voids	Filled w	ith Asoha	# (VFA)	72	65-75			Dus	t Propo	tion (DP)	1.2	0.6-1.4
10100	Absorb	ed Apsha	ut (P):	0.59			Effe	ctive Asph	alt Cont	ent (P):	4.44	
Dens	ity Per	ent G	@ Nini:	97.6	≤ 89.0		De	nsity Perce	ont G	@ Ndes:	96	96.0
Densit	Perce	nt G @	D Nmax	97.3	≤98.0	-	verage	Rut Depth	@15.00	0 Passes	5.2	10.0 ma
Uncompacted Void C	ontent o	f Fine An	arenate:	51	40 min		10.090	in Doput	Sand F	auivalent	60	40 min
racture Face Coarse Age	menate	/1 Eaco/	2 Eacel	98/98	75/60			lat and Ek	ongeted	Particlae	0	10 may
racture race coarse Ag	heyate	[//ace/	21 409).	30/30	1000		Autorom	DEAL C	T Crock	ina ladav	123	80 min
Contraction of the local			WE Tarro	et Agare	cate Br	eakdow	Prop	ortions	1 Crack	ing moex.	123	00 11111
Bland Percentages:		A-Dilo:	12.0	B. Dilo:	24.0	C. Dilo:	37.0	IE DAD	20.0	DE DAD	10.0	
Diena Percentages.		Job Mix	Formula	Agarea	ate Grac	lations /	Stock	nile Avera	20.0	DO IVAF.	10.0	
	4.	2/4"	1/07	2/0"	#4	#0	#16	#20	450	#100	#200	
IME Societation	1	3/4	00 100	3/0	#4	#0	#10	#30	#50	#100	#200	
IME Torget Credetien	400	100	90-100	90 max	40	28-58		47	40		2.0-10.0	<u> </u>
Next C.E. Credetion:	100 0	99	00	73	48	32	23	1/	12	8.0	5.2	
And Competing Factor	100.0	98.3	84.4	12.1	49.2	30.7	23.2	16.9	11.9	1.2	4.2	
Agg. Correction Factor:	0.0	0.7	0.6	0.3	-1.2	1.3	-0.2	0.1	0.1	0.8	1.0	
A-P110	100	91	21	5	2	1	1			1	0.6	
B-Pile	100	100	80	45	5	2	2	1	1	1	0.7	
Breakdown:	100	100	100	100	100	100	100	100	100	100	90.0	
C-Pile:	100	100	100	100	80	53	35	24	16	10	7.3	
IF Category 1 RAP:	100	100	96	88	62	43	33	28	22	14	7.6	
	400	100	90	78	51	36	28	22	47	42	70	

Figure D.1: Job Mix Formula for Project No. 1

Key Number Pro	oject Numbe	r		_	P	roject Na	ame				_		District
19863		A019	(863)				SH	1-75, (	Old US-93 to	Richfield			4
Identification Num	ber (Program/	Task/Phase/Sa	mple#)		C	ontract I	tem Number	Testin	g Laboratory N	ame & Loca	tion	Mix Des	sign No.
Supe	erpave HM	A 3/8" SP3	/P-17488	0		405	-435A		ITD Central	Laborator	y I	A52	20-0221
Doi	ug Yearsle	ineers Name V	s) Sampled	D. Kil	mer		22046	mber IL	6/10/2020	6/10/2	020	Date La 6/1	1/2020
Asphalt Binder Sup	oplier	Asphalt B	inder Grade	ISp. Gr.	of Bind	er (from	Mix Design)	JMF I	ntended Binder	% (by Wt o	(Mix)	Source	Number
Idaho Aspha	alt Supply	PG	70-28			1.031	• •		6	.2		L	n-84s
Sample Location (	Sta./offset, tr Lab Prep	uck, plant, la bared	o, etc.)	Mix De A	sign La	b & Loca st, Meri	tion S dian	PMDT Da	an Kilmer	P.E. in Re Ad	esponsibl rian Ma	e Charg scorro	e of Desig , P.E.
SALs 1 <10 (75 G	vrations)	Nom. Max. S	Size Aggrega	le	Prin	nary Con N	trol Sieve	Per	cent Passing P 47	rimary Contr	ol Sieve	Class	s of Mix SP3
Combined Agg	regate Bul	k SPG G.,	from ITD	0802	_	2.5	548	_		70		_	010
55	5				т	est Res	sults						
	Grad	dation Anayi	sis				1	Asphall	Binder Conte	nt (By Weig	ht of Mix	()	
	FOP f	or AASHTO	Т 30			I			FOP for AA	SHTO T 308		20	
	Lab No.	Lab No.	Lab No.						Lab No.	Lab No.	Lab N	o.	
	209MX-								209MX-				
Sieve Size	-0088			8. 7		I			-0088				Average
(mm) (in.)				Avg.	JMF	Total As	phalt Binder	Conter	nt 6.59				1
(50) 2	100			100	100	NCAT (	Correction Fa	ictor	0.42				
(37.5) 1/1/2	100			100	100	Moistu	re % (-)		0.03				+
(25) 1	100			100	100	Act. As	ph. Binder Co	ontent ?	6.14				6.1
(19) 3/4	100			100	100	1							
12.5) 1/2	100			100	100	Compa	action Tempe	erature,	۴F				
(9.5) 3/8	95			95	94	1	304		FOP for AASH	TO T 312/	ASHTO	M 323	Average
4.75) No. 4	70			70	70	Lab Air	Voids % at i	Design	3.6				3.6
(2.36) No. 8	46			46	46	G <sub>mb</sub> (co	mpacted mix	dure)	2.307				2.307
(1.18) No. 16	31			31	32	G <sub>mm</sub> (m	ax spec grav	rity)	2.392				2.392
0.600) No. 30	23			23	23	VMA, %	•		15.0				15.0
0.300) No. 50	10	-		17	11	VFA, %			/6		-		76
0.150) No. 100	7.2			72	64	Dust Pr	oportion (DP	2	1.4				1.4
1.075) NO. 204	1.4			1.2	0.4	0 5	factive Re. C	and the second	2 610		1		
hun Comple						D Ef	E Binder Con	stant %	5.12		-	-	
Height mm	115.4		I	1		P. Bi	nder Absorbs	ad %	1.00				
Longine, mini	110.4			1		ta Di		50, 70	1.05				
FOP for AASH FOP for AASH	TO T 209 res TO T 166 res	sult within 0.0 sult within 0.0	20 of JMF? 20 of JMF?	Gmm fi Gmb fi	rom JM rom JM	F= 2.3 F= 2.2	389 Gmm 1 293 Gmb 1	from Sa from Sa	emple Tested= emple Tested=	2.392 2.307	Yes Yes		e.
Sample #	ASTM D10	075 & AASH1	O T 167				Sample #	n	AASHTO T 3	40			
<u>84</u> %@_0	.50 % FAIL	MO	RLIFE 5000			Rutting Le	Depth, mm ft Sample enter Sample		n/a n/a n/a	Max 0.2	imum Alk Rut Dep in. (5	owable th mm)	
		1	Mix Desia	n Volu	metri	cs Con	firmation:			- Eail	<u> 11/a</u>		
Remarks									11 435	ran			
astad By	A	PA becam	e inoperab	le prior	to tes	t. Retai	ning pucks	s for in	formation o	nly testing.			
esieu by	Dan Hens	scheid/Jain	ne Conley/	Shelby	Alvar	ado/ H	eather Mile	ЭУ		24083	mber /24080/	24079	/24047
Alteron Maneu	020			Labora	NORY MA	anagers	Signature	Ch	ad Clawson	DE			
0/10/2	020		to ad the	-	Dig	itally signe	d by Chad W.	Ulla	au Clawsoll	1.0.6.		_	

Figure D.2: Job Mix Formula for Project No. 2

	Class:		12.5mm SP	-3 PG 70-28	<b>i</b>
	Project:		US-93, 200	South	
Proposed	Job Mix F	ormula	I		
Laboratory Values			Target		Spec.
Total Asphalt by Weight of Mix % (Pb)			5.6		
Total Aspalt by Weight of Aggregate			5.98		
Air Voids % (Va)			4.0		3.0-5.0
Voids in Mineral Aggregate (VMA)			14.7		14.3
Voids Filled with Asphalt (VFA)			73		65-75
Bulk Specific Gravity (Gmb)			2.304		
Unit Weight lb./cuft.			143.4		
Theo Max Spec Gravity (Gmm)			2.401		
Theo Max Spec Gravity lb./cuft.			149.5		
Effective Specific Gravity of Blend (Gse)			2.608		
Effect of Water on Compressive Strength (AllWes	1)		95		85 min
Ninitial (7 Gyrations)			86.4		≤ 89.0
Ndesign SP-5 (75 Gyrations)			96.0		= 96.0
Nmax (115 Gyrations)			97.6		≤ 98.0
NCAT Asphalt Correction Factor			0.23		
Dust to Asphalt			1.1		
Laboratory Mixing Temperature( deg in F)			320		
Laboratory Compaction Temperature(deg in F)			300		
Plant Mixing Temperature(deg in F)**		316	-	324	
Field Compaction Temperature(deg in F)**		295	-	303	
Superpave Design Sample Wt. in grams			4580		

\*Field mixing and compaction may be adjusted +/- 25 degrees per Viscosity Graph

### **Aggregate Gradation Data**

Sieve Size	Ln-80c B 23.0%	Cs-201 C 16.0%	Ln-80c WC 25.5%	Md-101c Sand 5.0%	RAP 30.0%	Break down 0.5%	JMF Blended Gradation
1" / 25mm	100	100	100	100	100	100	100
3/4" / 19mm	100	100	100	100	100	100	100
1/2" / 12.5mm	88	100	100	100	95	100	96
3/8" / 9.5mm	46	100	100	100	86	100	83
No. 4 / 4.75mm	2	85	75	100	62	100	57
No. 8 / 2.36mm	2	58	43	85	45	100	39
No.16 / 1.18mm	1	41	26	66	34	100	27
No. 30 / 600um	1	30	16	54	26	100	20
No. 50 / 300um	1	22	9	20	20	100	14
No. 100 / 150um	1	15	4	4	14	100	9
No. 200 / 75um	0.7	9.9	1.4	1.4	9.5	90.0	5.5

Figure D.3: Job Mix Formula for Project No. 3

## ASPHALT CONCRETE JOB MIX FORMULA

Designate	Rie West River Reides Restaurout	
Project:	Big wood River Bridge Replacement	
Paving Contractor:	Valley Paving	
Asphalt Supplier:	Idaho Asphalt	Spec
Anti-Strip Agent:	0.5% Morelife 5000	

Aggregate Sources: BE 106c

Design Specification: ITD

Date: 7/18/2016 Class of Mixture: SP-3 cified Grade of Asphalt: PG 64-34 Prepared by: LLC Gyratory Compactor: Pine AFGC125X

Page 5

	AASHTO	JMF	Requirements
1. Asphalt by Weight of Total Mix, %	R 35	5.8%	
<ol><li>Asphalt by Weight of Aggregates, %</li></ol>		6.1%	
3. Asphalt by Weight of Total Mix (Added), %		5.0%	
<ol><li>Asphalt by Weight of Aggregates (Added), %</li></ol>		5.2%	
5. Air Voids (Va), %	T 269	4.0%	3.0-5.0
<ol><li>Voids in Mineral Aggregate (VMA), %</li></ol>	R 35	15.0%	13.3 min
7. Bulk Specific Gravity @ Ndes (Gmb)	T 166	2.296	142.9 pcf
8. Theoretical Maximum Specific Gravity (Gmm)	T 209	2.391	148.8 pcf
9. Relative Density %Gmm @ Nini (7 Gyrations)	R 35	87.4	≤ 89.0
10. Relative Density %Gmm @ Nmax (115 Gyrations)	R 35	97.2	≤ 98.0
11. Voids Filled w/ Asphalt (VFA), %	R 35	73.3%	65-75
12. Film thickness, microns		1	
13. Absorbed Asphalt (Pba) by Weight of Aggregate, %	R 35	0.90%	
14. Effective Asphalt Content (Pbe) by Total Wt of Mixture, %	R 35	4.9%	
15. Specific Gravity of Asphalt		1.026	
16. Laboratory Mixing Temp, °C/°F		318-333	
17. Laboratory Compaction Temp, °C/°F		288-304	
18. Recommended Plant Mixing Temp, °F		325	
19. Compaction Temp Range, °F		296	
20. NCAT Ignition Oven Correlation Factor @ 538° C	T 308	0.17	
21. Dust to Asphalt Ratio	R 35	1.2	0.6-1.2
22. Immersion Compression Retained Strength, %	T 165	85%	85% min
23. Gyratory Gmb specimen weight, grams		4500	
24. Combined Bulk Dry Specific Gravity of Aggregate (Gsb)	T 85 / IT 144	2.545	

#### AGGREGATE STOCKPILE GRADATION

	A-Pile 15%	B-Pile 18%	C-Pile 50%	RAP 17%	0%	Blended Gradation	Mix Design Tolerances
25.0 mm (1")	100%	100%	100%	100%	0%	100%	100
19.0 mm (3/4")	99%	100%	100%	100%	0%	100%	100
12.5 mm (1/2")	12%	95%	100%	96%	0%	85%	79-91
9.5 mm (3/8")	2%	61%	100%	88%	0%	76%	70-82
4.75 mm (No.4)	1%	2%	87%	66%	0%	55%	49-61
2.36 mm (No.8)	1%	1%	55%	46%	0%	36%	31-41
1.18 mm (No.16)	1%	1%	36%	34%	0%	24%	19-29
600 um (No.30)	1%	1%	26%	25%	0%	18%	13-23
300 um (No.50)	1%	1%	18%	19%	0%	13%	9-17
150 um (No.100)	1%	1%	13%	14%	0%	9%	5-13
75 um (No.200)	0.2%	0.1%	8.3%	9.5%	0.0%	5.8%	3.8-7.8

#### Figure D.4: Job Mix Formula for Project No. 4

				Class: Project:		12.5mm SI SH-27, Pol	P-2 PG 58-28 eline INT In	3 aprovement
		Pro	posed J	ob Mix I	Formula	I		
Laboratory Valu	ies					Target		Spec.
Total Asphalt by We	ight of Mi	x % (Pb)				5.7		
Total Aspalt by Weig	t of Aggi	regate				6.01		
Air Voids % (Va)						4.0		3.0 - 5.0
Voids in Mineral Ag	gregate (V	MA)				14.9		14.3
Voids Filled with Asp	ohalt (VFA	)				73		65 - 78
Bulk Specific Gravity	y (Gmb)					2.304		
Unit Weight lb./cuft.						143.4		
Theo Max Spec Grav	ity (Gmm	)				2.401		
Theo Max Spec Grav	ity lb./cuf	t.				149.5		
Effective Specific Gra	avity of Bl	end (Gse)				2.610		
IDEAL-CT Index						93		≥ 80
Rut Depth @ 15,000	Passes					1.8mm		≤ 10.0 mm
Stripping Passes						NA		15,000
Ninitial (6 Gyrations	•)					86.3		≤ 60.5
Ndesign SP-5 ( 50 Gy	rations)					96.0		= 96.0
Nmax (75 Gyrations	)					97.6		≤ 98.0
NCAT Asphalt Corre	ection Fact	tor				0.05		
Dust to Asphalt						1.2		0.6 - 1.4
Laboratory Mixing T	<b>Femperatu</b>	re( deg in l	F)			300		
Laboratory Compact	tion Temp	erature(de	g in F)			280		
Plant Mixing Tempe	rature(deg	in F)**			316	-	303	
Field Compaction Te	mperature	e(deg in F)	**		275	-	283	
Superpave Design Sa	mple Wt.	in grams				4580		
*Field mixing and comp	action may l	be adjusted Agg	+/- 25 degre gregate	es per Visco <b>Gradati</b>	sity Graph on Data			
Sieve Size		Ln-80c B	Cs-201	Ln-80c WC	Md-101c Sand	RAP	Break	JMF

		Aggregate	breakdown	will be	controlled	by the	Hot	Plant dust	control s	vstem.
--	--	-----------	-----------	---------	------------	--------	-----	------------	-----------	--------

34.0%

0.7

1"/25mm

3/4" / 19mm

1/2" / 12.5mm

3/8" / 9.5mm No. 4 / 4.75mm

No. 8 / 2.36mm

No.16 / 1.18mm No. 30 / 600um

No. 50 / 300um

No. 100 / 150um

No. 200 / 75um

13.0%

10.2

12.0%

1.4

5.0%

1.4

35.0%

9.5

1.0%

90.0

Gradation

6.0

Figure D.5: Job Mix Formula for Project No. !	e D.5: Job Mix Formula fo	r Proiect	No. 5
---	---------------------------	-----------	-------

#### ASPHALT CONCRETE JOB MIX FORMULA

Page 5

Project:	Big Wood River Bridge Replacement	Date:	7/18/2016
Paving Contractor:		Class of Mixture:	SP-3
Asphalt Supplier:	Idaho Asphalt	Specified Grade of Asphalt:	PG 64-34
Anti-Strip Agent:	0.5% Morelife 5000	Prepared by:	LLC
Aggregate Sources:	BE 106c	Gyratory Compactor:	Pine AFGC125X
Design Specification:	ITD		

	AASHTO	JMF	Requirements
1. Asphalt by Weight of Total Mix, %	R 35	5.8%	
<ol><li>Asphalt by Weight of Aggregates, %</li></ol>		6.1%	
3. Asphalt by Weight of Total Mix (Added), %		5.0%	
<ol><li>Asphalt by Weight of Aggregates (Added), %</li></ol>		5.2%	
5. Air Voids (Va), %	T 269	4.0%	3.0-5.0
<ol><li>Voids in Mineral Aggregate (VMA), %</li></ol>	R 35	15.0%	13.3 min
<ol><li>Bulk Specific Gravity @ Ndes (Gmb)</li></ol>	T 166	2.296	142.9 pcf
8. Theoretical Maximum Specific Gravity (Gmm)	T 209	2.391	148.8 pcf
9. Relative Density %Gmm @ Ninl (7 Gyrations)	R 35	87.4	≤ 89.0
10. Relative Density %Gmm @ Nmax (115 Gyrations)	R 35	97.2	≤ 98.0
11. Voids Filled w/ Asphalt (VFA), %	R 35	73.3%	65-75
12. Film thickness, microns		1	
13. Absorbed Asphalt (Pba) by Weight of Aggregate, %	R 35	0.90%	
14. Effective Asphalt Content (Pbe) by Total Wt of Mixture, %	R 35	4.9%	
15. Specific Gravity of Asphalt		1.026	
16. Laboratory Mixing Temp, °C/°F		318-333	
17. Laboratory Compaction Temp, °C/°F		288-304	
18. Recommended Plant Mixing Temp, °F		325	
19. Compaction Temp Range, °F		296	
20. NCAT Ignition Oven Correlation Factor @ 538° C	T 308	0.17	
21. Dust to Asphalt Ratio	R 35	1.2	0.6-1.2
22. Immersion Compression Retained Strength, %	T 165	85%	85% min
23. Gyratory Gmb specimen weight, grams		4500	
24. Combined Bulk Dry Specific Gravity of Aggregate (Gsb)	T 85 / IT 144	2.545	

#### AGGREGATE STOCKPILE GRADATION

	A-Pile 15%	B-Pile 18%	C-Pile 50%	RAP 17%	0	Blended Gradation	Mix Design Tolerances
25.0 mm (1*)	100%	100%	100%	100%	0%	100%	100
19.0 mm (3/4")	99%	100%	100%	100%	0%	100%	100
12.5 mm (1/2*)	12%	95%	100%	96%	0%	85%	79-91
9.5 mm (3/8*)	2%	61%	100%	88%	0%	76%	70-82
4.75 mm (No.4)	1%	2%	87%	66%	0%	55%	49-61
2.36 mm (No.8)	1%	1%	55%	46%	0%	36%	31-41
1.18 mm (No.16)	1%	1%	36%	34%	0%	24%	19-29
600 um (No.30)	1%	1%	26%	25%	0%	18%	13-23
300 um (No.50)	1%	1%	18%	19%	0%	13%	9-17
150 um (No.100)	1%	1%	13%	14%	0%	9%	5-13
75 um (No.200)	0.2%	0.1%	8.3%	9.5%	0.0%	5.8%	3.8-7.8

Figure D.6: Job Mix Formula for Project No. 6

			Class:		12.5mm SF	-3 PG 70-28	\$	
	-		Project:		US-93, 200	South		
Laboratory Values	Pro	posed J	ob Mix	Formula	Target		Spee	
Laboratory values	of Mix % (Pb)				Target		spec.	
Total Aspirat by Weight o	of MIX 76 (FD)				5.98			
Air Voids % (Va)	Angeregate				4.0		3.0-5.0	
Voids in Mineral Aggreg	ate (VMA)				14.7		14.3	
Voids Filled with Asphalt	t (VFA)				73		65-75	
Bulk Specific Gravity (G	mb)				2.304		00-10	
Unit Weight lb./cuft.					143.4			
Theo Max Spec Gravity (	(Gmm)				2.401			
Theo Max Spec Gravity	b./cuft.				149.5			
Effective Specific Gravity	of Blend (Gse)				2.608			
Effect of Water on Comp	ressive Strength	(AllWest)			95		85 min	
Ninitial (7 Gyrations)	(7 Gyrations) 86.4							
Ndesign SP-5 (75 Gyrati	Gyrations ) 96.0							
Nmax (115 Gyrations)	5 Gyrations ) 97.6							
NCAT Asphalt Correctio	n Factor				0.23			
Dust to Asphalt					1.1			
Laboratory Mixing Tem	perature( deg in	F)			320			
Laboratory Compaction	Temperature(de	g in F)			300			
Plant Mixing Temperatu	re(deg in F)**			316	-	324		
Field Compaction Tempe	erature(deg in F)	**		295	-	303		
Superpave Design Sampl	e Wt. in grams				4580			
Field mixing and compactio	n may be adjusted	+/- 25 degre	es per Visco	sity Graph				
	Ag	gregate	Gradati	on Data				
01 - 01	Ln-80c	Cs-201	Ln-80c	Md-101c		Break	IMF	
Sieve Size	В	C	WC	Sand	RAP	down	Blended	
	23.0%	16.0%	25.5%	5.0%	30.0%	0.5%	Gradation	
1"/25mm	100	100	100	100	100	100	100	
3/4" / 19mm	100	100	100	100	100	100	100	
1/2" / 12 5mm	89	100	100	100	05	100	06	
2/8" / 0.5mm	00	100	100	100	95	100	90	
No. 4 / 4 75	2	96	75	100	60	100	67	
No. 8 / 2.36mm	2	65	43	85	45	100	30	
No. 16 / 1. 18mm		41	26	66	34	100	27	
No.10/1.18mm	1	30	16	54	26	100	20	
No. 30 / 000um	1	30	10	29	20	100	20	
No. 507 300um	1	22	9	20	20	100	19	

\* Aggregate breakdown will be controlled by the Hot Plant dust control system.

1

0.7

No. 100 / 150um

No. 200 / 75um

Figure D.7: Job Mix Formula for Project No. 7

4

1.4

4

1.4

14

9.5

100

90.0

9

5.5

15

9.9

Gy	ratory Compactor: Model # Serial #	AFG2AS 8436	Job M	ix Formula	Spec
1	Percen	t Asphalt by Weight of Total Mix	c	5.4	
2	Percent	Asphalt by Weight of Aggregate		5.71	
3		Percent Air Voids (Pa	)	4.0	4.0
4	Ve	oids in Mineral Aggregate (VMA	)	14.7	14 min
5	C	ompacted Unit Weight Gmb, pc	f 2.313	144.0	
6	Theoret	ical Maximum Density Gmm, pc	f 2.410	150.0	
7	Percent	Effective Asphalt Content (Pbe	)	4.74	
8	I	Percent Absorbed Asphalt (Pba	)	0.70	
9		Specific Gravity of Binder (Gb	) .	1.026	
10	Percer	nt Gmm @ N Initial (7 Gyrations	)	87.2	≤ 89.0
11	Percent G	mm @ N Design (75 Gyrations	)	96.0	96.0
12	Percent	Gmm @ N Max (115 Gyrations		97.6	≤ 98.0
13		Dust to Asphalt Ratio (DP	)	1.3	0.6-1.4
14		Percent Passing #200 Sieve		6.0	2.0-10.0
15		Voids Filled w/ Asphalt (VFA	)	73	65-75
16	Laboratory Mix	ing Temperature for Design (°F		311	307-315
17	Laboratory Compact	ion Temperature for Design (°F	)	290	286-294
18	Laboratory Sample V	Veight for Volumetric Testing (g	)	4590	
19	Ignition Oven (NC	AT) Correction Factor @ 538 °F	:	0.37	
20		Sand Equivalen	t	64	40 min
21		Fracture Face Count (%	) 9	99/97	75/60
22		Fine Aggregate Angularity (%		46.3	40 min
23	Flat and Elongated Par	ticles in Coarse Aggregates (%		0.0	10 max

Figure D.8: Job Mix Formula for Project No. 8

C	Gyratory Model # compactor: Serial #	AFG2AS 8436	Job Mix	Formula	Spec
1	Percen	t Asphalt by Weight of Total Mix	5.	5	-
2	Percent	Asphalt by Weight of Aggregate	5.7	79	-
3		Virgin Asphalt by Weight of Mix	4.3	36	-
4	Virgin	Asphalt by Weight of Aggregate	4.6	61	-
5		Percent Air Voids (Pa)	4.	0	4.0
6	Ve	oids in Mineral Aggregate (VMA)	15	.1	14 min
7	C	ompacted Unit Weight Gmb, pcf	2.397	149.2	-
8	Theoret	ical Maximum Density Gmm, pcf	2.497	155.4	-
9	Percent	Effective Asphalt Content (Pbe)	4.7	75	-
10		Percent Absorbed Asphalt (Pba)	0.7	76	
11		Specific Gravity of Binder (Gb)	1.0	27	-
12	Percer	t Gmm @ N Initial (7 Gyrations)	86	.3	≤ 89.0
13	Percent 0	mm @ N Design (75 Gyrations)	96	.0	96.0
4	Percent	Gmm @ N Max (115 Gyrations)	97	.2	≤ 98.0
15	arranda a suba an	Dust to Asphalt Ratio (DP)	1.	.3	0.6-1.5
16	ingle. Failer in the	Percent Passing #200 Sieve	6.	.3	2.0-10.0
.7		Voids Filled w/ Asphalt (VFA)	7	4	65-75
.8	Laboratory Mix	ing Temperature for Design (°F)	31	12	308-316
9	Laboratory Compact	ion Temperature for Design (°F)	29	00	286-294
20	Laboratory Sample V	Veight for Volumetric Testing (g)	47	75	-
1	Ignition Oven (NC	AT) Correction Factor @ 538 °F	0.	77	-
12		Sand Equivalent	6	9	40 min
23		Fracture Face Count (%)	100/	/100	75/60
24		Fine Aggregate Angularity (%)	48	3.4	40 min
25	Flat and Elongated Pa	ticles in Coarse Aggregates (%)	0	.9	10 Max
	F	ecycled Asphalt Pavement (RA	P) Properti	es	Service and the service of the servi
.6	Percentag	e of Asphalt in RAP (Wt. of Mix)	5.	57	-
27	Percentage of R/	P by Total Weight of Aggregate	2	0	-
28	Percent of RAP	Binder by Weight of Total Binder	2	0	30 max
29		RAP Contribution by Mix	1.	11	-
30		RAP Contribution by Aggregate	1.	18	-
31		RAP NCAT Correction Factor	1.	09	-

Figure D.9: Job Mix Formula for Project No. 9

	Gyratory Model # Compactor: Serial #	AFG2AS 8732	Job Mix	Formula	Spec
1	Percen	t Asphalt by Weight of Total Mix	5.	7	
2	Percent	Asphalt by Weight of Aggregate	6.	1	
3		Virgin Asphalt by Weight of Mix	4.3	30	
4	Virgin	Asphalt by Weight of Aggregate	4.5	i8	
5		Percent Air Voids (Pa)	4.	0	4.0
6	Ve	oids in Mineral Aggregate (VMA)	15	.0	14 min
7	C	ompacted Unit Weight Gmb, pcf	2.391	148.8	
8	Theoreti	cal Maximum Density Gmm, pcf	2.491	155.1	
9	Percent	Effective Asphalt Content (Pbe)	4.7	'3	
10		Percent Absorbed Asphalt (Pba)	1.0	6	
11		Specific Gravity of Binder (Gb)	1.0	30	
12	Percer	t Gmm @ N Initial (7 Gyrations)	86	.6	≤ 89.0
13	Percent G	mm @ N Design (75 Gyrations)	96	.0	96.0
14	Percent	Gmm @ N Max (115 Gyrations)	97	.4	≤ 98.0
15		Dust to Asphalt Ratio (DP)	1.4		0.6-1.5
16		Percent Passing #200 Sieve	6.	4	2.0-10.0
17		Voids Filled w/ Asphalt (VFA)	73	3	65-75
18	Laboratory Mix	ing Temperature for Design (°F)	32	4	316-324
19	Laboratory Compact	ion Temperature for Design (°F)	30	2	294-303
20	Laboratory Sample V	/eight for Volumetric Testing (g)	475	50	
21	Ignition Oven (NC	AT) Correction Factor @ 482 °F	1.5	1	
22	*L	os Angeles Abrasion (LAR) (%)	20	)	30 max
23		*Idaho Degredation ∆ % -200	4.	3	5.0 max
24		Sand Equivalent	6		40 min
25		*Fracture Face Count (%)	99/	98	75/60
26		Fine Aggregate Angularity (%)	47	.4	40 min
27	*Flat and Elongated Par	ticles in Coarse Aggregates (%)	4.	8	10 Max
	R	ecycled Asphalt Pavement (RAP	) Propertie	s	
28	**Percentag	e of Asphalt in RAP (Wt. of Mix)	4.7	5	
29	Percentage of RA	P by Total Weight of Aggregate	30	)	-
30	Percent of RAP E	Binder by Weight of Total Binder	25	5	17 Max
31		RAP Contribution by Mix	1.4	3	
32		RAP Contribution by Aggregate	1.5	0	
33		RAP NCAT Correction Factor	1.8	2	

Figure D.10: Job Mix Formula for Project No. 10

С	la	<b>S</b> !	51	
р	ro	i.	D.C	÷

12.5mm SP-3 PG 70-28 SII-55 Snake River Bridge Marsing, SII-55

Laboratory Values		Target		Spec.
Total Asphalt by Weight of Mix % (Pb)		5.3		
Total Aspalt by Weight of Aggregate		5.63		
Air Voids % (Va)		4.0		3.0-5.0
Voids in Mineral Aggregate (VMA)		14.4		14.0
Voids Filled with Asphalt (VFA)		72		65-75
Bulk Specific Gravity (Gmb)		2.320		
Unit Weight lb./cuft.		144.4		
Theo Max Spec Gravity (Gmm)		2.418		
Theo Max Spec Gravity lb./cuft.		150.5		
Effective Specific Gravity of Blend (Gse)		2.616		
Effect of Water on Compressive Strength (AllWest)		89		85 min
Ninitial (7 Gyrations)	-	88.9	≤ 89.0	
Ndesign SP-3 ( 75 Gyrations )		96.0		= 96.0
Nmax (115 Gyrations)		97.8		≤ 98.0
NCAT Asphalt Correction Factor		0.23		
Dust to Asphalt		1.2		0.6 - 1.2
Laboratory Mixing Temperature( deg in F)		320		
Laboratory Compaction Temperature(deg in F)		300		
Plant Mixing Temperature(deg in F)**	316	(h) -	324	
Field Compaction Temperature(deg in F)**	295	-	303	
Superpave Design Sample Wt. in grams		4615		

### Proposed Job Mix Formula

\*Field mixing and compaction may be adjusted +/- 25 degrees per Viscosity Graph

### **Aggregate Gradation Data**

Sieve Size	1/2" Chips 15.0%	#4 Chips 12.0%	C Pile	Washed C Pile 4.0%	C-33 Sand	RAP 31.0%	Break Down 1.0%	JMF Blended Gradation
1" / 25mm	100	100	100	100	100	100	100	100
3/4" / 19mm	100	100	100	100	100	100	100	100
1/2" / 12 5mm	74	100	100	100	100	94	100	94
3/8* / 9 5mm	16	79	100	100	100	86	100	. 81
No 4/4.75mm	2	4	81	75	98	65	100	57
No. 8 / 2.36mm	2	2	58	46	87	50	100	43
No.16 / 1.18mm	1	2	42	28	78	40	100	34
No. 30 / 600um	1	2	29	16	56	30	100	25
No. 50 / 300um	1	1	20	.7	15	18	100	14
No 100 / 150um	1	1	12	3	2	12	100	9
No. 200 / 75um	0.8	0.9	7.8	1.7	0.7	7.5	90.0	5.7

\* Aggregate breakdown will be controlled by the Hot Plant dust control system

### Figure D.11: Job Mix Formula for Project No. 11

				Class:		12.5mm S	P-3 PG 64-3	4
				Project:		SH-55, M	P91 to Smith	's Ferry
		Pro	posed J	ob Mix	Formula	9		
Laboratory Val	1165		Poord o	OD MALA	T	Torget		Snoo
Total Asphalt by W	eight of M	x % (Pb)			1	\$3	/	spec.
Total Aspalt by Wei	cht of Ace	regate				5.64		
Air Voids % (Va)	6		_			4.0		30.50
Voids in Mineral Ag	gregate (V	(MA)			<u> </u>	14.3		14.0
Voids Filled with As	phalt (VF	A)			<u> </u>	72		65-75
Bulk Specific Gravit	y (Gmb)	-/				2.320		03-73
Unit Weight lb./cuft						144.4		
Theo Max Spec Gra-	vity (Gmm	1)				2.417		
Theo Max Spec Gra	vity lb./cut	ft.				150.5		
Effective Specific Gr	ravity of B	lend (Gse)				2.616		
Effect of Water on C	ompressiv	e Strength	(AllWest)			91		85 min
Ninitial (7 Gyration	5)					88.8		< 80.0
Ndesign SP-3 ( 75 G	vrations)					96.0		= 06.0
Nmax (115 Gyration	ne )	_				90.0		= 96.0
NCAT Asphalt Corr	ection Fac	tor				97.7		<u>≤ 98.0</u>
Dust to Asphalt	ection Pac					0.25		
Laboratory Mixing	Temperate	urel deg in	E)			1.2		0.6 - 1.2
Laboratory Gampoo	tion Ton	re( deg in	r)			16 m		
Laboratory Compac	tion Temp	erature(de	g in F)			285	_	
Plant Mixing Tempe	rature(deg	; in F)**			302		310	
Field Compaction Te	emperatur	e(deg in F)	••		281	-	289	
Superpave Design Sa	mple Wt.	in grams		(		4615	_	
<ul> <li>Field mixing and comp</li> </ul>	paction may	be adjusted	+/- 25 degr	ees per Visco	osity Graph			
	_	Agg	gregate	Gradati	on Data			
Sieve Size	1/2" Chips	#4 Chips	C Pile	Washed C Pile	C-33 Sand	RAP	Break Down	JMF Blended
	14.0%	15.0%	22.0%	8.0%	9.0%	31.0%	1.0%	Gradation
1" / 25mm	100	100	100	100	100	100	100	100
3/4" / 19mm	100	100	100	100	100	100	100	100
1/2" / 12.5mm	74	100	100	100	100	94	100	95
3/8" / 9.5mm	16	79	100	100	100	86	100	81
No. 4 / 4.75mm	2	4	81	75	98	65	100	55
No. 8 / 2.36mm	2	2	58	46	87	50	100	41
No.16 / 1.18mm	1	2	42	28	78	40	100	32
No. 30 / 600um	1	2	29	16	56	30	100	23
No. 50 / 300um	1	1	20	7	15	18	100	13
No. 100 / 160-m	1	1	12	1	2	12	100	
No. 1007 150um					6 1	1.6	100	

Figure D.12: Job Mix Formula for Project No. 12

						Projec Project I	t Name: Number:	FY 20 D A019(81	6 Balla 12)	st Stabili	zation	
					Proj	ect Key I	vumber:	19812				
Asphalt Mix Type:	12.5mm	n SP5 (D	esign Es	SALs, 2	10)	Mix Producer: Knife River						
JMF Identification:	20030-1	TD-12.5	mmSP5-	R27			Spec	Hied Oil:	PG 64	34		
Binder Supplier:	Idaho /	sphalt \$	Supply				Adiu	sted Oil:	PG 58-	34		
Anti-Strip:	Morlife	rlife 5000					Source	Number:	CI-56s			
		Jo	b Mix Fo	mula 1	argets a	nd Volun	netric D	ata				
Target Oil Content:	4.9			RAPO	atagory:	Cat 1		JMP	Gmb	Specimen	Weight:	4725 g
Virgin Oil Added:	3.5	-	RAP	Oil Con	tribution:	1.4		JM	F Ncat	Correction	Factor:	0.07
Percent Anti-Strip:	0.75	Compaction Temp. Range			291-306		Lai	b Comp	ection Te	mp (F*):	300	
RAP Percent By Binder:	29		Mixin	ng Temp	Range:	320-333			RAP	Correctio	n Factor	0.07
		As	gregate	RAP at	nd Binde	r Specifi	c Gravit	lies				
Combined G at :	2.619	Effect	ve Speci	fic Gravit	y (G):	2.647		R	AP Bul	k Specific	Gravity:	2.617
Fine Aggregate G	2.616	F	ine SSD	Specific	Gravity :	2.636	Fin	e Appare	nt Spec	ific Gravit	y (G):	2.670
Coarse Agg. Coarse G ab :	2.625	Coa	se SSD	Specific	Gravity :	2.639	Coarse Apparent Specific Gravity (G):				2.665	
Fine Agg.% Absorbtion:	0.8	Coarse	Aggrege	te % Ab	sorbtion:	0.6		Bind	ler Spe	clfic Grav	ity (G ):	1.0287
		Jo	b Mix Fo	rmula T	arget Vo	lumetric	Proper	tles				
				Design	Spec.						Design	Spec.
Maximum Theoretics	d Specifi	ic Gravity	(G .mm):	2.458	-		Bu	lk Specifi	c Gravi	ty (G mb):	2.360	-
	Percent Air Voids (P.)					V	oids In i	Mineral A	greget	e (VMA):	14.3	14.0 min
Voids	Filled wi	th Aspha	# (VFA):	72	65-75			Dust	Proport	ton (DP):	0.9	0.8-1.6
	Absorb	ed Apsha	Att (P to ):	0.42	-		Effecti	ve Aspha	t Conte	nt (P to):	4.50	- 1
Den	sity Perc	ent G am	@ Nini:	87.8	≤ 89.0		Dens	ity Percen	t G <sub>mm</sub>	@ Ndes:	96	96.0
Densit	y Percer	nt G mm	Nmax:	97.3	≤98.0	Imn	nersion (	Comp. Re	tained	Strength:	90	85 min
Uncompacted Vold Co	ontent of	Fine Ag	pregate:	52	45 min			5	and E	uivalent:	74	45 min
Fracture Face Coarse Age	gregate	(1 Fecel	2 Face):	100/99	98/98		Fla	t and Elor	galed	Particles:	0	10 max
		JM	F Target	Apgreg	ate, Brea	kdown, i	Proport	lons				
Blend Percenteoes:		CI-56e	28	BD	0.0	CI-56s	44.0	RAP:	28.0	-		
	Jo	b Mix Fe	ormula A	ggregat	e Grada	tions & S	tockpil	e Average	05	2 5 5	A CENT	1000
	1*	3/4*	1/2"	3/8"	84	#8	#16	#30	#50	#100	#200	
JMF Specifications:		100	90-100	90 max		28-58		102.9			20-10.0	É.
JMF Target Gradation:	100	100	92	80	54	36	25	19	13	8	4.0	
Neat C.F. Gradation:	100.0	100.0	90.9	80.2	55.5	36.8	26.1	18.8	13.2	7.6	4.1	
Aga. Correction Factor:	0.0	0.0	1.1	-0.2	-1.5	-0.8	-1.1	0.2	-0.2	0.4	-0.1	
8-Pile Average:	100	100	84	53	4	1	1	1	1	1	0.3	-
C-Pile Ava.:	100	100	100	100	88	59	40	28	19	11	6.5	
		100		and the second second		-		1001				

Figure D.13: Job Mix Formula for Project No. 13

	Gyratory Model # Compactor: Serial #	AFG2AS 8435		Job Mix	Formula	Spec
1	Perc	ent Asphalt by Weight o	f Total Mix	5.	3	-
2	Perce	nt Asphalt by Weight of	Aggregate	5.	6	-
3		Virgin Asphalt by We	ight of Mix	3.7	72	-
4	Virg	in Asphalt by Weight of	Aggregate	3.9	96	-
5		Percent Air	Voids (Pa)	4.	.0	4.0
6		Voids in Mineral Aggreg	ate (VMA)	14	14,8	14 min
7		Compacted Unit Weigh	t Gmb, pcf	2.331	145.1	-
8	Theor	elical Maximum Density	Gmm, pcf	2.428	151.1	-
9	Perce	nt Effective Asphalt Co	ntent (Pbe)	A	83 4,76	-
10		Percent Absorbed As	phałt (Pba)	بمر	71 0.57	-
11		Specific Gravity of E	inder (Gb)	1.0	29	-
12	Perc	ent Gmm @ N Initial (7	Gyralions)	86	5.8	≤ 89.0
13	Percen	Gmm @ N Design (75	Gyrations)	96	5.0	96.0
14	Perce	nt Gmm @ N Max (115	Gyrations)	97	.2	≤ 98.0
15		Dust to Asphall	Ralio (DP)	1	.2	0.8-1.6
16		Percent Passing	#200 Sieve	5	.6	2.0-10.0
17		Voids Filled w/ As	phalt (VFA)	27	2 73	65-75
18	Laboratory M	lixing Temperature for I	Design (°F)	33	22	317-326
19	Laboratory Compa	action Temperature for I	Design (*F)	2	99	295-303
20	Laboratory Sample	Weight for Volumetric	Testing (g)	4,6	580	-
21	(L5-134) Ignition Oven (I	(CAT) Correction Facto	r @ 538 °F	0.	33	
22		*Los Angeles Abrasion	(LAR) (%)	2	.7	30 max
23		*Idaho Degradatio	n ∆ % -200	4	.3	5.0 max
24		Sand	Equivalent	6	6	40 min
25		*Fracture Face	Count (%)		/97	75/60
26		Fine Aggregate An	gularity (%)	46	5.7	40 min
27	*Flat and Elongated F	Particles in Coarse Agg	regates (%)	0	.0	10 max
		Recycled Asphalt Pav	ement (RAP	Propertie	36	
28	Percentage	of Asphalt in Yard RAP	Wt. of Mix)	3.	78	
29	Percentage of /	Asphalt in Coarse RAP	Wt. of Mix)	2.	67	-
30	Percentage of RAP by	Total Wt of Aggregate	(Yard RAP)	3	32	
31	Percentage of RAP by To	tal Wt of Aggregate (Ce	parse RAP)	1	14	-
32	Percent of RA	P Binder by Weight of 1	fotal Binder	3	30	30 max
33		RAP Contribu	tion by Mix	1.	.58	-
34		RAP Contribution by	Aggregate	1.	.64	-

Figure D.14: Job Mix Formula for Project No. 14

	Gyratory Model # AFG2AS Compactor: Serial # 8732	Job Mix Formula	Spec
1	Percent Asphalt by Weight of Total Mix	5.2	
2	Percent Asphalt by Weight of Aggregate	5.5	
3	Virgin Asphalt by Weight of Mix	3.88	
4	Virgin Asphalt by Weight of Aggregate	4.11	
5	Percent Air Voids (Pa)	4.0	4.0
6	Voids in Mineral Aggregate (VMA)	14.3	14 Min
7	Compacted Unit Weight Gmb, pcf	2.497 155.4	
8	Theoretical Maximum Density Gmm, pcf	2.602 162.0	
9	Percent Effective Asphalt Content (Pbe)	4.2	
10	Percent Absorbed Asphalt (Pba)	1.03	
11	Specific Gravity of Binder (Gb)	1.031	
12	Percent Gmm @ N Initial (7 Gyrations)	87.3	≤ 89.0
13	Percent Gmm @ N Design (75 Gyrations)	96.0	96.0
14	Percent Gmm @ N Max (115 Gyrations)	97.3	≤ 98.0
15	Dust to Asphalt Ratio (DP)	1.4	0.8-1.6
16	Percent Passing #200 Sieve	6.0	2.0-10.0
17	Voids Filled w/ Asphalt (VFA)	72	65-75
18	Laboratory Mixing Temperature for Design (°F)	324	316-324
19	Laboratory Compaction Temperature for Design (°F)	298	295-303
20	Laboratory Sample Weight for Volumetric Testing (g)	4950	
21	Ignition Oven (NCAT) Correction Factor @ 482 °F	1.36	
22	Sand Equivalent	74	40 min
23	Fracture Face Count (%)	100/99	75/60
24	Fine Aggregate Angularity (%)	49.5	40 min
25	Flat and Elongated Particles in Coarse Aggregates (%)	5.5	10 Max
	Recycled Asphalt Pavement (RA	P) Properties	
26	Percentage of Asphalt in RAP (Wt. of Mix)	4.4	
27	Percentage of RAP by Total Weight of Aggregate	30	
28	Percent of RAP Binder by Weight of Total Binder	25.4	30 Max
29	RAP Contribution by Mix	1.32	
30	RAP Contribution by Aggregate	1.38	
31	RAP NCAT Correction Factor	1.33	

## Figure D.15: Job Mix Formula for Project No. 15

(	Gyratory Model # Compactor: Serial #	AFG2AS 8732	Job Mix	Formula	Spec
1	Percer	nt Asphalt by Weight of Total Mi	× 5	5.6	
2	Percent	Asphalt by Weight of Aggregat	e 5	5.9	
3		Virgin Asphalt by Weight of Mi	x 5.	.10	
4	Virgin	Asphalt by Weight of Aggregat	e 5.	.40	
5		Percent Air Voids (Pa	) 4	4.0	4.0
6	V	oids in Mineral Aggregate (VMA	) 1:	5.0	14 min
7	c	compacted Unit Weight Gmb, po	f 2.453	152.7	
8	Theoret	ical Maximum Density Gmm, po	f 2.555	159.0	
9	Percent	t Effective Asphalt Content (Pbe	) 4.	.66	
10		Percent Absorbed Asphalt (Pba	) 0.	.95	
11		Specific Gravity of Binder (Gb	) 1.0	034	
12	Perce	nt Gmm @ N Initial (7 Gyrations	) 8	6.3	≤ 89.0
13	Percent 0	Gmm @ N Design (75 Gyrations	) 90	6.0	96.0
14	Percent	Gmm @ N Max (115 Gyrations	) 9	7.4	≤ 98.0
15		Dust to Asphalt Ratio (DF	) 1	1.4	0.6-1.5
16		Percent Passing #200 Siev	в 6	3.5	2.0-10.0
17		Voids Filled w/ Asphalt (VFA	) 7	73	65-75
18	Laboratory Mix	king Temperature for Design (°F	) 3	25	317-327
19	Laboratory Compac	tion Temperature for Design (°F	) 3	04	295-306
20	Laboratory Sample \	Weight for Volumetric Testing (g	) 48	825	
21	Ignition Oven (NC	CAT) Correction Factor @ 538 °	F 0.	.59	
22	*	Los Angeles Abrasion (LAR) (%	) 1	17	30 max
23		*ldaho Degredation ∆ % -20	0 4	4.1	5.0 max
24		Sand Equivaler	it e	65	40 min
25		*Fracture Face Count (%	) 100	0/100	75/60
26		Fine Aggregate Angularity (%	) 4	6.3	40 min
27	*Flat and Elongated Pa	rticles in Coarse Aggregates (%	) 1	1.6	10 Max
	F	Recycled Asphalt Pavement (I	AP) Propert	ies	
28	**Percentag	ge of Asphalt in RAP (Wt. of Mix	) 4.	.64	
29	Percentage of R/	AP by Total Weight of Aggregat	e	10	
30	Percent of RAP	Binder by Weight of Total Binde	r	8	17 Max
31		RAP Contribution by M	x 0.	.46	
32		RAP Contribution by Aggregat	e 0.	.49	
33		RAP NCAT Correction Facto	r 1.	.67	

## Figure D.16: Job Mix Formula for Project No. 16

September 29, 2020

RE:	1/2"	SP-5	HMA	C-JMF
-----	------	------	-----	-------

#### Project: I-84 Jerome IC to Twin IC Key: 20559 & 20596

Mr.

is planning to provide 1/2" SP-5 hot mix per JMF A520-0304 from our hot plant located south of the Con Paulos Dealer off I-84 exit 168. The JMF will be altered to target the following C-JMF during production. We anticipate a potential need to adjust the Gmm and Gse again after we get results for our first day's production.

Asphalt Content target: 5.2% by weight of mix (before AAO adjustment).

Gmm: 2.442 Gse: 2.632

Gradation Targets:

3/4" 100% 1/2" 93% 3/8" 80% #4 52% #8 34% #16 23% #30 17% #50 12% #100 9% #200 5.3% Please call or email me with any questions

or cell phone

Thank you,

Figure D.17: Job Mix Formula for Project No. 17

	Design Specifica	ations: Blend 1 / 100 Gyrations	s @ N Desig	n PG	70-28
	Gyratory Model # Compactor: Serial #	AFG2AS 8436	Job Mix Fo	ormula	Spec
1	Percer	t Asphalt by Weight of Total Mix	5.7		
2	Percent	Asphalt by Weight of Aggregate	6.09		
3		Virgin Asphalt by Weight of Mix	4.78		-
4	Virgin	Asphalt by Weight of Aggregate	5.08		
5		Percent Air Voids (Pa)	4.0		4.0
6	Ve	oids in Mineral Aggregate (VMA)	13.5		13 min
7	С	ompacted Unit Weight Gmb, pcf	2.206	137.3	-
8	Theoret	ical Maximum Density Gmm, pcf	2.298	143.0	
9	Percent	Effective Asphalt Content (Pbe)	4.44		
10		Percent Absorbed Asphalt (Pba)	1.38		-
11		Specific Gravity of Binder (Gb)	1.031		
12	Percer	nt Gmm @ N Initial (8 Gyrations)	86.6		≤ 89.0
13	Percent Gr	nm @ N Design (100 Gyrations)	96.0		96.0
14	Percent	Gmm @ N Max (160 Gyrations)	97.2		≤ 98.0
15		Dust to Asphalt Ratio (DP)	1.3		0.8-1.6
16		Percent Passing #200 Sieve	5.6		2.0-8.0
17		Voids Filled w/ Asphalt (VFA)	70		65-75
18	Laboratory Mix	ing Temperature for Design (°F)	330		324-338
19	Laboratory Compact	ion Temperature for Design (°F)	304		297-311
20	Laboratory Sample V	Veight for Volumetric Testing (g)	4340		
21	Ignition Oven (NC	AT) Correction Factor @ 538 °F	0.27		
22		Sand Equivalent	47		45 min
23		Fracture Face Count (%)	100/10	0	98/98
24		Fine Aggregate Angularity (%)	45.6		45 min
25	Flat and Elongated Par	ticles in Coarse Aggregates (%)	0.0		10 Max
	R	ecycled Asphalt Pavement (RAP	) Properties		
26	Percentag	e of Asphalt in RAP (Wt. of Mix)	5.07		
27	Percentage of RA	P by Total Weight of Aggregate	19		
28	Percent of RAP E	Binder by Weight of Total Binder	17		17 max
29		RAP Contribution by Mix	0.96		-
30		RAP Contribution by Aggregate	1.01		
31		RAP NCAT Correction Factor	0.51		

Figure D.18: Job Mix Formula for Project No. 18

Class:		SP3 19mm (3	(4"')	
Fiojec		20051		
BROBOSED IOD MU	FORMU	20051		
I aboratory Curatory Values	Min	Target	Max	Ener
Total Asphalt by Weight of Mix % (Ph)	Min	5.00	Max	Spec.
Virgin Asphalt by Weight of Mix Hot Plant		3.64		
Ran Rinder Replacement 27.2%	_	1 36		
Air Voids % (Va)	-	4.0		4.0
Voids in Mineral Aggregate (VMA)	-	13.3		13.0 Min
Voids Filled with Asphalt (VFA)	-	73.0%		65.75
Dust Patio(PCS 35% passing #8 / 0.8%-1.6% MS2)	-	1.18		0.6-1.4
Bulk Specific Gravity (Gmb)	-	2 368		0.0-1.4
Unit Weight Ib./cuft.		147.4		
Theo Max Spec Gravity (Gmm)		2.466		
Theo Max Spec Gravity lb./cuft.	-	153.5		
% Gmm @ Nini(7 gyrations)		85.6%		89% Max
% Gmm @ Ndes(75 gyrations)		95.9%		96% Max
% Gmm @ Nmax(115 gyrations)		97.0%		98% Max
Effective Specific Gravity of Blend (Gse)		2.661		
Specific Gravity of Aggregate (Gsb provided by ITD)		2.595		
Ideal CT, Cracking Test		93.1		80 index value
Hamburg Wheel Track T324		20816		15000
Fine Aggregate Angularity		46%		40% Min
NCAT Asphalt Correction Factor(538 deg C)		-0.44		
Sand Equivalency (SE)		67%		40% Min
Flat and Elongation		1%		10% Max
Percent Fracture 1 Face		97%		75% Min
Percent Fracture 2 Face	12	96%		60% Min
Laboratory Mixing Temperature( deg in F)		320 deg		
Laboratory Compaction Temperature(deg in F)		295 deg		
Plant Mixing Temperature(deg in F)	315 d eg		3329 deg	
Field Compaction Temperature(deg in F)	288 deg		302 deg	
Super pave Design Sample Wt. in grams		4650		

Figure D.19: Job Mix Formula for	Pro	ject	No.	19
----------------------------------	-----	------	-----	----

									i on						
Key Number Pro	ject Number				TF	roječt Na	me	_					_	District	
19373	US-95, Whitebird GradeRehab														
Identification Numb	er (Program/1	ask/Phase/Sam	pie#)			Contract I	tem Numb	er Te	esting Lat	poratory N	ame & Loca	tion I	Mix Dec		
	E	163590				405-435A ITD Central Laboratory						1	9020		
Send Reports To(R	esident Eng	neer's Name)	Sampled	By			WAQTCN	Numbe	er Date :	Sampled	Date Lab R	eceived	Date Lat	Tested	
Curtis Arnzen Greg Mitche							49376 6/17/2019					019	6/18	8/2019	
Asphalt Binder Sup	plier	Asphalt Bir	nder Grade	ISp. Gr. (	of Bind	er (from Mix (Jesion) 1 ME Intended Bioder W. (by ML of Mix)							Courses 1		
Idaho Aspha	It Supply	PG	70-28	1.0.00		1.032		" [""	an interio	5	Q	( ( ( ( ( ( ( ( ( ( ( ( ( ( ( ( ( ( (		256o	
Sample Location (S	ta./offset, tri	ck, plant, lab,	etc.)	Mix Des	ign La	& Local	tion	ISPM	IDT		PE in P	esoonsible	- Charge	-ZODC	
	Lab Prod	uced		1 S.S.	B	oise Id	0	0	Grea M	itchell		Garv th	omnso	r or Design n	
ESALs		Nom, Max, Si	ze Aooreca	le l	1200	any Con	rol Sidero	-	ll aroad	longing l			0		
1 <10 (75 Gy	rations)		1/2"		1	N	0.8		Percent	Passing P 30	nmary Cont	rol Sievę	Class	of Mix	
Combined Asso	name Date	000.0			_		0.0		<u> </u>	- 35	70			5P3	
Combined Aggr	egate Bull	CSPG G <sub>sb</sub> I	rom ITD (	0802	_	2.7	03	_							
					T	est Res	ults	_							
	Grad	ation Anavisi	s			1		Aso	halt Bine	ler Conte	nt /Du Mala	bt of M			
	FOP for	AASHTO T	30					мэр	nait bill(	D for At	ento z ace	Int of Mix	,		
	Lab No.	Lab No.	Lab No						FU	- TOP AA	3010 1 308				
	190MX	200 110.	200 140,						L	ab No.	Lab No.	Lab No	0.		
Sieve Size	0061								1	90MX					
(mm) (in)	0001			L Av 1	10.000			-		0061				Average	
(50) 2	100			Avg.	JMF	Total As	phalt Binde	er Cor	ntent	6.25				1	
(30) 2	100			+ +	100	NCAT C	orrection F	Factor	r	0.55					
(37.5) 1/1/2	100				100	Moistur	e % (-)		·····	0.02				+	
(25) 1	100				100	Act. Asp	h. Binder	Conte	ent %	5.72			— r		
(19) 3/4	100				100	1									
(12.5) 1/2	92				92	Compaction Temperature, °F									
(9.5) 3/8	80				81	#	/ALUE!		FOP	for AASH	TO T 312 / /	ASHTO I	M 323	Average	
(4.75) No. 4	57				58	Lab Air	Voids % a	t N <sub>De</sub>	sign	4.4		1			
(2.36) No. 8	37				38	G <sub>mb</sub> (cor	mpacted m	ixture	e) 2	.422		<u> </u>			
(1.18) No. 16	24				25	G <sub>mm</sub> (ma	ax spec gra	avity)	2	.532		-			
(0.600) No. 30	15				17	VMA, %				15.5		<u> </u>			
(0.300) No. 50	11				12	VFA, %	********			72		1-			
(0.150) No. 100	8				9	Dust Pro	portion (D	P)		14		<u> </u>			
(0.075) No. 200	6.6				6.7			· · · · · ·		1.4	_				
						G., - F#	ective So	Gravil	ity 2	777		I I			
Avg. Sample						P. Eff	Binder Co	olani	NY 2	1 76		<u> </u>			
Height, mm	114.4			1		P. Bin	der Abcort	had 6	V. /0 ·	102					
				1		- 54 - 1511	001 70501	Jeu, 7	/a	1.02					
FOP for AASHTO	O T 209 resu	lt within 0.020	of IME2	Gmm fro	m IME	- 25		6.0.00	Complet			-			
FOP for AASHTO	O T 166 resu	It within 0.020	of JME2	Conh from	m JAF	- 2.5	Gmm	nom	Sample	i ested≌	2.532	Pass			
			of own, t	Sino no	ni JMP	= 2.4	Gmb	nom	sample	lested=	2.422	Pass			
	ASTM D103	5 8 AACUTO	T 467						_						
Sample #	B1	a Manio	1 10/			-		No.	AAS	HTO T 34	40				
Sample #	01	-				S	ample #	100	61						
100	EN	110-21	ICC COOL		-				_						
100 % @ 0.05 % MORLIFE 5000						Rutting Depth, mm 2.2 Maximum Allowable									
<u> </u>	M35					Lef	t Sample		2	.274		Rut Depth	1		
						Center Sample 1.998 0.2 in. ( 5 mm)									
						Rig	ht Sample		2	.278	F	ASS	-		
								_							
Pomoder	_	_			_	_		_			_		_		

Figure D.20: Job Mix Formula for Project No. 20

### ASPHALT CONCRETE JOB MIX FORMULA

Page 5

	-
Project:	Date: 7/18/2016
Paving Contractor: V	Class of Mixture: SP-3
Asphalt Supplier:	Specified Grade of Asphalt: PG 64-34
Anti-Strip Agent: 0.5% Morelife 5000	Prepared by: LLC
Aggregate Sources: BE 106c	Gyratory Compactor: Pine AFGC125X
Design Specification: ITD	

	AASHTO	JMF	Requirements
1. Asphalt by Weight of Total Mix, %	R 35	5.8%	
2. Asphalt by Weight of Aggregates, %		6.1%	
3. Asphalt by Weight of Total Mix (Added), %		5.0%	
4. Asphalt by Weight of Aggregates (Added), %		5.2%	
5. Air Voids (Va), %	T 269	4.0%	3.0-5.0
6. Voids in Mineral Aggregate (VMA), %	R 35	15.0%	13.3 min
7. Bulk Specific Gravity @ Ndes (Gmb)	T 166	2.296	142.9 pcf
8. Theoretical Maximum Specific Gravity (Gmm)	T 209	2.391	148.8 pcf
9. Relative Density %Gmm @ Nini (7 Gyrations)	R 35	87.4	≤ 89.0
10. Relative Density %Gmm @ Nmax (115 Gyrations)	R 35	97.2	≤ 98.0
11. Voids Filled w/ Asphalt (VFA), %	R 35	73.3%	65-75
12. Film thickness, microns		1	
13. Absorbed Asphalt (Pba) by Weight of Aggregate, %	R 35	0.90%	
14. Effective Asphalt Content (Pbe) by Total Wt of Mixture, %	R 35	4.9%	
15. Specific Gravity of Asphalt		1.026	
16. Laboratory Mixing Temp, °C/°F		318-333	
17. Laboratory Compaction Temp, °C/°F		288-304	
18. Recommended Plant Mixing Temp, °F		325	
19. Compaction Temp Range, °F		296	
20. NCAT Ignition Oven Correlation Factor @ 538° C	T 308	0.17	
21. Dust to Asphalt Ratio	R 35	1.2	0.6-1.2
22. Immersion Compression Retained Strength, %	T 165	85%	85% min
23. Gyratory Gmb specimen weight, grams		4500	
24. Combined Bulk Dry Specific Gravity of Aggregate (Gsb)	T 85 / IT 144	2.545	

Figure D.21: Job Mix Formula for Project No. 21

Key Number Project Number							Project Name District											
Key Number	Piojec	a Number															District	
20051	A020	(051)		Cont	been the	an bhamh a		US-30,	Carib	ou Co. to Georgetown	Bea	ar Lake C	io, ID		n filme	000	5	000
Send Reports To	(Reside	nt Engineer's	Name)	Con	uact ne	m Number	r	Class of	Jass of Mix ESALS Nomin					ax Ag	g Size	PUS	Passing	PCS
Eric Staats				405	-435A			SP3		1 < 10 (75 Gyrations)	)		3/4"			No. 4		47 %
C-JMF Number	C-JMP Number C-JMP Target P <sub>b</sub> Aggregate Sour						Irce Num	ber	Contractor Producing N	/lix		Designed by						
231043	231043 4.8 BL70									Idaho Materials and C	Cons	struction HK C				Contractors		
Virgin Binder Grade Anti-strip Additive Listed on G						QPL	Asphalt	Binder	Supplier			% Anti-Str	ip Add	itive	% Binder Re	eplacement		
PG 58-34		Moorelife	5000			Yes		Idaho A	sphalt	t Supply			0.5			22.0		
FOP for AAS	нто т	168 Samp	oling Bitu	minous	s Pavir	ng Mixtur	es			Sample	ID I	Number	SB/P1	75550	)-A-CE/A#	659		
Test Number	Date :	Sampled	Time Sa	mpled	Samp	le Temper	ature	Samplin	g Meth	od Nasifyin Demeska)			Sample Lo	cation	(Sta./offse	et, truck, plan	t, lab, etc.)	
A#659 Semeled By	10/21	/2020	2:30 PI	1	WAG	293 TC Number	- <b>F</b>	Other (I	Must a	specity in Remarks)			MP 413.3	5, 6.5 <sup>°</sup>	LI OF CL	4	Lift Thial	
Sampled by					0054	A NUMBER	-	MTI	e cuit	loyer			Que	nuty ry	epresente 4400		0.2	10000
S. Borchert					2351	1		MIT							1408	IONS	0.2	
EOD for AASHTO R47 Reducing Samples of Hot Mix Asphalt (HMA) to Testing Size (Initial Reduction of Sample)																		
Oualified Lab No	Testin	a Laborato	ry Locatio	0		Aspirait	(1111	Initial Re	aductio	n Method	110	i əanipi	•)	8	nlit Retain	ed for Disput	Solit ID	Number
LO 5014	Poce	tello ID	ry cocare					Quarter	ing M	ethod (Full Quartering	۱.				pin retain	са юг отврал Vac	A#650	Annoei
Initial Reduction	Perform	ed By			WAO	TC Numbe	ər	Technici	an's E	molover	/		Date Red	Iced	Time Rev	duced Sa	nole Tempe	erature
D. Essenth	Chom	ed by			2255	A	-	MTI		mpioyer			10/22/20	20	2.00 AM		100 Temps 200	*F
B. French					2355	14		MIT					10/22/20	20	2:00 AN		200	Ŧ
FOP for AASHT	ОТ 20	9 Theoreti	cal Max	Specific	Gravit	v (Bowl M	Netho	(bc				Summa	rv of Mix	Prope	erties			
T209 Sample Red	duction	Method			Date	Reduced	Ti	me Redu	ced (	Sample Temperature	ТΓ	-						
Quartering Metho	od (By A	\pex)			10/2	2/2020				*F	11	Proper	ty Sample	9 1A S	ample 1B	Combine	I LSL	USL
Final Reduction f	or T209	Performed	i By		-		-			WAQTC Number	1 F	G						
B. French										23554	11	⊖sa						
I					Incre	ement 1	Incr	ement 2			11	G	2.64	79	2.6495	2.649		
Mass of Boy	vl (Re	quired)			27	705.8 2707.4 C A						- 50						
Mass of Boy	wl and	Sample			52	228.8	5	$230.1$ $G_{mm} = A - C$		$O_{mm} - A - C$	11	Gab	2.50	9	2 599	2 599		
Mass of Dry	/ Sam	ole in Air (	(A)		25	523.0	- 25	522.7	0.3							2.000		
Agitation Me	ethod					Mecha	anica				11	G	2 46	88	2 4682	2.467		
Water Bath	Temp	erature			77.	5 *F	77.	6 *F				-mm						
Submerged	Weig	ht of Bow	and Sa	nple	32	207.1	- 32	208.2			11	Gent	2.33	85	2.3374	2,338		
Submerged	Weig	ht of Bow			17	706.9	1	707.6			11	Omb				2.000		
Submerged	Weig	ht of Sam	ple (C)		15	500.2	1	500.6	0.4		11	Abe	1 42	10	1 2345	1 328		
G <sub>mm</sub> (Maxim	um S	pecific Gr	avity)		2	.467	2	.468	0.0		11	AD STIC	16			1.020		
Average G <sub>m</sub>	m					2.4	67				11	G.	1.02	20	1 0202	1 0203		
Range 0.0	01 (W	ithin d2s pr	recision?)		1	YES			-		11	06	1.02	~	1.0200	1.0200		
FOP for AASHT	O T 31	2 SuperPa	ve Gyrat	ory Con	npacto	r					Ì	P.	4.66	9	4,669	4 67	4 50	5 10
T312 Sample Rec	duction	Method			Date	Reduced	Ti	me Reduc	ced (	Sample Temperature	Τl	• •				4.01	4.00	0.10
Quartering Metho	od (By A	Apex)			10/2	2/2020				*F	11	Pha	0.73	1	0.755	0.74		
Final Reduction f	or 1312	Performed	ву							WAQTC Number	H	0				0.00		
B. French										23554	11	Pbe	3.97	2	3.949	3.96		
Gyratory Compac	ctor Bra	nd		Model N	Number			Serial N	umber		ы					05.0		
PINE				AFG1A				2243			11	Ps	95.	3	95.3	95.3		
					Spec	cimen 1	Spe	cimen 2	Desi	gn Mass								
Mass of Sam	ple				40	52.3	4	552.5	46	550.0	11	SA	19.	6	19.6	19.6		
Temp. or Sar	npie w	nen Place	a in Moi	1	2:10	95 1	2.00	95 °F	-	Concert inside	H							
Sample Heig	bt (ma	Nogina N			3.10	16.0	3.28	15.7	1	115+5	11	AFT	10.3	7	10.31	10.34		
Sample Heig	nic (nini	9				10.0		19.7	1	11010	Τŀ	D			5 000	E 0E	20	50
FOP for AASHT	O T 16	6 Bulk Spe	cific Gra	rity of C	ompac	ted Mix (	Neth	od A)			- I	Fa	5.20	n	5.298	5.25	3.0	0.0
					Spec	cimen 1	Spe	cimen 2	1		11	VMA	14.2	25	14.264	14.24	13.0	100
Surface Tem	Surface Temperature *F					*F		$G_{-1} = \frac{A}{-1}$	11									
Water Bath Temperature 78 *F 77.					9 *F	1 '	B - C	11	VFA	63.4	40	62.860	63.1	65.0	75.0			
Mass of Puck Dry (A) 4640.7 46					538.8	1.9												
Submerged \	Weight	of Puck in	n Water (	C)	26	584.4	20	578.7	1		11	P200	3,81	9	3.819	3.8	32	62
Wt. of Puck §	SSD (B	9)			- 46	668.9	- 40	663.3	1		11							
G <sub>mb</sub> (Bulk Sp	ecific (	Gravity)			2	.338	2	.337	1			DP	0.96	2	0.967	0.96	0.6	14
Average G <sub>m</sub>	b					2.3	38				11		0.00	-	0.001	0.00	0.0	1.4
Range 0.0	01 (W	ithin d2s pr	recision?)			YES					1 7							

Figure D.22: Job Mix Formula for Project No. 22



# **Appendix E. Additional Figures of Chapter 4**

Figure E.1: Effect of RAP Content of RAP No. 1 on IDT<sub>Strength</sub>



Figure E.2: Effect of RAP Content of RAP No. 1 on WeibullCRI


Figure E.3: Effect of Rejuvenator Type at 25% RAP No. 1 on IDT<sub>Strength</sub>



Figure E.4: Effect of Rejuvenator Type at 25% RAP No. 1 on WeibullCRI



Figure E.5: Effect of Rejuvenator Type at 50% RAP No. 1 on IDT<sub>Strength</sub>



Figure E.6: Effect of Rejuvenator Type at 50% RAP No. 1 on WeibullCRI



Figure E.7: Effect of Rejuvenator Type at 75% RAP No. 1 on IDT<sub>Strength</sub>



Figure E.8: Effect of Rejuvenator Type at 75% RAP No. 1 on WeibullCRI



Figure E.9: Effect of Rejuvenator dose at 75% RAP No. 1 on IDT<sub>Strength</sub>



Figure E.10: Effect of Rejuvenator Type at 75% RAP No. 1 on WeibullCRI

Implementation of Balanced Asphalt Mix Design of Asphalt Mixtures Prepared with Reclaimed AsphaltPavements and Rejuvenators for Enhanced Performance148



Figure E.11: Effect of RAP Content of RAP No. 2 on IDT<sub>Strength</sub>



Figure E.12: Effect of RAP Content of RAP No. 2 on WeibullCRI



Figure E.13: Effect of Rejuvenator Type at 25% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.14: Effect of Rejuvenator Type at 25% RAP No. 2 on WeibullCRI



Figure E.15: Effect of Rejuvenator Type at 50% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.16: Effect of Rejuvenator Type at 50% RAP No. 2 on WeibullCRI



Figure E.17: Effect of Rejuvenator Type at 70% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.18: Effect of Rejuvenator Type at 70% RAP No. 2 on WeibullCRI



Figure E.19: Effect of Rejuvenator Dose at 25% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.20: Effect of Rejuvenator Dose at 25% RAP No. 2 on WeibullCRI



Figure E.21: Effect of Rejuvenator Dose at 50% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.22: Effect of Rejuvenator Dose at 50% RAP No. 2 on WeibullCRI



Figure E.23: Effect of Rejuvenator Dose at 70% RAP No. 2 on IDT<sub>Strength</sub>



Figure E.24: Effect of Rejuvenator Dose at 70% RAP No. 2 on WeibullCRI







Figure E.26: Effect of Rejuvenators on Different Binder Grade at 70% RAP No. 2 on WeibullCRI



Figure E.27: Effect of RAP Content of RAP No. 3 on IDT<sub>Strength</sub>



Figure E.28: Effect of RAP Content of RAP No. 3 on WeibullCRI



Figure E.29: Effect of Rejuvenator Type at 70% RAP No. 3 on IDT<sub>Strength</sub>



Figure E.30: Effect of Rejuvenator Type at 70% RAP No. 3 on WeibullCRI

## **Appendix F. Cost Analysis Detail**

Mix type		Contro	ol 0%	IDEAL- CT <sub>index</sub>	Control 0% +0.5BC		IDEAL- CT <sub>index</sub>	25% RAP		IDEAL- CT <sub>index</sub>	25% RAP + 0.5 BC		IDEAL- CT <sub>index</sub>	25% RAP + R1		IDEAL- CT <sub>index</sub>	EAL- 25% RAP + R5		IDEAL-
Material	Unit Cost	Quantity (Top)	Cost \$		Quantity (Top)	Cost \$		Quantity (Top)	Cost \$		Quantity (Top)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Top)	Cost \$	
Virgin Aggregate	\$ 13.5	0.94	\$12.72		0.94	\$11.92		0.71	\$ 9.54		0.70	\$ 9.49	1	0.71	\$ 9.54		0.71	\$ 9.54	
RAP Aggregate	\$ 6.5	-	-		-	-		0.25	\$ 1.62	1	0.25	\$ 1.61	1	0.25	\$ 1.62	1	0.25	\$ 1.62	
RAP Binder		-	-		-	-	1	-	-	1	-	-	1	-	-	1	-	-	
Binder Content		0.06	-		0.06	-	1	0.04	-	1	0.05	-	1	0.04	-	1	0.04	-	
PG 58-28	\$ 750		\$43.50			\$47.25	1		\$33.06	1		\$36.86	1		\$33.06	1		\$33.06	
PG 58-34	\$ 875		\$50.75			\$55.13	1		\$38.57	1		\$43.00	1		\$38.57	1		\$40.00	
PG 64-28	\$ 800	0.06	\$46.40		6.3%T	\$50.40	1	5.80	\$35.26	1	6.3T	\$39.31	1	5.80	\$35.26	1	5.80	\$35.26	
PG 70-28	\$ 825		\$47.85			\$51.98	1		\$36.37	1		\$40.54	1		\$36.37	1		\$36.37	
Tall Oil	\$ 4,450	-	-		-	-		-	-	]	-	-	]	0.004	\$18.07	]	-	-	
Waste Vegetable Oil	\$ 3,800	-			-	-		-		]	-		]	-	-	]	0.002	\$ 8.46	
	PG 58-28		1.1		-			-	1.1	]	-		]		1.1	]	-	1.1	
Tableset	PG 58-34		\$ 63	39	-	\$ 67	102	-	\$ 50	20	-	\$ 54	46	-	\$ 68	52	-	\$ 60	26
Total Cost \$	PG 64-28	-	-		-	-	1	-	-	1	-	-	-	-	-	1	-	-	
	PG 70-28	-	-		-	-	1	-	-	1	-	-	-	-	-	1	-	-	
	•	1		IDEAL.			IDEAL.			IDEAL.			IDEAL.			IDEAL.			IDEAL.
Mix type		Contro	ol 0%	CT.	Control 0% +0.5BC		CT.	50% RAP		CT.	50% RAP + 0.5 BC		CT.	50% RA	AP + R1	CT.	50% RA	P + R5	CT.
		Ouentitu		Index	Oursetitu		Clindex	Ouentitu		Index	Ouantitu	C		Ouentitu		Clindex	Quantity		Cindex
Material	Unit Cost	Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$		Quantity (Ten)	Cost \$		Quantity (Ten)	Cost \$	
Virgin Aggregate	\$ 135	0.04	\$12.72		(1011)	611.02		0.47	\$ 6.26		0.47	\$ 6.32		0.47	\$ 6.26		0.47	\$ 6.26	
RAD Aggregate	\$ 13.5	0.94	\$12.72		0.94	\$11.92		0.47	\$ 3.24		0.47	\$ 0.32		0.47	\$ 3.24		0.47	\$ 3.24	
RAP Aggregate	\$ 0.5							0.50	\$ 5.24		0.50	\$ 3.22		0.50	\$ 5.24		0.50	\$ 5.24	
Binder Content		0.06			0.06	-		0.03	-		0.04	-		0.03	-		0.03		
Dinder Content	É 750	0.00	- \$42.50		0.00	- ¢ 47.25		0.03	\$22.06		0.04	- ¢26.02		0.05	÷22.06		0.03	\$22.06	
PG 58-26	\$ 975		\$50.75			\$47.25 \$55 12			\$25.00			\$21.42			\$25.00			\$25.00	
PG 50-34	\$ 800	0.06	\$46.40		6.3%个	\$50.40		5.80	\$24.50		6.3个	\$ 28 73		5.80	\$24.50		5.80	\$24.59	
PG 70-28	\$ 825		\$47.85			\$51.98			\$25.36			\$29.63			\$25.36			\$25.36	
Tall Oil	\$ 4,450	-	-		-	-		-	-		-	-		0.004	\$18.07		-	-	
Waste Vegetable Oil	\$ 3,800	-	-		-	-		-	-		-	-		-	-		0.004	\$16.57	
	PG 58-28	-	-		-	-		-	-		-	-		-	-		-		
	PG 58-34	-	\$ 63	39	-	\$ 67	102	-	\$ 36	14	-	\$ 41	18	-	\$ 55	33	-	\$ 53	25
Total Cost \$	PG 64-28																		
	PG 70-28								-										
	101020			IDCAL			IDEAL						IDEAL			IDEAL			IDEAL
Mix type		Control 0%		IDEAL-	Control 0%	+0.5BC	IDEAL-	70% R	AP	IDEAL-	70% RAP + 0.5 BC		IDEAL-	70% RAP + R1		IDEAL-	70% RAP + R5		IDEAL-
				Clindex			Clindex			Clindex			CTindex			Clindex			Clindex
Material	Unit Cost	Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$		Quantity	Cost \$	
	\$/Ton	(Ton)			(Ton)			(Ton)			(Ton)			(Ton)			(Ton)		
Virgin Aggregate	\$ 13.5	0.94	\$12.72		0.94	\$11.92		0.28	\$ 3.82		0.28	\$ 3.79		0.28	\$ 3.82		0.28	\$ 3.82	
RAP Aggregate	\$ 6.5	-	-		-			0.70	\$ 4.53		0.69	\$ 4.51		0.70	\$ 4.53		0.70	\$ 4.53	
RAP Binder		-	-		-	-		-	-			-					-	-	
Binder Content		0.06	-		0.06	-		0.02	-		0.04	-		0.02			0.02	-	
PG 58-28	\$ 750		\$43.50			\$47.25			\$15.23			\$28.35			\$15.23			\$15.23	
PG 58-34	\$ 875	0.06	\$50.75		6.3%↑	\$55.13		5.80	\$17.76		6.3个	\$33.08		5.80	\$17.76		5.80	\$17.76	
PG 64-28	\$ 800		\$46.40			\$50.40			\$16.24			\$30.24			\$16.24			\$16.24	
PG 70-28	\$ 825		\$47.85			\$51.98			\$16.75			\$31.19			\$16.75			\$16.75	
Tall Oil	\$ 4,450	-	-		-			-			-	-		0.004	\$18.07				
Waste Vegetable Oil	\$ 3,800	-	-		-	-		-	-		-	-					0.004	\$16.57	
	PG 58-28		-		-	1.1			1.1		1.1	1.1							
Total Cost \$	PG 58-34	-	\$ 63	39	-	\$ 67	102	-	\$ 26	14	-	\$41	30	-	\$44	40	-	\$43	39
	PG 64-28	-	\$ 59	46	-	-		-	\$ 25	10	-	-		-	\$43	31	-	\$41	41
	PG 70-28	-	\$ 61	40	-			-	\$ 25	14		-			\$43	21		\$ 42	40

Figure F1. Detailed Cost Estimate of Mixtures with RAP No. 2 and Rejuvenators

Mix type		Control 0%		ol 0%	IDEAL- CT <sub>index</sub>	Control 0% +0.5BC		IDEAL- CT <sub>index</sub>	70% RAP		IDEAL- CT <sub>index</sub>	70% RAP + 0.5 BC		IDEAL- CT <sub>index</sub>	70% RAP + R1		IDEAL- CT <sub>index</sub>	70% RAP + R5		IDEAL- CT <sub>Index</sub>
Material	Un	it Cost /Top	Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$	
Virgin Aggregate	\$	13.5	0.94	\$12.72		0.94	\$ 12.65		0.28	\$ 3.82		0.28	\$ 3.79		0.28	\$ 3.82		0.28	\$ 3.82	
RAP Aggregate	\$	6.5	-	-		-	-	1	0.69	\$ 4.48		0.69	\$ 4.46		0.69	\$ 4.48		0.69	\$ 4.48	1
RAP Binder			-	-		-	-	1	-	-		-	-	]	-	-	1	-	-	
Binder Content			0.06	-		0.06	-	]	0.03	-		0.03	-		0.03	-		0.03	-	
PG 58-28	\$	750		\$43.50			\$ 47.25	-	5.80	\$ 20.88			\$ 24.57		5.80 \$ 20.88 \$ 24.36 \$ 22.27 \$ 22.97			\$ 20.88		
PG 58-34	\$	875	F 99/	\$50.75		6.3%↑	\$ 55.13			\$ 24.36		624	\$ 28.67 \$ 26.21			\$ 24.36		F 90	\$ 24.36	
PG 64-28	\$	800	5.6/0	\$46.40			\$ 50.40			\$ 22.27		0.5.1.				\$ 22.27		5.60	\$ 22.27	
PG 70-28	\$	825		\$47.85			\$ 51.98			\$ 22.97			\$ 27.03	1		\$ 22.97			\$ 22.97	
Tall Oil	\$	4,450	-	-		-	-	]	-	-		-	-		0.004	\$ 18.07		-	-	
Waste Vegetable Oil	\$	3,800	-			-	-		-	-		-	-		-	-		0.005	\$ 18.34	
	PG	i 58-28	-	\$ 56	49	-	\$ 60	102		\$ 29	23	-	\$ 33	72	-	\$47	118	-	\$ 48	92
Total Cost \$	PG	58-34	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
	PG	64-28	-	-	-	-	-	-	-	-	-	-	-	-	-	-		-	-	-
	PG	i 70-28	-	-	-	-	-	-	-	-		-	-	-	-	-		-	-	1.1

## Figure F2. Detailed Cost Estimate of Mixtures with RAP No. 3 and Rejuvenators

Mix type		Control 0%		IDEAL- CT <sub>index</sub>	25% RAP		IDEAL- CT <sub>index</sub>	25% RAP+ (R6) 6.6%		IDEAL- CT <sub>index</sub>	25% RAP+ (R6) 8.3%		IDEAL- CT <sub>index</sub>	25% RAP+ (R7)		IDEAL- CT <sub>index</sub>	25% RAP+ Softer Binder (PG 64-34)		IDEAL- CT <sub>index</sub>
Material	Unit Cost \$/ton	Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$	
Virgin Aggregate	\$ 13.5	0.95	\$ 12.78		0.72	\$ 9.72		0.72	\$ 9.72		0.72	\$ 9.72		0.72	\$ 9.72		0.71	\$ 9.58	
RAP Aggregate	\$ 6.5				0.24	\$ 1.56		0.24	\$ 1.56		0.24	\$ 1.56		0.24	\$ 1.56		0.237	\$ 1.54	
RAP Binder	-	-				-		-	-		-				-		-	-	
Binder Content	-	-				-		-	-		-	-		-	-		-	-	
PG 70-28	\$ 825	0.05	\$ 43.72		0.04	\$ 33.00		0.04	\$ 32.50		0.04	\$ 32.34		0.04	\$ 32.01			-	
PG 64-34	\$ 925					-	1	-	-		-	•			-		0.053	\$ 49.02	
Bio-Based Oil	\$ 2,200						]	0.0008	\$ 1.76		0.0011	\$ 2.42							
Petroluem-Based Oil	\$ 1,900	•			-	-	]	-	-		-			0.0014	\$ 2.66	]	-	-	1
Total Cost \$	PG 70-28		\$ 57	90		\$ 44	59		\$ 45.54	43		\$ 46	56		\$ 46	54			
Total Cost 3	PG 64-34	-	-			-	-		-	-	-	-	-		-		-	\$ 60	60
Mix type		Control 0%		IDEAL- CT <sub>index</sub>	50% RAP		IDEAL- CT <sub>index</sub>	50% RAP+ (R6) 6.6%		IDEAL- CT <sub>index</sub>	50% RAP+ (R6) 8.3%		IDEAL- CT <sub>index</sub>	50% RAP+ (R7)		IDEAL- CT <sub>index</sub>			
Material	Unit Cost \$/ton	Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$				
Virgin Aggregate	\$ 13.5	0.95	\$ 12.78		0.49	\$ 6.56		0.49	\$ 6.56	]	0.49	\$ 6.56		0.49	\$ 6.56	]			
RAP Aggregate	\$ 6.5		-		0.49	\$ 3.16	1	0.49	\$ 3.16	1	0.49	\$ 3.16		0.49	\$ 3.16	1			
RAP Binder	-	1.1	1.1				]		1.1	]	1.1					]			
Binder Content	-		-		-	-	1	-	-	]		-		-	-	]			
PG 70-28	\$ 825	0.05	\$ 43.72		0.03	\$ 22.44	]	0.03	\$ 21.45	]	0.03	\$ 20.71		0.02	\$ 20.05	]			
PG 64-34	\$ 925		-	1	-	-	1		-	1	-				-	1			
Bio-Based Oil	\$ 2,200		-	1	-	-	1	0.0017	\$ 3.74	1	0.0021	\$ 4.62			-	1			
Petroluem-Based Oil	\$ 1,900	•	-	1	-	-	1		-	1				0.0029	\$ 5.51	1			
Total Cost 6	PG 70-28		\$ 57	90	-	\$ 32	37	-	\$ 35	36		\$ 35	44	-	\$ 35	33			
Total Cost \$	PG 64-34	1.1		-	-	-	-	-	-		-	-	-	-	-	-			
Mix type		Control 0%		IDEAL- CT <sub>index</sub>	70% RAP		IDEAL- CT <sub>index</sub>	70% RAP+	+ (R6) 6.6%	IDEAL- CT <sub>index</sub>	70% RAP+	+ (R6) 8.3%	IDEAL- CT <sub>index</sub>	70% RA	P+ (R7)	IDEAL- CT <sub>index</sub>			
Material	Unit Cost \$/ton	Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$		Quantity (Ton)	Cost \$				
Virgin Aggregate	\$ 13.5	0.95	\$ 12.78	1	0.29	\$ 3.98	1	0.295	\$ 3.98	1	0.295	\$ 3.98		0.295	\$ 3.98	1			
RAP Aggregate	\$ 6.5		-	1	0.69	\$ 4.47	1	0.688	\$ 4.47	1	0.688	\$ 4.47		0.688	\$ 4.47	1			
RAP Binder	-		-	1	-	-	1	-	-	1	-	-			-	1			
Binder Content	-		-	1	-	-	1	-	-	1	-	-		-	-	1			
PG 70-28	\$ 825	0.05	\$ 43.72	1	0.02	\$ 13.61	1	0.01	\$ 11.63	1	0.01	\$ 8.66		0.01	\$ 6.85	1			
PG 64-34	\$ 925		-		-	-	1	-	-	1	-	-			-	1			
Bio-Based Oil	\$ 2,200	-	-		-	-	1	0.0024	\$ 5.28	1	0.006	\$ 13.42		-	-	1			
Petroluem-Based Oil	\$ 1,900		-		-	-	1	-	-	1	-	-		0.008	\$ 15.58	1			
Total Cout (	PG 70-28	-	\$ 57	90	-	\$ 22	24	-	\$ 25.37	39	-	\$ 31	33	-	\$ 31	29			
Total Cost \$	PG 64-34	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-			

Figure F3. Detailed Cost Estimate of Mixtures with RAP No. 4 and Rejuvenators

Implementation of Balanced Asphalt Mix Design of Asphalt Mixtures Prepared with Reclaimed AsphaltPavements and Rejuvenators for Enhanced Performance160